General Moly, Inc Form PRE 14A October 20, 2017

Use these links to rapidly review the document TABLE OF CONTENTS

Table of Contents

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

SCHEDULE 14A

)

Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934 (Amendment No.

Filed by the Registrant ý

Filed by a Party other than the Registrant o

Check the appropriate box:

- ý Preliminary Proxy Statement
- o Confidential, for Use of the Commission Only (as permitted by Rule 14a-6(e)(2))
- o Definitive Proxy Statement
- o Definitive Additional Materials
- o Soliciting Material under §240.14a-12

General Moly, Inc.

(Name of Registrant as Specified In Its Charter)

N/A

(Name of Person(s) Filing Proxy Statement, if other than the Registrant)

Payment of Filing Fee (Check the appropriate box):

- ý No fee required.
- o Fee computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.
 - (1) Title of each class of securities to which transaction applies:
 - (2) Aggregate number of securities to which transaction applies:

(3)

(4)

o

Proposed maximum aggregate value of transaction:

Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11 (set forth the amount on which the filing fee is calculated and state how it was determined):

Fee paid previously with preliminary materials. Check box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for which the offsetting fee was paid previously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing. (1) Amount Previously Paid: (2) Form, Schedule or Registration Statement No.: (3) Filing Party: (4) Date Filed:	(5)	Total fee paid:
 was paid previously. Identify the previous filing by registration statement number, or the Form or Schedule and the date of its filing. (1) Amount Previously Paid: (2) Form, Schedule or Registration Statement No.: (3) Filing Party: 	Fee p	aid previously with preliminary materials.
(2) Form, Schedule or Registration Statement No.:(3) Filing Party:		
(3) Filing Party:	(1)	Amount Previously Paid:
	(2)	Form, Schedule or Registration Statement No.:
(4) Date Filed:	(3)	Filing Party:
	(4)	Date Filed:

General Moly, Inc.

1726 Cole Blvd., Suite 115 Lakewood, Colorado 80401

November 7, 2017 Dear Stockholder:

You are invited to attend General Moly's special meeting of stockholders. The meeting will be held on December 15, 2017, at 9:00 a.m., local Colorado time, at the Denver West Office Park, Building 22 Room 130, 1726 Cole Blvd., Lakewood, Colorado 80401.

At the meeting, stockholders will vote on a number of important matters. Please take the time to carefully read each of the proposals described in the attached proxy statement.

Your vote is important. Whether or not you plan to attend the meeting, it is important that your shares be represented and voted at the meeting. Therefore, I urge you to sign and date the enclosed proxy card and promptly return it in the enclosed postage paid return envelope so that your shares will be represented at the meeting.

Please note that pursuant to NYSE rules, brokers are not permitted to vote your shares on proposals for any non-routine matters if you have not given your broker specific instructions on how to vote your shares. PLEASE BE SURE TO GIVE SPECIFIC VOTING INSTRUCTIONS TO YOUR BROKER SO THAT YOUR VOTES CAN BE COUNTED.

We look forward to seeing those of you who will be able to attend the meeting.

Sincerely,

Bruce D. Hansen

Chief Executive Officer and Chief Financial Officer

General Moly, Inc.

1726 Cole Blvd., Suite 115 Lakewood, Colorado 80401

Notice of Special Meeting of Stockholders To be Held on December 15, 2017

November 7, 2017

Dear Stockholder:

We are pleased to invite you to attend the Special Meeting of Stockholders (the "Special Meeting") of General Moly, Inc. (the "Company"), which will be held at 9:00 a.m., local Colorado time, on December 15, 2017, at the Denver West Office Park, Building 22 Room 130, 1726 Cole Blvd., Lakewood, Colorado 80401. The meeting will be held to:

approve the issuance of shares of our common stock in connection with a transaction in which the total shares issued represent more than 20% of our outstanding common stock, and may be issued at a discount to the greater of book or market value of our common stock;

approve the ratification of equity awards granted since December 12, 2016, and

act on such other matters as may properly come before the meeting or any adjournment thereof.

Only stockholders of record on the books of the Company at the close of business on October 31, 2017, the record date fixed by the Board of Directors, are entitled to notice of and to vote at the Special Meeting and at any postponements or adjournments thereof. A complete list of stockholders entitled to vote at the Special Meeting will be available for inspection by stockholders during normal business hours at our corporate headquarters at 1726 Cole Boulevard, Suite 115, Lakewood, Colorado 80401 during the 10 days before our Special Meeting and at the Special Meeting.

It is important that your shares be represented at the Special Meeting regardless of the size of your holdings. Whether or not you expect to attend the Special Meeting, please complete, date and sign the enclosed proxy and return it in the enclosed postage paid return envelope, which does not require postage if mailed in the United States. If you choose to attend the Special Meeting, you may still vote your shares in person even though you have previously returned your proxy. If your shares are held in a bank or brokerage account, please refer to the materials provided by your bank or broker for voting instructions. The proxy is revocable at any time prior to its use.

Sincerely,

Michael K. Branstetter

Secretary IMPORTANT NOTICE REGARDING THE AVAILABILITY OF PROXY MATERIALS FOR THE SPECIAL STOCKHOLDER MEETING TO BE HELD ON DECEMBER 15, 2017

The Company's proxy statement and form of proxy card are available at: www.generalmoly.com.

TABLE OF CONTENTS

	Page
PURPOSE OF THE SPECIAL MEETING	<u>1</u>
Shares Outstanding and Voting Rights	<u>1</u>
VOTING SECURITIES AND	
PRINCIPAL HOLDERS	<u>4</u>
PROPOSAL 1 APPROVAL OF THE	
ISSUANCE OF SHARES OF OUR	
COMMON STOCK IN CONNECTION	
WITH A TRANSACTION IN WHICH	
THE TOTAL SHARES ISSUED	
REPRESENT MORE THAN 20% OF	
OUR OUTSTANDING COMMON	
STOCK, AND MAY BE ISSUED AT A	
DISCOUNT TO THE GREATER OF	
BOOK OR MARKET VALUE OF OUR	
COMMON STOCK	<u>7</u>
Execution of the Amer Transaction and	
Tranche 1 Closing	<u>7</u>
Later Amendments to the Purchase	
Agreement and Warrant	<u>8</u>
Third Amendment to the Purchase	
Agreement and Tranche 2 Closing	<u>8</u>
NYSE American Private Placement Rule	9
Effect on Existing Stockholders	9
Vote Required	9 9 9 9
Recommendation	9
PROPOSAL 2 RATIFICATION OF	
EQUITY AWARDS GRANTED SINCE	
DECEMBER 12, 2016	<u>10</u>
General	<u>10</u>
Requested Stockholder Approval	<u>11</u>
Vote Required	<u>12</u>
Recommendation	<u>12</u>
EXPLANATORY NOTE	<u>13</u>
COMPENSATION DISCUSSION AND	
<u>ANALYSIS</u>	<u>14</u>
Executive Summary	<u>14</u>
Executive Compensation Philosophy and	
<u>Objectives</u>	<u>16</u>
Our Executive Compensation Process	<u>17</u>
Elements of Compensation and 2016	
Compensation Decisions	<u>17</u>
Employment/Change of Control	
Agreements	<u>22</u>
Individual Executive Officers and the CEO	<u>22</u>
SUMMARY COMPENSATION TABLE	<u>23</u>
GRANTS OF PLAN-BASED AWARDS	<u>24</u>
Compensation Arrangements and	
Employment Agreements	<u>24</u>
OUTSTANDING EQUITY AWARDS AT	
DECEMBER 31, 2016	<u>25</u>
OPTION/SAR EXERCISES AND STOCK	
VESTED DURING 2016	<u>26</u>
	27

POTENTIAL PAYMENTS UPON			
TERMINATION OR CHANGE IN			
CONTROL			
Employment Agreements and Stay			
Agreements	<u>27</u>		
Change of Control Employment			
Agreements	<u>32</u>		
2006 Equity Incentive Plan	32 32		
Severance and Change in Control			
Payments	<u>32</u>		
POTENTIAL PAYMENTS UPON			
TERMINATION OR CHANGE IN			
CONTROL ON DECEMBER 31, 2016	<u>33</u>		
DIRECTOR COMPENSATION	33 33		
Director and Secretary Compensation			
<u>Program</u>	<u>35</u>		
COMPENSATION COMMITTEE			
INTERLOCKS AND INSIDER			
<u>PARTICIPATION</u>	<u>35</u>		
WHERE YOU CAN FIND MORE			
<u>INFORMATION</u>	<u>35</u>		
ADDITIONAL STOCKHOLDER			
<u>INFORMATION</u>	<u>36</u>		
Stockholder Proposals and			
Recommendations for Director Nominees			
for the 2018 Annual Meeting	<u>36</u>		
<u>Householding</u>	36 36 36 36		
Annual Report	<u>36</u>		
Other Matters	<u>36</u>		
ANNEX A INVESTMENT AND			
SECURITIES PURCHASE			
AGREEMENT, AS AMENDED	<u>A-i</u>		
		i	

General Moly, Inc.

1726 Cole Blvd., Suite 115 Lakewood, Colorado 80401

PROXY STATEMENT

Relating to Special Meeting of Stockholders To be held on December 15, 2017

We are sending this proxy statement to the holders of our common stock, \$0.001 par value, in connection with the solicitation by our Board of Directors (the "Board") of proxies to be voted at the General Moly, Inc. (the "Company," "we," or "us," or "our") Special Meeting of Stockholders (the "Special Meeting") to be held on December 15, 2017 at 9:00 a.m., local Colorado time, at the Denver West Office Park, Building 22 Room 130, 1726 Cole Blvd., Lakewood, Colorado 80401, and any postponements or adjournments thereof, for the purposes set forth in the accompanying Notice of Special Meeting of Stockholders. This proxy statement and the accompanying proxy card are first being mailed to our stockholders on or about November 7, 2017.

A proxy card is enclosed for your use. The Board requests that you sign, date, and return it in the enclosed postage paid return envelope, which does not require postage if mailed in the United States. Your execution of the enclosed proxy will not affect your right as a stockholder to attend the Special Meeting and to vote in person.

PURPOSE OF THE SPECIAL MEETING

At the Special Meeting, stockholders entitled to vote will be asked to consider and take action on the following matters:

approve the issuance of shares of our common stock in connection with a transaction in which the total shares issued represent more than 20% of our outstanding common stock, and may be issued at a discount to the greater of book or market value of our common stock;

approve the ratification of equity awards granted since December 12, 2016; and

act on such other matters as may properly come before the meeting or any adjournment thereof.

Your vote is important. We are requesting that you complete, sign and date the enclosed proxy card and mail it promptly in the enclosed postage paid return envelope, which does not require postage if mailed in the United States. Shares cannot be voted at the meeting unless the owner is present to vote or is represented by proxy.

Shares Outstanding and Voting Rights

Record Date; Quorum. Our Board has fixed the close of business on October 31, 2017, as the record date for the purpose of determining stockholders of the Company entitled to notice of and to vote at the Special Meeting. At the close of business on that date, we had 125,802,023 issued and outstanding shares of common stock. A majority of votes that could be cast by holders of all

Table of Contents

outstanding shares of stock entitled to vote will constitute a quorum for the transaction of business at the Special Meeting. Proxies that are submitted, whether voted for or against, abstentions, broker non-votes, or otherwise, on at least one item will be treated as present for all matters considered at the meeting, and will be counted for determining whether we have a quorum, however, broker non-votes are not deemed eligible to vote on items as to which they have no authorization to vote.

Solicitation of Proxies. The accompanying proxy is solicited on behalf of our Board and the entire cost of solicitation will be borne by us. Following the original mailing of the proxies and soliciting materials, our directors, officers and employees may solicit proxies by mail, telephone, facsimile or other electronic means of communication, or personal interviews. We will request brokers, custodians, nominees, and other record holders to forward copies of the proxies and soliciting materials to persons for whom they hold shares of the Company and to request authority for the exercise of proxies. In such cases, the Company will reimburse such holders for their reasonable expenses. The Company has retained Alliance Advisors LLC to perform proxy management services in preparation for the Special Meeting. Alliance Advisors' services will include consulting with General Moly regarding all aspects of proxy solicitation and management; and (b) if requested, contacting banks, brokers and proxy intermediaries to determine the quantity of documents needed in connection with the meeting, and distributing appropriate quantities of such documents. Fees for Alliance Advisors' services are \$9,090. General Moly may request additional services on an as needed basis.

If you have additional questions, need assistance in submitting your proxy or voting your shares of our Common Stock, or need additional copies of the Proxy Statement or the enclosed proxy card, please contact Alliance Advisors LLC.

Alliance Advisors LLC

200 Broadacres Drive, 3rd Floor, Bloomfield, NJ 07003 855-928-4487 Banks and Brokers Call: (973) 873-7700

Revocation of Proxy. Any proxy delivered in the accompanying form may be revoked by the person executing the proxy by either (1) providing our Corporate Secretary with a later-dated proxy prior to the Special Meeting or presenting a later-dated proxy at the Special Meeting, (2) providing our Corporate Secretary a written revocation prior to the Special Meeting, or (3) attending the Special Meeting and voting in person.

How Proxies will be Voted. Assuming a quorum is present, proxies received by our Board in the accompanying form will be voted at the Special Meeting as specified by the person giving the proxy. All shares represented by a valid proxy will be voted at the discretion of the proxy holders on any other matters that may properly come before the meeting. The Board, however, does not know of any matters to be considered at the meeting other than those specified in the Notice of Special Meeting.

Required Votes. The affirmative vote of the holders of a majority of the shares entitled to vote that are present in person or represented by proxy is required to approve, each of Proposals 1 and 2.

Effect of Abstentions and Broker Non-Votes. Abstentions may be specified and will be counted as present for the purposes of each of Proposals 1 and 2. For purposes of determining whether Proposals 1 and 2 have received the requisite vote, an abstention by a stockholder will have the same effect as a vote against the proposal.

Brokers and other intermediaries, holding shares in street name for their customers, are generally required to vote the shares in the manner directed by their customers. If their customers do not give any direction, brokers may vote the shares if (1) the broker holds the shares in a fiduciary capacity, or (2) the broker is acting pursuant to the rules of any national securities exchange of which it is a member. On certain routine matters, brokers may, at their discretion, vote shares on behalf of their

Table of Contents

customers. Each of Proposals 1 and 2 are considered a non-routine matter for which brokers are not permitted to vote shares without customer direction. Therefore, brokers are not permitted to vote shares for Proposals 1 or 2 without customer direction. *Therefore, we urge you to give voting instructions to your broker on both proposals*. Shares that are not voted by a broker given the absence of customer direction are called "broker non-votes." Broker non-votes will have no direct effect on whether any proposal is approved.

Voting Power. Holders of our common stock are entitled to one vote for each share held.

3

VOTING SECURITIES AND PRINCIPAL HOLDERS

The following table sets forth information as of October 31, 2017, regarding the ownership of our common stock by:

each person who is known by us to own more than 5% of our shares of common stock;

each of our named executive officers and directors; and

all of our current executive officers and directors as a group.

For the purposes of the information provided below, beneficial ownership is determined in accordance with the rules of the United States Securities and Exchange Commission (the "SEC"), and for each person includes shares of our common stock that person has the right to acquire within 60 days following October 31, 2017, upon exercise of options, stock appreciation rights or warrants.

Except as indicated in the footnotes to the tables below, and as affected by applicable community property laws, all persons listed have sole voting and investment power for all shares shown as beneficially owned by them.

We have no knowledge of any arrangements, including any pledge by any person of our securities, the operation of which may at a subsequent date result in a change in our control. We are not, to the best of our knowledge, directly or indirectly owned or controlled by another corporation or foreign government.

BENEFICIAL OWNERSHIP

	Amount and Nature of	Percent of
Name and Address of Beneficial Owner(1)	Beneficial Ownership	Class(2)
Stockholders Holding 5% or More:	27.07.470	22.207
AMER International Group Co., Ltd(3)	27,967,479	22.2%
Hanlong (USA) Mining Investment, Inc.		
Hanlong Resources Ltd		
Sichuan Hanlong Group Co., Ltd		
Geng Liu		
YiFan Liu		
XiaoPing Liu		
Xue Yang		
Nelson F. Chen(4)	11,843,341	9.4%
APERAM		
AMO Holding 7 S.A.(5)	8,256,699	6.6%
F. Steven Mooney(6)	10,000,000	7.4%
Executive Officers:		
Bruce D. Hansen(7)	5,205,869	4.1%
Robert I. Pennington(8)	1,298,962	1.0%
R. Scott Roswell(9)	490,252	*
Amanda Corrion	59,182	*
Directors (not including Chief Executive Officer):		
Ricardo M. Campoy	202,506	*
Mark A. Lettes	147,700	*
Gary A. Loving(10)	598,648	*
Gregory P. Raih(11)	195,000	*
Tong Zhang(3)	28,039,465	22.3%
Directors and executive officers as a group (9 persons)(12)	36,237,584	28.2%

Less than 1%.

- (1)
 The address for each of our directors and officers, other than Mr. Zhang, is c/o General Moly, Inc., 1726 Cole Blvd., Suite 115,
 Lakewood, Colorado 80401. The address for Mr. Zhang is 29/F, Block A, East Pacific International Center, 7888th Shenzhen Blvd.,
 Shenzhen, 518040, China.
- Based on 125,802,023 shares of our common stock outstanding as of October 31, 2017. In accordance with SEC rules, percent of class as of October 31, 2017, is calculated for each person and group by dividing the number of shares beneficially owned by such person or group by the sum of the total number of shares of our stock outstanding, plus the number of shares exercisable by that person or group within 60 days of October 31, 2017.
- AMER and Mr. Zhang share the power to vote, direct the vote, dispose and direct the disposition of all shares shown as beneficially owned by AMER. All of the voting and investment power with respect to shares held in the name of AMER have been delegated to Mr. Zhang. The address for both AMER and Mr. Zhang is 29/F, Block A, East Pacific International Center, 7888th Shenzhen Blvd., Shenzhen, 518040, China.
- Based on a Schedule 13D/A jointly filed with the SEC on March 10, 2014, by Hanlong (USA) Mining Investments, Inc. ("Hanlong USA") and a Form 4/A filed on February 12, 2013. All of the voting and investment power with respect to shares held in the name of Hanlong USA have been delegated to Mr. Chen. The address for Hanlong USA is Suite 6303-04, 63/F., Central Plaza, 18 Harbour Road, Wanchai, Hong Kong. The shares that are directly owned by Hanlong USA are

Table of Contents

also indirectly beneficially owned by each of Hanlong Resources, Sichuan Hanlong, Geng Liu, YiFan Liu, XiaoPing Liu, and Xue Yang. The Form 4/A does not indicate whether any of such persons exercises any power to vote, direct the vote, dispose or direct the disposition of the shares shown as indirectly beneficially owned by them. The addresses for each such person (other than Hanlong USA and Mr. Chen which are above) are: (a) in the case of Hanlong Resources, Suite 6303-04, 63/F, Central Plaza, 18 Harbour Road, Wanchai, Hong Kong; and (b) in the case of Sichuan Hanlong, Geng Liu, YiFan Liu Xiaoping Liu, and Xue Yang, 20F, Hongda Building, No. 2 East Jin Li Road, Chengdu, Sichuan 610041, China.

- Based on a Schedule 13G filed with the SEC on January 28, 2011, by APERAM and AMO Holding 7 S.A. and a Form 3 filed with the SEC on January 28, 2011, by APERAM. According to such Form 3, on January 25, 2011, the Board of Directors of ArcelorMittal S.A. ("ArcelorMittal") and APERAM each approved the transfer of the assets comprising ArcelorMittal's stainless and specialty steels business from its carbon steel and mining business to APERAM, a separate entity incorporated in the Grand Duchy of Luxembourg. Following such transfer, AMO Holding 7 S.A. became a wholly owned subsidiary of APERAM. APERAM and AMO Holding 7 S.A. share voting and disposition power for all shares shown as beneficially owned by them. The addresses for APERAM and AMO Holding 7 S.A., respectively, are 12C, rue Guillaume Kroll L-1882 Luxembourg, Grand Duchy of Luxembourg and 19, Avenue de la Liberté, L-2930 Luxembourg, Grand Duchy of Luxembourg.
- (6)
 Based on a Schedule 13G/A filed with the SEC on February 9, 2016, by Mr. Mooney. Includes 5,000,000 shares that would be received upon conversion of a Senior Convertible Note which is currently convertible and 5,000,000 shares that would be received upon exercise of a warrant which is currently exercisable.
- (7) Includes 750,000 shares that would be received upon conversion of a Senior Convertible Note that is currently convertible and 1,500,000 shares that would be received upon exercise of a warrant which is currently exercisable.
- Includes 145,000 shares of unvested performance-based restricted stock that was granted and previously reported on Form 4 but not yet issued, 158,000 shares held by Robert Pennington Dolores R. Pennington P/ADM Mineral Development LLC Dated 10/15/2007, of which Mr. Pennington is the sole member, and 150,000 shares that would be received upon exercise of a warrant which is currently exercisable.
- (9) Includes 13,260 shares held in Mr. Roswell's individual retirement account and 60,000 shares that would be received upon exercise of a warrant which is currently exercisable.
- (10)
 Includes 100,000 shares that would be received upon conversion of a Senior Convertible Note that is currently convertible and 200,000 shares that would be received upon exercise of a warrant which is currently exercisable.
- (11) Includes 35,000 shares held in Mr. Raih's individual retirement account.
- Includes 145,000 shares of restricted stock, 850,000 shares that would be received upon conversion of Senior Convertible Notes which are currently convertible and 2,010,000 shares that would be received upon exercise of warrants which are currently exercisable.

6

PROPOSAL 1

APPROVAL OF THE ISSUANCE OF SHARES OF OUR COMMON STOCK IN CONNECTION WITH A TRANSACTION IN WHICH THE TOTAL SHARES ISSUED REPRESENT MORE THAN 20% OF OUR OUTSTANDING COMMON STOCK, AND MAY BE ISSUED AT A DISCOUNT TO THE GREATER OF BOOK OR MARKET VALUE OF OUR COMMON STOCK

We are seeking approval of the issuance of certain shares of common stock to Amer International Group Co., Ltd. ("Amer"), under the Investment and Securities Purchase Agreement, as amended ("Purchase Agreement"), described below, for purposes of the Private Placement Rule (as defined below).

The transactions described in this subsection are further described in our periodic filings with the SEC, including our Current Reports on Form 8-K filed on April 21, 2015, December 1, 2015, April 18, 2017, June 20, 2017, July 18, 2017, August 10, 2017, October 2, 2017, and October 16, 2017. We refer you to those filings, and the documents filed therewith, and incorporate them by reference into this Proxy Statement. See "Where You Can Find Additional Information" below.

The discussion herein is qualified in its entirety by the full text of the Purchase Agreement, which is attached to this Proxy Statement as *Appendix A* and is incorporated herein by reference.

Execution of the Amer Transaction and Tranche 1 Closing

On April 17, 2015, we announced the execution of the Purchase Agreement with Amer, under which Amer agreed to (1) acquire an equity interest in the Company by purchasing shares of our common stock and warrants, and (2) assist the Company in obtaining a loan from one or more prime Chinese banks to fund our share of costs related to the development of the Mt. Hope Project (the "Loan"), of approximately \$700 million, including providing a guarantee of the Loan. Because the shares of common stock and warrants issuable to Amer under the Purchase Agreement were anticipated to exceed 20% of the Company's outstanding common stock at a price below the greater of book or market value of the Company's common stock on the date the Purchase Agreement was signed, the Company sought stockholder approval of the issuance of common stock and warrants to Amer. The Company's stockholders approved the issuances under the Purchase Agreement at the 2015 Annual Meeting of Stockholders held on June 30, 2015.

On November 2, 2015, the Company and Amer entered into an amendment to the Purchase Agreement, pursuant to which the parties agreed to a three tranche closing structure to the transactions contemplated by the Purchase Agreement, in lieu of the single tranche closing structure that had originally been contemplated under the Purchase Agreement.

On November 24, 2015, the parties closed the first tranche of investment under the Purchase Agreement (the "Tranche 1 Closing"), pursuant to which Amer invested \$4 million in the Company in exchange for 13,333,333 shares of the Company's common stock, at a price of \$0.30 per share, and warrants to purchase 80,000,000 shares of common stock at \$0.50 per share. The warrants were issued pursuant to a Common Stock Purchase Warrant executed by the Company and Amer (the "Warrant"). Pursuant to its original terms, the Warrant was exercisable at any time commencing on the date the Company first draws down funds under the Loan (the "Commencement Date"), and expiring on the earlier of (a) the sixty (60) month anniversary of the Commencement Date or (b) April 17, 2017 (the twenty-four (24) month anniversary of the Purchase Agreement), if the Loan documentation was not executed by that date.

In addition, at the Tranche 1 Closing, Amer and the Company entered into a Stockholder Agreement (the "Stockholder Agreement") pursuant to which Amer was permitted to immediately nominate one member of the Company's Board of Directors, as well as additional directors following the completion of the third tranche closing under the Purchase Agreement, and drawdown of a senior

Table of Contents

secured loan, respectively. Amer designated Tong Zhang to serve as a director of the Company pursuant to this provision. The Stockholder Agreement also governs Amer's acquisition and transfer of shares of the Company's common stock.

Also at the Tranche 1 Closing, Amer and the Company entered into an Expense Reimbursement Agreement (the "Expense Reimbursement Agreement") pursuant to which the Company agreed to deposit \$2 million into a joint account (the "Joint Account") to cover anticipated Mt. Hope financing costs and other jointly sourced business development opportunities.

Later Amendments to the Purchase Agreement and Warrant

As the April 17, 2017 expiration date of the Warrant approached, the Company and Amer discussed potential amendments to the agreements between them, supportive of their existing strategic partnership. While longer-term modifications were being negotiated, the parties entered into short-term amendments to the Warrant extending the deadline for satisfaction of all conditions to vesting of the Warrant. The Warrant amendments were executed on April 17, 2017 (extending the deadline to June 17, 2017), June 16, 2017 (extending the deadline to July 17, 2017) and July 16, 2017 (extending the deadline to August 17, 2017).

On August 7, 2017, the Company and Amer entered into a Second Amendment (the "Second Amendment") to the Purchase Agreement. The Second Amendment accelerates the closing of Tranche 2 of Amer's investment under the Purchase Agreement by removing certain conditions to closing related to minimum molybdenum prices and the reissuance of water permits for the Company's Mt. Hope Project in Nevada. Under the Second Amendment, Tranche 2 will consist of \$6,000,000 of shares of the Company's common stock, priced at the volume weighted average price for the 30-day period ending August 7, 2017, or \$0.41 per share, for a total of 14.6 million common shares. The Tranche 2 closing was originally expected to occur on or about September 30, 2017.

The Second Amendment also revised the conditions to the closing of Tranche 3 of Amer's investment. Tranche 3 will consist of \$10,000,000 of shares of the Company's common stock, priced at \$0.50 per share. The Tranche 3 closing will be conditioned on either (1) the completion by Company and Amer of a mutually agreed acquisition involving more than 10 million shares of the Company's common stock as consideration; or (2) the reissuance of the Mt. Hope water permits. Tranche 3 must close by the later of March 31, 2018 or 90 days after the earlier occurrence of one of the foregoing conditions.

Also on August 7, 2017, the Company and Amer entered into a fourth amendment to the Warrant, which extends the deadline for satisfaction of all conditions to vesting of the Warrant from August 17, 2017 to the third anniversary of the reissuance of the Record of Decision for the Mt. Hope Project.

Third Amendment to the Purchase Agreement and Tranche 2 Closing

Effective September 30, 2017, the Company and Amer entered into a Third Amendment (the "Third Amendment") to the Purchase Agreement. The Third Amendment extended the date to close Tranche 2 of Amer's investment under the Purchase Agreement to October 16, 2017, due to unanticipated process delays. All other provisions of the Purchase Agreement remained unchanged.

On October 13, 2017, the Company and Amer closed the second tranche (the "Tranche 2 Closing") under the Purchase Agreement. In the Tranche 2 Closing, Amer invested \$6 million in the Company in exchange for 14,634,146 shares of the Company's common stock, at a price of \$0.41 per share. As required by the Purchase Agreement, upon the Tranche 2 Closing, the Company deposited \$500,000 into the Joint Account.

NYSE American Private Placement Rule

Under Section 713(a)(ii) of the Company Guide of the NYSE American LLC (formerly known as the NYSE MKT), on which our common stock is traded, prior stockholder approval is required for the sale, issuance, or potential issuance of common stock (or securities convertible into common stock) equal to 20% or more of the outstanding common stock for less than the greater of book or market value of the stock (the "Private Placement Rule"). We previously obtained stockholder approval of the issuance of shares under the Purchase Agreement at the 2015 Annual Meeting of Stockholders held on June 30, 2015. However, the amendments to the Purchase Agreement have significantly altered the timing and conditions of Amer's investment. The 20 million shares of common stock issuable to Amer in Tranche 3 under the Purchase Agreement will be priced at \$0.50 per share, which may be less than the closing price of our common stock at the time of issuance, and which, together with all shares issued under the Purchase Agreement, will exceed 20% of our common stock outstanding. Accordingly, the Private Placement Rule may apply to the transactions contemplated by the Purchase Agreement as amended, and we are seeking your approval of Proposal 1 in order to ensure we have satisfied the requirements of the Private Placement Rule.

Effect on Existing Stockholders

Issuance of shares of common stock under the Purchase Agreement will have a dilutive effect on the ownership percentage of the Company's existing stockholders. If our stockholders approve Proposal 1, the transaction will comply with the NYSE American listing rules. However, significant conditions precedent to the closing of Tranche 3 remain, as described above. If we are able to satisfy the other conditions to closing of Tranche 3 and the Loan Agreement, we expect the development of the Mt. Hope Project will be substantially funded.

If our stockholders fail to approve the proposal, we may seek to restructure the Tranche 3 investment again, or we may decide to proceed with the closing of Tranche 3, as the application of the Private Placement Rule to amended Tranche 3 is unclear. If we proceed with the closing, we may face penalties, including the possibility of the delisting of our common stock by the NYSE American if it determines that the previous stockholder approval of the Purchase Agreement no longer applies to Tranche 3 as amended, and therefore that we have violated the Private Placement Rule. If our common stock is delisted from the NYSE American, the liquidity of your investment in our common stock will be negatively affected, and you may find it difficult to sell your shares at an attractive price or at all. The Company may also face increased difficulty in obtaining financing, including the Loan, and in hiring and retaining executive officers and employees.

The Board believes it is in the best interest of the Company to issue shares of common stock to Amer under Tranche 3 of the Purchase Agreement. The Board expects that the Loan Agreement that is a part of the Amer transaction will further our efforts to complete construction of the Mt. Hope Project and commence commercial production. In addition, the future exercise of the Warrant for cash will result in additional proceeds to the Company.

Vote Required

The approval of the issuance of shares of our common stock in connection with a transaction in which the total shares issued represent more than 20% of our common stock currently outstanding, at a price that may be less than the greater of book or market value of our common stock, requires the affirmative vote of a majority of the total votes cast on the proposal at the Special Meeting, either in person or by proxy. Abstentions will have the effect of a vote against the proposal. Broker "non-votes" will have no effect with respect to the proposal.

Recommendation

The Board of Directors unanimously recommends that stockholders vote *FOR* Proposal 1. If not otherwise specified, proxies will be voted *FOR* Proposal 1.

PROPOSAL 2 RATIFICATION OF EQUITY AWARDS GRANTED SINCE DECEMBER 12, 2016

We are seeking ratification by our stockholders of equity awards granted under our 2006 Equity Incentive Plan, as amended and restated (the "2006 Plan"), after December 12, 2016, including awards to directors and executive officers. All such awards were timely disclosed in the appropriate periods in our periodic reports on Form 10-Q and Form 10-K, in our annual proxy statements and in each recipient's Section 16 filings, as applicable.

General

The 2006 Plan was approved by our Board on February 25, 2010 and by our stockholders on May 13, 2010. An additional amendment to the 2006 Plan, increasing the number of shares that may be issued thereunder, was approved by our Board on March 13, 2016 and by our stockholders on June 8, 2016. The 2006 Plan authorizes the Board, or a committee of the Board, to issue or transfer up to an aggregate of 14,600,000 shares of common stock, of which 5,043,019 remain available for issuance as of the Record Date. Awards under the 2006 Plan may include incentive stock options, non-statutory stock options, restricted stock units, restricted stock awards, and stock appreciation rights ("SARs"). At the option of the Board, SARs may be settled with cash, shares, or a combination of cash and shares. The Company settles the exercise of other stock-based compensation with newly issued common shares.

Section 14(a) of the 2006 Plan provides that the 2006 Plan shall terminate "on the day before the tenth anniversary of the date the Plan is adopted by the Board or approved by the shareholders of the Company, whichever is later." The 2016 amendment did not modify the 2006 Plan's termination date. Accordingly, the term of the 2006 Plan will expire on May 12, 2020, the tenth anniversary of the date on which it was approved by our stockholders. The full text of the 2016 Plan was attached as *Annex A* to the Company's 2010 Proxy Statement. However, the 2010 Proxy Statement, in summarizing the proposed 2006 Plan, also stated that no stock awards would be made under the 2006 Plan after December 12, 2016, a reference to the tenth anniversary of the original adoption of the 2006 Plan. This statement, along with the rest of the summary of the proposed 2006 Plan in the 2010 Proxy Statement, was qualified in its entirety by reference to the text of the 2006 Plan, including its statement that the 2016 Plan would terminate on the day before the tenth anniversary of its date of adoption by the Company's stockholders, which is May 12, 2010. Nonetheless, to avoid any doubt concerning the termination date of the 2006 Plan, the Company hereby supplements its prior disclosures to state that the 2006 Plan will expire on May 12, 2020 and that the Company may grant awards under the 2006 Plan until that date.

Between December 13, 2016 and the Record Date, the Compensation Committee of the Board approved stock and restricted stock unit awards covering an aggregate of 1,435,000 shares of common stock pursuant to and in accordance with the 2006 Plan. The grants were made to officers, directors and employees of the Company. The grants of these awards (a) was consistent with the corporate powers granted in the Company's Certificate of Incorporation, (b) satisfied the requirements of the Delaware General Corporation Law, and (c) complied with the terms of the 2006 Plan. Below is a

Table of Contents

chart detailing the grants that were made in 2017 to executive officers, directors and non-officer employees of the Company:

Name and Position	Grant Date	Number of Shares
Bruce D. Hansen	1/16/2017	360,000(1)
President and Chief Executive Officer		
Robert I. Pennington	1/16/2017	300,000(1)
Chief Operating Officer		
R. Scott Roswell	1/16/2017	240,000(1)
Chief Legal Officer		
Amanda Corrion(2)	1/16/2017	20,000(1)
Principal Accounting Officer		
Ricardo M. Campoy	1/4/2017	25,000(3)
Chairman of the Board		
Mark A. Lettes	1/4/2017	25,000(3)
Director		
Gary A. Loving	1/4/2017	25,000(3)
Director		
Gregory P. Raih	1/4/2017	25,000(3)
Director		
Tong Zhang	1/4/2017	25,000(3)
Director		
Non-Executive Officer Employee Group	1/16/2017	390,000

- Represents a grant of restricted stock units that vest on the earliest to occur of (i) a financing plan for the Mt. Hope Project approved by the Company's Board; (ii) a Change of Control (as defined in the executive officer's employment agreement with the Company); (iii) involuntary termination (absent cause); or (iv) January 16, 2018. Each restricted stock unit represents a contingent right to receive one share of the Company's common stock. Vested shares will be delivered to the reporting person within sixty (60) days, except as otherwise required by Section 409A of the Internal Revenue Code of 1986, as amended.
- (2) Ms. Corrion became Principal Accounting Officer on May 12, 2017.
- (3) Represents a grant of fully vested common stock.

Requested Stockholder Approval

The stock awards to directors in January 2017 were made pursuant to the Company's regular annual Director Compensation Program. As a smaller public company with limited cash resources, equity compensation is essential to our ability to attract and retain qualified and experienced director candidates, especially during the duration of our cash conservation program as we continue to seek full financing for construction of the Mt. Hope Project.

The grants to executive officers and employees during 2017 were made pursuant to the retention program adopted as a part of the Company's cost reduction program, initially implemented in 2015 and renewed annually in each year since. The cost reduction program includes base salary reductions to our executive officers as part of our cash conservation efforts, and the suspension of equity awards or annual cash incentives to executive officers other than equity stay incentives granted under the stay incentive agreements entered into between the Company and each executive officer.

We believe that the stay incentive agreements are necessary to retain our executives who are experienced in project financing, mine development, and operations to finance and develop the

Table of Contents

Mt. Hope Project, grow the Company, and increase our stockholder value. In establishing the agreement with each executive officer, our Compensation Committee took into account many factors, including the individual's retention considerations, prior business experience, historical compensation levels, work performance, and our business need for the executive's skills. The committee also considered external market data, market trends, and drew upon the individual experience of the committee members. Equity awards historically have been a significant element of our executive compensation program, and critical to the ongoing retention of our executives.

We also believe that the equity grants to our non-officer employees are a critical component of our efforts to retain them duration of our cash conservation program and efforts to obtain full financing for the Mt. Hope Project.

In Proposal 2, the Company is requesting the stockholders approve the ratification of all equity awards granted by the Company since December 12, 2016. We believe that the equity awards were duly authorized and validly issued under Delaware law, compliant with the terms of the 2006 Plan and consistent with applicable NYSE American listing standards governing stockholder approval of equity compensation plans. However, the Board has determined that, out of an abundance of caution, stockholder ratification of these awards is in the best interests of the Company and our stockholders.

If our stockholders do not ratify the equity awards listed above, we believe that outcome may have an adverse effect on our ability to retain the directors and executive officers who received such awards.

Vote Required

The approval of the ratification of equity grants since December 12, 2016, requires the affirmative vote of a majority of the total votes cast on the proposal at the Special Meeting, either in person or by proxy. Abstentions will have the effect of a vote against the proposal. Broker "non-votes" will have no effect with respect to the proposal.

Recommendation

The Board recommends that stockholders vote *FOR* Proposal 2. If not otherwise specified, proxies will be voted *FOR* Proposal 2.

EXPLANATORY NOTE

The following Compensation Discussion and Analysis, Executive Compensation, Director Compensation and Compensation Committee Interlocks disclosure was previously included by the Company in its definitive proxy statement for the 2017 Annual Meeting of Stockholders on April 14, 2017. The Company has included this disclosure in this proxy statement for the Special Meeting as required by Schedule 14A and related rules of the SEC. The content of this disclosure is identical to the content of the corresponding disclosure included in the definitive proxy statement for the 2017 Annual Meeting, does not reflect events occurring after the date of the definitive proxy statement for the 2017 Annual Meeting, and does not modify or update disclosures that may have been affected by subsequent events.

The Company notes that the following events have occurred since the filing of the definitive proxy statement for the 2017 Annual Meeting, each of which has been publicly disclosed in the Company's filings with the SEC.

Record of Decision

As previously disclosed, to resolve the issues identified by the Ninth Circuit in connection with the vacated Record of Decision for the Mt. Hope Project, the BLM determined to prepare a Supplemental Environmental Impact Statement ("SEIS"). The SEIS will disclose additional information to the public related to the selection of appropriate background concentrations to use for dispersion modeling of air pollutants. Because the SEIS must be prepared in accordance with the NEPA guidelines, the SEIS process will include three publications in the Federal Register, each of which may take several weeks to process. The first of these publications is the Notice of Intent ("NOI") which declares the BLM's intent to prepare the SEIS. The NOI was published in the Federal Register on July 19, 2017. With publication of notice announcing preparation of a SEIS, we are working with the BLM to complete the draft SEIS and participating with the necessary public review to receive a new ROD authorizing the eventual construction and operation of the Mt. Hope Project.

Water Rights

On September 29, 2017, the Company announced that the Nevada Supreme Court issued its Opinion affirming the Nevada District Court's March 2016 Order which vacated the water permits for the Mt. Hope Project. The Company intends to move forward with water applications currently pending before the Nevada State Engineer for the development and operation of the Mt. Hope Project.

AMER International Group

On October 13, 2017, the Company and Amer completed the Tranche 2 Closing under the Purchase Agreement. In the Tranche 2 Closing, Amer invested \$6 million in the Company in exchange for 14,634,146 shares of the Company's common stock, at a price of \$0.41 per share. As required by the Purchase Agreement, upon the Tranche 2 Closing, the Company deposited \$500,000 into the Joint Account.

Under the Purchase Agreement, as amended, Tranche 3 of Amer's investment will consist of \$10,000,000 of shares of the Company's common stock, priced at \$0.50 per share. The Tranche 3 closing will be conditioned on either (1) the completion by Company and Amer of a mutually agreed acquisition involving more than 10 million shares of the Company's common stock as consideration; or (2) the reissuance of the Mt. Hope water permits. Tranche 3 must close by the later of March 31, 2018 or 90 days after the earlier occurrence of one of the foregoing conditions.

Under the Warrant, as amended, which the deadline for satisfaction of all conditions to vesting of the Warrant has been extended to the third anniversary of the reissuance of the Record of Decision for the Mt. Hope Project.

COMPENSATION DISCUSSION AND ANALYSIS

This Compensation Discussion and Analysis provides information about our executive compensation program. It describes the philosophy and objectives of our executive compensation program and how we applied those objectives in compensating our executive officers during 2016. For 2016, our "named executive officers," or NEOs, included the following individuals:

Bruce D. Hansen, Chief Executive Officer or CEO;

Lee M. Shumway*, Chief Financial Officer or CFO;

Robert I. Pennington, Chief Operating Officer or COO; and

R. Scott Roswell, Chief Legal Officer or CLO.

Our executive team is key to the Company's achievement of its business strategy. Our executives were carefully selected, and retained, as a result of their significant experience in mine development, project financing, and operations, to lead the implementation of our business strategy.

Executive Summary

Our Business Strategy

We are in the business of the exploration, development and future mining of properties containing molybdenum. Our business strategy is to acquire and develop highly profitable advanced stage mineral deposits. Our principal asset is an 80% interest in the Mt. Hope Project, a primary molybdenum property located in Eureka County, Nevada, operated by Eureka Moly, LLC ("EMLLC"). EMLLC is a joint venture of General Moly, through its wholly owned subsidiary Nevada Moly, LLC (80% membership interest) and POS-Minerals Corporation ("POS-Minerals") (20% membership interest), a division of POSCO, a large Korean steel company. We also have a second significant wholly owned molybdenum and copper project, the Liberty Property, located in Nye County, Nevada.

In the near-term, our objective is continue our cash conservation efforts as we look toward completing efforts to reobtain water permits, the Record of Decision approving a supplement to our Environmental Impact Statement from the Bureau of Land Management ("BLM"), and seeking and obtaining project financing to construct and operate the Mt. Hope Project; to evaluate acquisition and other business opportunities that we can collectively pursue with our strategic partner AMER International Group, as identified below; and to continue our evaluation of the molybdenum and copper properties at the Liberty Project.

Developments During 2016

AMER International Group

The Amended Investment and Securities Purchase Agreement and other agreements between AMER International Group ("AMER") and the Company remain in place. We closed the first tranche of AMER's \$20 million investment in November 2015 for a \$4 million investment purchasing 13.3 million shares, and the issuance of warrants to purchase 80 million shares at \$0.50, which become exercisable upon the completion of a \$700 million dollar bank loan to fund our EMLLC membership share of construction costs for the Mt. Hope Project. In 2016, hampered by continuing stagnant molybdenum prices, the Company worked with its strategic partners at AMER to identify and explore potential value accretive opportunities to the Company concerning base metal projects in North America. This process continues into 2017. We remain committed to our strong relationship with AMER, including our association with Tong Zhang, nominated by AMER to serve on our Board of

^{*}

Table of Contents

Directors. Mr. Zhang works closely with our management group and our financial advisors as we review such opportunities.

Cash Conservation

Now in our fourth year of cash conservation, our executives continue to work aggressively to manage expenses and preserve liquidity while we seek opportunities to support financing options for the Mt. Hope Project, and evaluate potential business development opportunities with AMER. Our cash conservation efforts have reduced planned expenditures to approximately \$1.5 million a quarter for corporate and Liberty Project spend.

The Company continues cash conservation efforts to reduce planned expenditures that maximize our financial flexibility and ensures readiness to seek and establish financing for the restart of construction activities at the Mt. Hope Project. The continuation of the Company's cash conservation programs maintains current liquidity by reducing engineering, administrative and procurement expenses, and trimming our G&A expenditures to approximately \$1.5 million per quarter.

During 2016 at the Mt. Hope Project, we worked closely with the BLM and our reclamation surety underwriters to reduce the cash collateral component of our financial guarantee bonding requirement for reclamation responsibilities during this period of continuing care and maintenance. We successfully reduced the approved three- year disturbance plan from \$75.3 million to \$2.8 million, allowing for a saving of \$4.3 million in bonding and collateral costs Coupled with other cash management efforts, we have successfully extended the EMLLC Reserve Account to fund ongoing care and maintenance costs for the Company and POS-Minerals through 2021.

The efforts of our executive officers have been instrumental in managing ongoing cost reduction programs to maintain continuity of employees and liquidity during efforts to secure project financing for the Mt. Hope Project. The Company feels that the management of its liquidity, retention of key personnel and cooperation from our vendor partners is critical to maintaining the Mt. Hope Project as one of the world's best and largest undeveloped molybdenum projects.

Permitting Update

As discussed more fully in our annual and quarterly reports, the ongoing legal challenges of the Record of Decision ("ROD") approving our Plan of Operations and Environmental Impact Statement ("EIS") has been on appeal to the U.S. Court of Appeals for the Ninth Circuit ("Ninth Circuit") since August of 2014 when the U.S. District Court dismissed the claims of Great Basin Resource Watch and the Western Shoshone Defense Project ("Plaintiffs"). The Ninth Circuit completed oral argument of the parties on October 18, 2016. On December 28, 2016, the Ninth Circuit issued its Opinion rejecting many of the arguments raised by the Plaintiffs challenging the EIS completed for the Mt. Hope Project, but issued a narrow reversal of the BLM's findings related to air quality analysis. Because of this technical deficiency, the Court vacated the ROD, and is requiring additional evaluation of air quality impacts and resulting cumulative impact analysis under the National Environmental Policy Act ("NEPA") before the ROD is reissued. The Company is confident in the BLM's process and working closely with the agency to resolve concerns with air quality baseline studies raised by the Ninth Circuit, and looks forward to completing the necessary public review to receive a new ROD for the eventual construction and operation of the Mt. Hope Project.

Water Rights Considerations

Again, as discussed more fully in our Annual Report on Form 10-K and Quarterly Reports, following the October 2015 Opinion of the Nevada Supreme Court, we received the District Court's Order on remand from the Nevada Supreme Court on March 14, 2016 vacating the Monitoring, Management and Mitigation Plan ("3M Plan"), denying the water applications and vacating the water

Table of Contents

permits issued by the State Engineer in Ruling 6127. The Nevada State Engineer filed an appeal to the Nevada Supreme Court concerning the District Court's interpretation of the Supreme Court's Opinion arguing that the District Court acted in excess of its judicial authority in violation of Nevada's Constitution and Statutes. The Company has filed a similar appeal to the Nevada Supreme Court.

Notwithstanding the pendency of the appeal to the Nevada Supreme Court discussed above, the Company is working, as expeditiously as possible, to reobtain its water permits with the new change applications that it has filed with the State Engineer in the fall of 2015 and spring of 2016. The State Engineer set a prehearing conference on August 25, 2016 to review the applications and set a determination hearing. On August 23, 2016, Eureka County filed a Writ of Prohibition or Mandamus to the Nevada Supreme Court requesting the Supreme Court to intervene and to stop further action by the State Engineer while the appeal discussed above of the District Court Order is pending. The State Engineer vacated the conference and has stopped any further action on our applications pending an outcome of the Writ from the Supreme Court. The Supreme Court ordered answers and responses to the Writ, which were filed on October 21, 2016. Along with the State Engineer, our briefs oppose the basis for filing the Writ, and believe the State Engineer can proceed with the review of our applications notwithstanding the appeal to the Supreme Court of the District Court Order. In the interim, we are continuing to advocate the authority of the State Engineer to act on our applications. In hearings to be held before the State Engineer, the Company will provide additional evidence of its ability to successfully mitigate any potential impacts to water rights in Kobeh Valley that could result from the Mt. Hope Project's new change applications for water use.

Executive Compensation Philosophy and Objectives

Because of our modest size, stage of development and our current cash conservation efforts, we do not use an extensive executive compensation program. Instead, we have continued to utilize a fairly simple executive compensation program that is intended to provide appropriate incentives for our executive officers to help us achieve our business strategy. Essential to our compensation philosophy is the avoidance of egregious or overly generous compensation, excessive perquisites or tax gross ups on perquisites, repricing or replacement of stock awards, and hedging of Company stock. Our executive compensation program historically has used three primary elements: base salary, annual cash incentives, and long-term equity incentives, which are divided between performance based and time/retention based equity incentives. The overall objective of our program is to enable us to obtain and retain the services of experienced executives.

In 2016, we continued base salary reductions to our executive officers as part of our cash conservation program. Further, as discussed below, we did not grant any equity awards or cash incentives to our named executive officers, except for the retention awards of restricted stock units to our named executive officers. A summary of each of the Stay Incentive Agreements is included following the executive compensation tables under the heading "Potential Payments Upon Termination or Change in Control." We believe that these agreements are necessary to retain our executives who are experienced in project financing, mine development, and operations to finance and develop the Mt. Hope Project, grow the Company, and increase our stockholder value. In establishing the agreement with each executive officer, our Compensation Committee took into account many factors, including the individual's retention considerations, prior business experience, historical compensation levels, work performance, and our business need for the executive's skills. The committee also considered external market data, market trends, and drew upon the individual experience of the committee members.

When project financing is obtained to construct and operate the Mt. Hope Project, we anticipate that we will reinstate pre-reduction base salaries and positive adjustments to the compensation packages for our executive officers. In the interim, the committee continues to balance cost reductions with a compensation program that is intended to promote teamwork as well as individual initiative and

Table of Contents

achievement; to enhance stockholder value by aligning the financial interests of our executive officers with those of our stockholders; and to incent the retention of executives whose knowledge, skills and performance are critical to our success.

Our Executive Compensation Process

Role of Compensation Committee and Executive Officers

Our Compensation Committee has overall responsibility for (1) establishing, overseeing and evaluating the compensation philosophy, policies and plans for non-employee directors and executive officers, (2) making recommendations to the Board regarding director compensation and (3) reviewing the performance and determining the compensation of our CEO and the other executive officers. The committee oversees the administration of our equity incentive plans, reviews and approves any employment, severance or change in control agreements and performs other functions set forth in its charter.

In carrying out its responsibilities, the committee works with members of our management team, including our CEO, and consults with legal counsel and has consulted with independent compensation consultants, when it deems appropriate. The management team assists the committee by providing information on Company and individual performance, market data and management's perspective and recommendations on compensation matters. Although the committee solicits and reviews management's recommendations, the committee considers management's recommendations as merely one factor in making compensation decisions for our executive officers. The committee regularly reports to, and sometimes consults with, our Board on the results of its reviews and any actions it takes or proposes to take with respect to compensation policies and executive officer compensation decisions.

Role of Compensation Consultant

As a result of ongoing cost reduction efforts implemented in the third quarter of 2013 and continuing throughout 2016, the Committee did not retain or use an outside compensation consultant, as no changes were made to the named executives' compensation throughout 2016. At the expiration of cost reduction programs, the Compensation Committee will re-evaluate the retention of a compensation consultant, including our previous compensation consultant Towers Watson (2008, 2010 - 2013), for future engagement.

Peer Group

Historically, one of the purposes of the committee in hiring a compensation consultant, including the historical retention of Towers Watson, was to assist the committee in comparing our executive compensation program with executive compensation programs of peer companies. In 2012, the committee, with the assistance of Towers Watson, selected designated peer group companies consisting of North American companies primarily engaged in the hard rock mining of metals and coal mining, as well as other general industry companies, to use for comparison. With the implementation of the cost reduction program in September 2013 and continuing throughout 2016, the committee did not complete an analysis to update the 2012 peer group. The committee will review the 2012 peer group analysis and update the benchmark analysis at the expiration of the cost reduction program, if appropriate.

Elements of Compensation and 2016 Compensation Decisions

Our compensation program has three primary elements: base salary, annual cash incentive awards and long-term equity-based incentives. Our executive officers also participate in employee benefits that are generally available to all of our employees. Each of these primary elements is discussed in further detail below.

Table of Contents

Base Salary

Base salary represents the fixed portion of our executive officers' compensation and is an important element of compensation to attract, retain and motivate experienced executives. We establish our executives' salaries based on consideration of, among other things:

Performance and experience,

Scope of their responsibilities,

Competitive market compensation data for similar positions,

Seniority of the individual, and

Ability to replace the individual.

The committee reviews base salaries annually and makes adjustments from time to time. An adjustment to an executive's salary may be made, for example, to align that salary with the committee's perception of market levels, taking into account the individual's responsibilities, performance and experience. From 2006 through 2012, the committee made periodic adjustments to some executive salaries to bring the salaries closer to amounts the committee believed more closely reflected salaries paid to individuals in operating companies with similar positions and responsibilities. Beginning in 2013 and continuing through present, as a result of implementation of cost reduction programs, the committee temporarily reduced NEOs' salaries effective September 7, 2013 - January 2015, and in January 2015 re-instated the 2012 established base salaries.

In December 2015, the committee also approved the promotion of Lee Shumway to CFO, and authorized a 2016 salary increase to \$275,750, and also approved a 2016 salary increase for Scott Roswell to \$265,750 along with his title change to CLO inclusive of his additional responsibilities associated with the resignation of our former Director of Investor Relations. Additionally, at the same December 2015 meeting, the committee accepted the recommendation of management to re-institute the 2013 temporary salary reductions discussed above to be effective January 16, 2016. Again, at its December 2016 meeting the committee, at the recommendation of management, approved the continuation of salary reductions for 2017.

	2012 Annual ase Salary	and 20 201 Appro Base Sa	3 - 2014 1 2016 - 2017 2014 Reduc proved Annual e Salary Base Salar		nual Salary	2015 Annual			16 - 2017 Reduced mual Base Salary
Name	(\$)	Reduc	tions	(\$)		(\$)		(\$)
Bruce D. Hansen	\$ 550,000		(25)%	\$ 4	112,500	\$	550,000	\$	412,500
Lee M. Shumway*	234,350		(15)%	1	199,198		234,500		234,388
Robert I. Pennington	297,000		(20)%	2	237,600		297,000		237,600
R. Scott Roswell	250,700		(15)%	2	213,095		250,700		225,888

Mr. Shumway has announced his departure from the Company effective May 12, 2017.

Annual Incentive Awards

Historically, our executive officers have had the opportunity to earn annual incentive awards in the form of a cash incentive award for achievement of corporate and individual goals and objectives. Annual incentive awards have traditionally been paid to executive officers to recognize specific accomplishments and overall performance, as determined by the committee in its discretion.

Although we target annual cash pay, the committee retains full discretion to adjust annual incentive awards based on its collective judgment of the CEO's and executives' achievement of business milestones and individual objectives. For 2016, all potential grants of annual incentive awards for our

18

Table of Contents

named executive officers were determined by the committee in December 2015, in its discretion, based on achievement of the following business and individual objectives:

Corporate Business Goals	Weight
1. Liquidity & Financing	30%
2. Strategic/Alternative Opportunities	20%
3. Engineering and Construction	15%
4. Permitting, Environmental & Water Rights	20%
5. Safety & Health	10%
6. Administration	5%

As a result of the ongoing efforts to obtain project financing for the Mt. Hope Project and continuing cash conservation efforts the committee reviewed the business goals listed above, and determined as a result of the continuation of cost reduction programs to again not grant any 2016 cash incentive awards. As a result 2016 actual total annual cash compensation is as follows:

Name	2016 Reduced Base Salary (\$)		2016 Annual Incentive Award (\$)		2016 Cash Stay Incentive Award (\$)			2016 Annual Cash Compensation (\$)		
Bruce D. Hansen	\$	412,500	\$		0	\$	C)	\$	412,500
Lee M. Shumway*		234,888			0		C)		234,888
Robert I. Pennington		237,600			0		C)		237,600
R. Scott Roswell		225,888			0		C)		225,888

Mr. Shumway has announced his departure from the Company effective May 12, 2017.

Long-Term Equity Incentives

As a company with limited financial resources, long-term equity awards are historically a significant element of our executive compensation program, and critical to the ongoing retention of our executives. No equity awards were granted in 2016 as a result of ongoing cash conservation efforts, except for the retention awards of restricted stock units to our named executive officers as discussed below.

Executive Compensation for 2016

The Compensation Committee formally reviews all aspects of the executive compensation program throughout the year and has the authority to make adjustments based on its collective judgment. When considering adjustments to the executive compensation program, the Compensation Committee takes into account the following factors during its decision making process:

- (1) Company performance;
- (2) Executive compensation and governance best practices;
- NEOs achievement of the Company's annual and long-term business milestones and individual performance objectives, and
- (4) Shareholder feedback via "Say on Pay" voting results.

As a result of the support that the Company's 2011 - 2016 "Say on Pay" proposals received from stockholders, the Company's compensation policies and decisions remained consistent with our objectives to enhance stockholder value by aligning the financial interests of our executive officers with those of our stockholders and to reward our executives when they have achieved our business objectives. In large part, our executive compensation decisions for 2016 were hampered by the ongoing difficulty of efforts to seek and obtain project financing with depressed molybdenum prices and reversal

Table of Contents

of our water permits, the December 2016 court ruling vacating the ROD, the ongoing suspension of pre-construction activities at the Mt. Hope Project, and the necessity to continue cost reductions and further base salary compensation reductions:

No 2016 Adjustments to Base Salary. Back in 2012, our NEO salaries ranged from 90% to 95% of the median base salary for our peer group, and as a result, the Compensation Committee did not approve any base salary increases for 2013. And, with the implementation of the cost reduction programs beginning in 2013 and continuing into 2017 discussed above, our NEOs have again received 15% - 25% reductions in base cash salary compensation. Minor adjustments to base salaries were made in mid-January 2016, discussed above, related to the internal selection and promotion of Lee Shumway to CFO, and additional responsibilities placed on our CLO, Scott Roswell. Mr. Shumway announced his departure from the Company effective May 12, 2017.

No 2016 Annual Cash Incentive Awards. With the continuation of the Company's cost reduction program, no cash incentive awards were made to our NEOs for 2013, 2014, 2015 or 2016 results, except for the 2014 cash retention awards to our named executive officers discussed below under "Stay Incentive Agreements."

<u>No 2016 Performance Based Equity Awards.</u> In December 2016, the Compensation Committee again determined not to re-institute the prior annual practice of granting equity based performance awards, as a result of continuing cost reductions. During 2013, 2014, 2015 and again in 2016, there was no annual equity component to our executive compensation program, except for the retention awards of restricted stock units to our named executive officers discussed below under "Stay Incentive Agreements."

Stay Incentive Agreements

2013-2015 Stay Incentive Agreements In September 2013, we implemented a cost reduction and retention program that included a 25% reduction in base cash compensation for the CEO and members of the Board of Directors with other senior officer and employees receiving 10 to 20% salary reductions. In parallel, the Company also implemented a personnel stay incentive and equity award program for the NEOs and other senior managers that provided cash and equity incentives for employees who remained with the Company until the earlier to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors, a Change of Control (as defined in the employment or change of control agreements between the Company and each of our named executive officers); involuntary termination (absent cause); or January 15, 2015

Effective January 15, 2015, the following cash retention awards were paid, with the exception of Mr. Hansen who agreed to defer payment of his cash retention award to January 15, 2016, and the following RSU grants vested and shares were issued to our executive officers: Mr. Hansen \$412,500 (deferred to 1/15/16) and 245,536 shares; Mr. Shumway \$117,175 and 69,747 shares; Mr. Pennington \$148,500 and 88,393 shares; and Mr. Roswell \$125,350 and 74,613 shares.

2015 Stay Incentive Agreements In January 2015, in conjunction with our decision to terminate the salary reductions to base salaries and again freeze base salaries at the 2012 level, we entered into Stay Incentive Agreements with certain employees, including our executive officers, in order to provide an incentive for each individual to continue his employment with the Company through the critical phase of seeking and obtaining financing for, and the construction of, the Mt. Hope Project. The Stay Incentive Agreements provide for the award of RSUs if the covered executive remained continuously employed by the Company until the earlier to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors, a Change of Control (as defined in the employment or change of control agreements between the Company and each of our named executive officers); involuntary termination (absent cause); or January 15, 2016.

Table of Contents

Effective January 15, 2016, the following RSU grants vested and shares were issued to our executive officers: Mr. Hansen 392,904; Mr. Shumway 167,413; Mr. Pennington 212,168; and Mr. Roswell 179,093. Mr. Hansen also received his \$412,500 cash retention award that he had deferred from 2015.

2016 Stay Incentive Agreements In December 2015, with the continuation of cash conservation, ongoing suspension of development activities at our projects resulting from the depressed molybdenum market and other factors, the compensation committee reviewed our compensation program for 2016, and authorized a return of the 2013/2014 salary reductions to our named executive officers. Further, a similar Stay Incentive Program, which included a greater than a 50% reduction in the amount of equity incentives offered in the 2015 Stay Incentive Program discussed above, was authorized by the committee and implemented on January 16, 2016 for our named executive officers who remained with the Company through the earliest to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors, a Change of Control (as defined in the employment or change of control agreements between the Company and each of our named executive officers); involuntary termination (absent cause); or January 16, 2017.

Effective January 15, 2017, the following RSU grants vested and shares were issued to our executive officers: Mr. Hansen 120,000; Mr. Shumway 80,000; Mr. Pennington 100,000; and Mr. Roswell 80,000.

2017 Stay Incentive Agreements In December 2016, with the continuation of cash conservation, ongoing suspension of development activities at our projects resulting from the depressed molybdenum market and other factors, the compensation committee reviewed our compensation program for 2017, and approved continuation of the 2016 salary reductions to our named executive officers. Further, a new Stay Incentive Program was authorized by the committee and implemented on January 16, 2017 for our named executive officers who remain with the Company through the earliest to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors, a Change of Control (as defined in the employment or change of control agreements between the Company and each of our named executive officers); involuntary termination (absent cause); or January 16, 2018.

Effective with the execution of Stay Incentive Agreements on January 16, 2017 the following RSU grants were made to our executive officers, subject to the vesting schedule described above: Mr. Hansen 360,000; Mr. Pennington 300,000; and Mr. Roswell 240,000. Mr. Shumway has announced his departure from the Company, effective May 12, 2017.

	2016 -				
	2017 Approved				
	Base Salary Reductions	2017 Annual	2016 - 2017 Reduced Annual	2016 Stay Incentive	2017 Stay Incentive
Name	(\$)	Base Salary (\$)	Base Salary (\$)	RSU Grant	RSU Grant
Bruce D. Hansen	(25)%	\$ 550,000	\$ 412,500	120,000	360,000
Lee M. Shumway*	(15)%	275,750	234,388	80,000	0
Robert I. Pennington	(20)%	297,000	237,600	100,000	300,000
R. Scott Roswell**	(15)%	265,750	225,888	80,000	240,000

Mr. Shumway's annual base salary was increased effective January 16, 2016 to \$275,750. Mr. Shumway has announced his departure from the Company effective May 12, 2017.

Mr. Roswell's annual base salary was increased effective January 16, 2016 to \$265,750.

21

Table of Contents

Employee Benefits

Our executive officers generally participate in the same employee benefit programs (401(k) plan, health, dental, vision, life, accident and disability insurance) as other employees. In 2012, the Company initiated an executive physical program with the University of Colorado Hospital. The Company covers the cost of the executive officer to participate in the executive physical program every two (2) years. Messrs. Hansen, Shumway, and Roswell participated in 2016.

Employment/Change of Control Agreements

In order to attract and retain key executives, the Company has entered into employment agreements with each of its named executive officers. In 2016, consistent with our ongoing cash conservation program, the agreements for Mr. Hansen and Mr. Pennington were amended and new employment agreements were provided to Mr. Shumway and Mr. Roswell, all on identical terms. The amended and new employment agreements provide for a term of one-year, subject to a one-year automatic renewal if not terminated earlier upon ninety (90) days' notice. Generally, if a change of control occurs and the Company (or its successor) terminates the employment of the named executive officer without cause during the one year period following the closing of the change of control event (a double-trigger arrangement) or the executive terminates employment for good reason, which includes a material diminution of the executive's duties or compensation; geographic relocation; direction to the executive that would violate local, state, or federal law; or failure of the Company to pay base compensation (not subject to salary reduction program), (ii) 100% of the executive's target annual incentive award for one year, and (iii) as to Mr. Hansen, his cash incentive award for major financing, if it had not previously been paid and (b) full vesting of all outstanding stock-based equity awards, if not otherwise accelerated under the provision of a change of control in the Company's Equity Incentive Plan. The severance payment is subject to execution of a binding termination release and confidentiality, non-competition, and non-solicitation covenants.

Individual Executive Officers and the CEO

Each of our executive officers is considered individually in the compensation setting process. In setting cash compensation, the primary factors are the scope of the executive officer's duties and responsibilities, the executive officer's performance of those duties and responsibilities, the executive officer's experience level and tenure with us, and a general evaluation of the competition in the market for key executives with the executive officer's experience. Long-term equity incentives are focused largely on retention of our executive officers and matching the financial interests of our executive officers with those of our stockholders.

SUMMARY COMPENSATION TABLE

The following table lists the annual compensation information for the fiscal years 2016, 2015, and 2014 of our Chief Executive Officer, Chief Financial Officer, Chief Operating Officer and our Chief Legal Officer.

			Non-Equity		All Other	
Name and Principal Position	Year	Salary (\$)	Incentive Award(1) (\$)	Stock Option/SA Awards(2) (\$Awards(2)	AICompensation (\$) (\$)	Total (\$)
Bruce D. Hansen(3)	2016 \$	412,500	\$ 415,500	\$ 21,600 \$	\$ 19,051(3)\$	865,551
Chief Executive Officer	2015	550,000		184,665	19,051(3)	753,716
	2014	412,500			19,051(3)	431,551
Lee M. Shumway(4)	2016	234,388		14,400	14,027(7)	262,815
Chief Financial Officer	2015	234,350	117,175	78,684	15,058(7)	445,267
	2014	199,198			11,885(7)	211,083
Robert I. Pennington(5)	2016	237,600		18,000	15,912(5)	271,512
Chief Operating Officer	2015	297,000	148,500	99,719	16,611(5)	561,830
	2014	237,600			14,981(5)	252,581
R. Scott Roswell(6)	2016	225,888		14,400	13,699(6)	253,987
Chief Legal Officer	2015	250,700	125,350	84,174	14,695(6)	474,919
	2014	213,095			12,578(6)	225,673
		- ,			, (-)	- /

- (1) No incentive award was approved for 2013, 2014 or 2015, based on the Company's cost reduction program. Under the Company's cash conservation program, cash retention awards were paid to each of the named executive officers on January 15, 2015, with the exception of Mr. Hansen, who deferred payment of his bonus to 2016.
- These amounts do not represent the actual amounts paid to or realized by these individuals. These amounts represent the aggregate grant date fair value for grants during the fiscal year, computed in accordance with applicable accounting rules (FASB ASC Topic 718), excluding the amount of estimated forfeitures. For information regarding the assumptions used to calculate the grant date fair value, see Note 9 to the Consolidated Financial Statements contained in our Annual Report on Form 10-K for the year ended December 31, 2016. The grant date fair value for restricted stock units granted on January 16, 2015, is \$0.47 per share. The grant date fair value for restricted stock units granted on January 16, 2016, is \$0.18 per share.
- The All Other Compensation amount for Mr. Hansen for 2016 represents \$12,750 in Company matching contributions to our 401(k) plan and \$6,301 in group term life insurance premiums paid by the Company. The All Other Compensation amount for Mr. Hansen for 2015 represents \$12,750 in Company matching contributions to our 401(k) plan and \$6,301 in group term life insurance premiums paid by the Company. The All Other Compensation amount for Mr. Hansen for 2014 represents \$12,750 in Company matching contributions to our 401(k) plan and \$6,301 in group term life insurance premiums paid by the Company.
- (4)

 The All Other Compensation amount for Mr. Shumway for 2016 represents \$11,719 in Company matching contributions to our 401(k) plan and \$2,308 in group term life insurance premiums paid by the Company. The All Other Compensation amount for Mr. Shumway for 2015 represents \$12,750 in Company matching contributions to our 401(k) plan and \$2,308 in group term life insurance premiums paid by the Company. The All Other Compensation amount for Mr. Shumway for 2014 represents \$9,577 in Company matching contributions to our 401(k) plan and \$2,308 in group term life insurance premiums paid by the Company.

Table of Contents

- The All Other Compensation amount for Mr. Pennington for 2016 represents \$12,051 in Company matching contributions to our 401(k) plan and \$3,861 in group term life insurance premiums paid by the Company. The All Other Compensation amount for Mr. Pennington for 2015 represents \$12,750 in Company matching contributions to our 401(k) plan and \$3,861 in group term life insurance premiums paid by the Company. The All Other Compensation amount for Mr. Pennington for 2014 represents \$11,120 in Company matching contributions to our 401(k) plan and \$3,861 in group term life insurance premiums paid by the Company.
- The all other compensation amount for Mr. Roswell for 2016 represents \$11,366 in company matching contributions to our 401(k) plan and \$2,333 in group term life insurance premiums paid by the Company. The all other compensation amount for Mr. Roswell for 2015 represents \$12,362 in company matching contributions to our 401(k) plan and \$2,333 in group term life insurance premiums paid by the company. The all other compensation amount for Mr. Roswell for 2014 represents \$10,245 in company matching contributions to our 401(k) plan and \$2,333 in group term life insurance premiums paid by the company.

GRANTS OF PLAN-BASED AWARDS

In 2016, we made stay incentive RSU grants to each of our named executive officers. Historically, we have issued stock options, stock appreciation rights, restricted stock and restricted stock unit awards to our executive officers and key employees as part of our compensation plans under our equity incentive plans. See "Equity Compensation Plan Information." The purpose of the 2006 Plan is to provide us with a greater ability to attract, retain, and motivate our officers, directors and key employees. In 2016, in keeping with our cash conservation efforts, we used the 2006 Plan to provide stay incentive RSU grants on January 16, 2016, as outlined below, and to other senior managers of the Company.

	Restricted Stock Units	Market Value at
Name	Awarded	Grant (\$)
Bruce D. Hansen	120,000	\$ 21,600
Lee M. Shumway	80,000	14,400
Robert I. Pennington	100,000	18,000
R. Scott Roswell	80,000	14,400

Our 2006 Plan provides for the grant of incentive stock options, nonstatutory stock options, restricted stock awards, restricted stock units and stock appreciation rights, which may be granted to our employees (including officers), directors and consultants. Each award is subject to an agreement between the Company and the recipient of the grant reflecting the terms and conditions of the award. Subject to the terms of the 2006 Plan, the Compensation Committee establishes grant dates, the numbers and types of stock awards to be granted and the terms and conditions of the stock awards, including the period of their exercisability and vesting. The Compensation Committee, in accordance with the 2006 Plan, sets the option exercise price, and, if applicable, the strike price for stock appreciation rights, in each case based on the closing price of the Company's common stock on the date of the grant.

Compensation Arrangements and Employment Agreements

The material terms of our NEOs annual compensation, including base salaries, cash incentive awards, our equity granting practices and employment, change in control and stay bonus agreements are described in our "Compensation Discussion and Analysis Elements of Compensation and 2016 Compensation Decisions Stay Agreements" and "Employment Agreements and Stay Agreements" sections of this proxy statement.

OUTSTANDING EQUITY AWARDS AT DECEMBER 31, 2016

The following table provides information with respect to outstanding stock options/SARs, restricted stock awards and restricted stock units held by our named executive officers as of December 31, 2016.

		OP'	TION AWARDS	S		STOCK AWARDS				
Name	Number of Securities Underlying Unexercised Options/SARs(I (#) Exercisable	Unexercised ptions/SARs((#) O	ptions/SARs(1)	Exercise		Number of Shares or Units of Stock That Have Not	Value of Shares or Units of Stock That Have Not	Equity Incentive Plan Awards: Number of Unearned Shares, Units or Other Rights That Have Not Vested(1) (#)	Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$)	
Bruce D. Hansen	66,666(3			\$ 0.76	2/27/2017		\$		\$	
	23,233(,	90,000(4) 140,000(5)		10/17/2025		•	30,000(4)	7,500	
			56,003(6) 56,004(7)	3.72 3.72	10/17/2025 10/17/2026			50,000(5)	12,500	
			56,003(8)	3.72	10/17/2025			18,481(6) 18,481(7)	4,620 4,620	
			56,004(9)	3.72	10/17/2026			18,481(8) 18,481(9) 120,000(10)	4,620 4,620 30,000	
Lee M. Shumway			27,000(4) 43,000(5)	5.49 3.28	10/17/2025 10/17/2025			9,000(4) 14,000(5)	2,250 3,500	
			20,365(6) 20,365(7)	3.72 3.72	10/17/2025 10/17/2026			6,720(6)	1,680	
			20,365(8)	3.72	10/17/2025			6,721(7)	1,680	
			20,365(9)	3.72	10/17/2026			6,720(8) 6,721(9) 80,000(10)	1,680 1,680 20,000	
Robert I. Pennington			40,000(4) 60,000(5)	5.49 3.28	10/17/2025 10/17/2025			13,500(4)	3,375	
			28,002(6) 28,002(7)		10/17/2025 10/17/2026			22,000(5)	5,500	
			28,002(8)	3.72	10/17/2025			9,240(6) 9,241(7)	2,310 2,310	
	26 6674	1)	28,002(9)		10/17/2026			9,240(8) 9,241(9)	2,310 2,310	
	26,667(1	1)		0.96	2/5/2017			100,000(10) 145,000(12)		

R. Scott Roswell 27,000(4) 5.49 10/17/2025 9,000(4)	2,250
43,000(5) 3.28 10/17/2025 14,000(5)	3,500
20,365(6) 3.72 10/17/2025	
20,365(7) 3.72 10/17/2026	
6,720(6)	1,680
6,721(7)	1,680
20,365(8) 3.72 10/17/2025	
20,365(9) 3.72 10/17/2026	
6,720(8)	1,680
6,721(9)	1,680
80,000(10)	20,000

⁽¹⁾ All of the awards were made under the 2006 Plan.

Performance SARs and restricted stock units were granted on December 16, 2010 and are scheduled to vest at the commencement of commercial production at the Mt. Hope Mine, subject to continuous employment. The market value of the restricted stock units was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by the number of shares granted, which assumes that all performance goals for the

⁽²⁾ The option/SAR exercise price is the closing market price of the stock on the day of the grant.

⁽³⁾ SARs were granted on February 27, 2009 and vested on February 27, 2012.

Table of Contents

shares are achieved. The SARs expire on the earliest of termination of service, the 5th anniversary of the vesting date, the 10th anniversary of the date of grant, or in the event of a change in control.

- Performance SARs and restricted stock units were granted on December 15, 2011 and are scheduled to vest at the commencement of commercial production at the Mt. Hope Mine, subject to continuous employment. The market value of the restricted stock units was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by the number of shares granted, which assumes that all performance goals for the shares are achieved. The SARs expire on the earliest of termination of service, the 5th anniversary of the vesting date, the 10th anniversary of the date of grant, or in the event of a change in control.
- Performance SARs and restricted stock units were granted on December 12, 2012 and are scheduled to vest at the commencement of commercial production at the Mt. Hope Mine, subject to continuous employment. The market value of the restricted stock units was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by the number of shares granted, which assumes that all performance goals for the shares are achieved. The SARs expire on the earliest of termination of service, the 5th anniversary of the vesting date, the 10th anniversary of the date of grant, or in the event of a change in control.
- Performance SARs and restricted stock units were granted on December 12, 2012 and are scheduled to vest one year following the commencement of commercial production at the Mt. Hope Mine, subject to continuous employment. The market value of the restricted stock units was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by the number of shares granted, which assumes that all performance goals for the shares are achieved. The SARs expire on the earliest of termination of service, the 5th anniversary of the vesting date, the 10th anniversary of the date of grant, or in the event of a change in control.
- Performance SARs and restricted stock units were granted on December 11, 2013 and are scheduled to vest at the commencement of commercial production at the Mt. Hope Mine, subject to continuous employment. The market value of the restricted stock units was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by the number of shares granted, which assumes that all performance goals for the shares are achieved. The SARs expire on the earliest of termination of service, the 5th anniversary of the vesting date, the 10th anniversary of the date of grant, or in the event of a change in control.
- Performance SARs and restricted stock units were granted on December 11, 2013 and are scheduled to vest one year following the commencement of commercial production at the Mt. Hope Mine, subject to continuous employment. The market value of the restricted stock units was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by the number of shares granted, which assumes that all performance goals for the shares are achieved. The SARs expire on the earliest of termination of service, the 5th anniversary of the vesting date, the 10th anniversary of the date of grant, or in the event of a change in control.
- Performance restricted stock units were granted on January 16, 2015 and vested on January 15, 2016. The market value of the restricted stock units was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by the number of shares granted, which assumes that all performance goals for the shares are achieved.
- (11) SARs were granted on February 5, 2009 and vested on February 5, 2012.
- An award of 165,000 shares of restricted stock was granted on October 19, 2007. During 2008, 10,000 shares of restricted stock vested upon satisfaction of the engineering and procurement phase 1 goal. During 2009, 10,000 shares of restricted stock vested upon satisfaction of the engineering and procurement phase 2 goals. The remaining shares are scheduled to vest based upon the achievement of designated performance goals. In addition, 30,000 shares will vest upon attainment of the construction completion goal, 30,000 shares will vest upon satisfying the cost of contracted construction goal, 35,000 shares (17,500 each) will vest upon satisfying the commissioning phase 1 and phase 2 goals and the remaining 50,000 shares will vest based upon satisfying the specified production goal within six months of initial start-up. The Company may adjust the timing of completion of the goals to accommodate changes in the schedule as a result of environmental permitting or financial considerations. The market value of the stock award was determined by multiplying the closing market price of a share of our common stock on December 30, 2016 of \$0.25 by 145,000 shares, which assumes that all performance goals for the shares are achieved.

OPTION/SAR EXERCISES AND STOCK VESTED DURING 2016

	OPTION/S	SAR AWARDS	STOCK AWARDS			
Name	Number of	Value Realized	Number of	Value Realized on		
	Shares	on	Shares Acquired	Vesting		
	Acquired	Exercise	on Vesting	(1) (\$)		
	on	(\$)	(#)			

Edgar Filing: General Moly, Inc - Form PRE 14A

Exe	rcise

(#)		
Bruce D. Hansen	392,904 \$	117,857
	24,642	7,146
Lee M. Shumway	167,413	33,479
	8,961 \$	2,599
Robert I. Pennington	212,168	42,429
	12,321	3,573
R. Scott Roswell	179,093	35,814
	8,961	2,599

(1)
Amount reported represents the closing price of our common stock, as reported on the NYSE MKT, on each vesting date, multiplied by the number of shares vested.

Table of Contents

POTENTIAL PAYMENTS UPON TERMINATION OR CHANGE IN CONTROL

Potential payments upon termination or change in control for Mr. Hansen, Mr. Shumway, Mr. Pennington and Mr. Roswell are set forth in their respective employment agreements, described below.

In the event of a change in control as defined in our 2006 Plan, all outstanding options and other stock awards under the plans may be assumed, continued or substituted by any surviving or acquiring entity. If the surviving or acquiring entity elects not to assume, continue or substitute the awards, the vesting of such awards held by award holders whose service with us or any of our affiliates has not terminated will be accelerated, the awards will be fully vested and exercisable immediately prior to the consummation of the transaction and the stock awards will automatically terminate upon consummation of the transaction if not exercised prior to such event.

Employment Agreements and Stay Agreements

The following is a summary of the employment agreements that were in effect between us and each of our named executive officers during the last fiscal year.

Bruce D. Hansen. On January 30, 2007, we entered into an employment agreement with Mr. Hansen to serve as our Chief Executive Officer for a term of three years. Mr. Hansen's agreement was subsequently amended and restated effective January 1, 2012, to extend the term of the agreement to terminate automatically on the earlier of (1) the one-year anniversary of the date on which the Company achieves Commercial Production (as such term is defined in the Amended and Restated Limited Liability Agreement of Eureka Moly, LLC dated February 26, 2008) and (2) December 31, 2015; and to eliminate the single-trigger change of control arrangement. Effective January 1, 2016 the employment agreement of Mr. Hansen was amended and restated to provide for a term of one-year, subject to an automatic one-year renewal if not terminated earlier upon ninety (90) day notice. The following is a description of the terms of his agreement, as in effect on December 31, 2016.

Under the terms of the agreement, as amended, Mr. Hansen's base salary is \$550,000, which was temporarily reduced to \$412,500 in September 2013 and re-instated effective January 16, 2015 and was again reduced to \$412,500 in January 2016 and continued to be reduced through 2016 and into 2017, as discussed in the next paragraph. Mr. Hansen is eligible to receive a discretionary cash incentive payment in an amount, if any, as determined by the Board from time to time. Upon the completion of an equity or debt financing that raises sufficient capital to commence production at the Mt. Hope Project, Mr. Hansen remains entitled to a cash payment of \$1,000,000. If a "change of control" occurs and the Company (or its successor) terminates the employment of Mr. Hansen without cause during the one-year period following the close of the change of control event (a double-trigger arrangement) or Mr. Hansen terminates his employment for good reason, which includes a material diminution of his duties or compensation, geographic relocation; a direction to violate local, state or federal law; or a failure of the Company to pay base compensation in a timely manner, Mr. Hansen would be entitled to receive a payment equal to two years of annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction), one year target annual incentive compensation, and full vesting of all outstanding unvested stock-based equity awards, if not otherwise accelerated under the provisions of a "change of control" in the Company's Equity Incentive Plan. In addition, he will be paid the cash incentive award of \$1,000,000 for major financing if it has not previously been paid. In the event the Company terminates Mr. Hansen's employment without cause, independent of a "change of control", Mr. Hansen would be entitled to any base salary earned but not yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). If Mr. Hansen terminates his employment for "good reason" as described above, independent of a "change of control", Mr. Hansen would be entitled to any base salary earned but not

Table of Contents

yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). Each of the described severance payments is subject to execution of a binding termination release and confidentiality, non-competition, and non-solicitation covenants.

On September 7, 2013 we entered into a Salary Reduction and Stay Incentive Agreement ("Stay Agreement") with Mr. Hansen. With the Stay Agreement, Mr. Hansen and the Company agreed to reduce Mr. Hansen's base salary to \$412,500 for the term of the Stay Agreement, and provided for a Stay Incentive Award of \$412,500 and a Restricted Stock Unit ("RSU") award of 245,536 RSUs if Mr. Hansen remains continuously employed through the End Date, as defined therein. The Stay Agreement provided that it would end on the earliest to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors; a Change of Control (as defined in Mr. Hansen's employment agreement; involuntary termination (absent cause); or January 15, 2015. On January 16, 2015, we entered into a First Amendment to Salary Reduction and Stay Incentive Agreement with Mr. Hansen, effective as of January 14, 2015. Pursuant to this amendment, the Company agreed to grant 392,904 RSUs to Mr. Hansen, in consideration for Mr. Hansen's agreement to extend the payment of his \$412,500 2014 cash incentive bonus under his Stay Agreement to January 16, 2016. These awards vested and were paid to Mr. Hansen on January 15, 2016. Effective January 16, 2016, we agreed to grant Mr. Hansen 120,000 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2017, we agreed to grant Mr. Hansen 360,000 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2017 through January 16, 2018.

Lee M. Shumway. On November 6, 2007, we entered into an offer letter agreement with Lee M. Shumway pursuant to which Mr. Shumway initially served as the Director of Business Process/Information Technology. Mr. Shumway served as our Controller and Treasurer in 2015, and was promoted to Chief Financial Officer effective October 16, 2015. On January 1, 2012, we entered into a Change of Control Severance, Confidentiality and Non-Solicitation Agreement with Mr. Shumway, which included the same definition of change of control as the agreements for Messrs. Hansen, Pennington and Roswell. In December 2015, we approved the promotion of Mr. Shumway to CFO, and authorized a 2016 salary increase to \$275,750, associated with the retirement of our former CFO David Chaput, which consistent with similar treatment of our other named executive officers was then subject to a salary reduction of 15% to \$234,350, which continued through 2016 and is continuing into 2017. On January 16, 2016 we entered in Employment Agreement and with its effectiveness terminated the Change of Control Severance, Confidentiality and Non-Solicitation Agreement. Consistent with the other named executive officers, the January 16, 2016 Employment Agreement of Mr. Shumway provides for a term of one-year, subject to an automatic one-year renewal if not terminated earlier upon ninety (90) day notice. The following is a description of the terms of his agreement, as in effect on December 31, 2016.

Under the terms of the agreement, as amended, Mr. Shumway's base salary is \$275,750, which as discussed above was temporarily reduced to \$234,350 in January 2016 and continued to be reduced through 2016 and into 2017, as discussed in the next paragraph. Mr. Shumway is eligible to receive a discretionary cash incentive payment in an amount, if any, as determined by the Board from time to time. If a "change of control" occurs and the Company (or its successor) terminates the employment of Mr. Shumway without cause during the one-year period following the close of the change of control event (a double-trigger arrangement) or Mr. Shumway terminates his employment for good reason, which includes a material diminution of his duties or compensation, geographic relocation; a direction to violate local, state or federal law; or a failure of the Company to pay base compensation in a timely manner, Mr. Shumway would be entitled to receive a payment equal to two years of annual base salary (determined by applying his base salary immediately preceding the implementation of the salary

Table of Contents

reduction), one year target annual incentive compensation, and full vesting of all outstanding unvested stock-based equity awards, if not otherwise accelerated under the provisions of a "change of control" in the Company's Equity Incentive Plan. In the event the Company terminates Mr. Shumway's employment without cause, independent of a "change of control", Mr. Shumway would be entitled to any base salary earned but not yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). If Mr. Shumway terminates his employment for "good reason" as described above, independent of a "change of control", Mr. Shumway would be entitled to any base salary earned but not yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). Each of the described severance payments is subject to execution of a binding termination release and confidentiality, non-competition, and non-solicitation covenants.

On September 7, 2013 we entered into a Stay Incentive Agreement with Mr. Shumway. With the Agreement, Mr. Shumway and the Company agreed to provide a Stay Incentive Award of \$117,175 and Restricted Stock Unit ("RSU") Award of 69,747 RSUs if Mr. Shumway remains continuously employed through the End Date, as defined therein. The Stay Agreement provided that it would end on the earliest to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors, a Change of Control (as defined in Mr. Shumway's Change of Control, Severance, Confidentiality and Non-Solicitation Agreement; involuntary termination (absent cause); or January 15, 2015. These awards vested and were paid to Mr. Shumway on January 15, 2015. Effective January 14, 2015, we agreed to grant Mr. Shumway 167,413 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2016 through January 15, 2016. These awards vested and were paid to Mr. Shumway on January 15, 2016. Effective January 16, 2016, we agreed to grant Mr. Shumway 80,000 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2016 through January 15, 2017. These awards vested and were issued to Mr. Shumway on January 15, 2017.

Mr. Shumway has announced his departure from the Company effective May 12, 2017. Accordingly, no stay incentive award was granted to Mr. Shumway in January 2017.

Robert I. Pennington. On October 5, 2007, we entered into an offer letter agreement with Robert I. Pennington pursuant to which Mr. Pennington served as our Vice President of Engineering and Construction and was named our Chief Operating Officer in January 2012. Pursuant to the terms of this agreement, as amended, Mr. Pennington was paid a base salary of \$200,000 per year in 2007, plus eligibility for a performance based annual incentive award. Mr. Pennington's base salary was subsequently increased to \$297,000, which was temporarily reduced to \$237,600 in September 2013 and re-instated effective January 16, 2015 as discussed in the next paragraph. Mr. Pennington received an option to purchase 150,000 shares of common stock under the 2006 Plan, all of which are fully vested. In addition, Mr. Pennington is also eligible to receive up to 165,000 shares of restricted common stock upon reaching certain pre-determined goals relating to the Mt. Hope Project, of which 20,000 shares have vested and been issued to Mr. Pennington. In January 2012, we entered into a Change of Control Severance, Confidentiality and Non-Solicitation Agreement, which was superseded by an Employment Agreement entered into with Mr. Pennington effective December 12, 2012 to which the term of the agreement will terminate automatically on the earlier of (1) the one-year anniversary of the date on which the Company achieves Commercial Production (as such term is defined in the Amended and Restated Limited Liability Agreement of Eureka Moly, LLC dated February 26, 2008) and (2) December 31, 2016. Effective January 1, 2016 the employment agreement of Mr. Pennington was amended and restated to provide for a term of one-year, subject to an automatic one-year renewal if not terminated earlier upon ninety (90) day notice. The following is a description of the terms of his agreement, as in effect on December 31, 2016.

Table of Contents

Under the terms of the agreement, as amended, Mr. Pennington's base salary is \$297,000, which was temporarily reduced to \$237,600 in September 2013 and re-instated effective January 16, 2015 and was again reduced to \$237,600 in January 2016 and continued to be reduced through 2016 and into 2017, as discussed in the next paragraph. Mr. Pennington is eligible to receive a discretionary cash incentive payment in an amount, if any, as determined by the Board from time to time. If a "change of control" occurs and the Company (or its successor) terminates the employment of Mr. Pennington without cause during the one-year period following the close of the change of control event (a double-trigger arrangement) or Mr. Pennington terminates his employment for good reason, which includes a material diminution of his duties or compensation, geographic relocation; a direction to violate local, state or federal law; or a failure of the Company to pay base compensation in a timely manner, Mr. Pennington would be entitled to receive a payment equal to two years of annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction), one year target annual incentive compensation, and full vesting of all outstanding unvested stock-based equity awards, if not otherwise accelerated under the provisions of a "change of control" in the Company's Equity Incentive Plan. In the event the Company terminates Mr. Pennington's employment without cause, independent of a "change of control", Mr. Pennington would be entitled to any base salary earned but not yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). If Mr. Pennington terminates his employment for "good reason" as described above, independent of a "change of control", Mr. Pennington would be entitled to any base salary earned but not yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). Each of the described severance payments is subject to execution of a binding termination release and confidentiality, non-competition, and non-solicitation covenants.

On September 7, 2013 we entered into a Salary Reduction and Stay Incentive Agreement ("Stay Agreement") with Mr. Pennington. With the Stay Agreement, Mr. Pennington and the Company agreed to reduce Mr. Pennington's base salary to \$237,600 for the term of the Stay Agreement, and provided for a Stay Incentive Award of \$148,500 and a Restricted Stock Unit ("RSU") award of 88,393 RSUs if Mr. Pennington remained continuously employed through the End Date, as defined therein. The Stay Agreement provided that it would end on the earliest to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors; a Change of Control (as defined in Mr. Pennington's employment agreement; involuntary termination (absent cause); or January 15, 2015. These awards vested and were paid to Mr. Pennington on January 15, 2015. Effective January 14, 2015, we agreed to grant Mr. Pennington 212,168 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2015 through January 15, 2016. These awards vested and were awarded to Mr. Pennington on January 15, 2016. Effective January 16, 2016, we agreed to grant Mr. Pennington 100,000 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2017, we agreed to grant Mr. Pennington 300,000 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2017 through January 16, 2018.

R. Scott Roswell. On August 17, 2010, we entered into an offer letter agreement with R. Scott Roswell pursuant to which Mr. Roswell served as our Corporate Counsel and Vice President of Human Resources and was named our Chief Legal Officer in October 2015. On January 1, 2012, we entered into a Change of Control Severance, Confidentiality and Non-Solicitation Agreement with Mr. Roswell, which included the same definition of change of control as the agreements for Messrs. Hansen, Pennington and Shumway. In December 2015, we approved a 2016 salary increase for Mr. Roswell to \$265,750 inclusive of his additional responsibilities associated with the resignation of our former Director of Investor Relations, which consistent with similar treatment of our other named executive

Table of Contents

officers was then subject to a salary reduction of 15% to \$225,888, which continued through 2016 and is continuing into 2017. On January 16, 2016 we entered in Employment Agreement, and with its effectiveness, terminated the Change of Control Severance, Confidentiality and Non-Solicitation Agreement. Consistent with the other named executive officers, the January 16, 2016 Employment Agreement of Mr. Roswell provides for a term of one-year, subject to an automatic one-year renewal if not terminated earlier upon ninety (90) day notice. The following is a description of the terms of his agreement, as in effect on December 31, 2016.

Under the terms of the agreement, as amended, Mr. Roswell's base salary is \$265,750, which as discussed above was temporarily reduced to \$225,888 in January 2016 and continued to be reduced through 2016 and into 2017, as discussed in the next paragraph. Mr. Roswell is eligible to receive a discretionary cash incentive payment in an amount, if any, as determined by the Board from time to time. If a "change of control" occurs and the Company (or its successor) terminates the employment of Mr. Roswell without cause during the one-year period following the close of the change of control event (a double-trigger arrangement) or Mr. Roswell terminates his employment for good reason, which includes a material diminution of his duties or compensation, geographic relocation; a direction to violate local, state or federal law; or a failure of the Company to pay base compensation in a timely manner, Mr. Roswell would be entitled to receive a payment equal to two years of annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction), one year target annual incentive compensation, and full vesting of all outstanding unvested stock-based equity awards, if not otherwise accelerated under the provisions of a "change of control" in the Company's Equity Incentive Plan. In the event the Company terminates Mr. Roswell's employment without cause, independent of a "change of control", Mr. Roswell would be entitled to any base salary earned but not yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). If Mr. Roswell terminates his employment for "good reason" as described above, independent of a "change of control", Mr. Roswell would be entitled to any base salary earned but not yet paid and a severance payment in an amount equal to six (6) months of his annual base salary (determined by applying his base salary immediately preceding the implementation of the salary reduction). Each of the described severance payments is subject to execution of a binding termination release and confidentiality, non-competition, and non-solicitation covenants.

On September 7, 2013 we entered into a Stay Incentive Agreement with Mr. Roswell. With the Agreement, Mr. Roswell and the Company agreed to provide a Stay Incentive Award of \$125,350 and Restricted Stock Unit ("RSU") Award of 74,613 RSUs if Mr. Roswell remained continuously employed through the End Date, as defined therein. The Stay Agreement provided that it would end on the earliest to occur of a financing plan for the Mt. Hope Project approved by the Board of Directors, a Change of Control (as defined in Mr. Roswell's Change of Control, Severance, Confidentiality and Non-Solicitation Agreement; involuntary termination (absent cause); or January 15, 2015. These awards vested and were paid to Mr. Roswell on January 15, 2015. Effective January 14, 2015, we agreed to grant Mr. Roswell 179,093 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2015 through January 15, 2016. These awards vested and were paid to Mr. Roswell on January 15, 2016. Effective January 16, 2016, we agreed to grant Mr. Roswell 80,000 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2016 through January 15, 2017. These awards vested and were issued to Mr. Roswell on January 15, 2017. Lastly, effective January 16, 2017, we agreed to grant Mr. Roswell 240,000 RSUs, subject to his execution of a new Stay Incentive Agreement covering the period from January 16, 2017 through January 16, 2018.

Table of Contents

Change of Control Employment Agreements

Generally, for purposes of the executive employment agreements, a change of control occurs if:

Any single holder (or group acting in concert) acquires ownership of 50% or more of the outstanding common stock or combined voting power of the Company (under the present capitalization, outstanding stock and voting power is the same). The following acquisitions are excluded: (a) acquisition of shares from the Company; and (b) acquisition of shares by the Company or by any employee benefit plan sponsored by the Company; or

There is a business combination (a merger, reorganization, etc.) involving the Company and another company unless substantially all of the holders who owned shares of the Company before the combination own more than half of the shares of the company resulting from the combination in substantially the proportion that they owned our shares and no one (including a group acting in concert) owns more than one-half of the resulting company. In other words, generally, if we merged with another company and our stockholders owned more than one-half of the resulting company there would not be a change of control. If they owned less than 50%, a change of control would have occurred; or

The current (incumbent) members of the Company's Board no longer constitute at least a majority of the Board; provided, however, that an individual that becomes a director whose election or nomination was approved by at least a majority of the directors serving on the incumbent Board is considered as though such individual was a member of the incumbent Board unless the individual assumed the office as a result of an actual or threatened election contest or solicitation of proxies or consents on the person's behalf; or

All or substantially all of our operating assets are sold to an unrelated party; or

Our stockholders approve a liquidation or dissolution of the Company.

2006 Equity Incentive Plan

In general, under the terms of the 2006 Equity Incentive Plan, in the event of a change in control (as defined in each of the plans), outstanding awards will either be assumed or substituted by the surviving corporation or automatically become fully vested and exercisable for a limited period of time.

Severance and Change in Control Payments

The following is a summary of potential payments payable to our named executive officers upon termination of employment or a change in control of the Company under each circumstance assuming the event occurred on December 31, 2016. Actual payments would be paid in a lump sum and may be more or less than the amounts described below. In addition, the Company may enter into new arrangements or modify these arrangements, from time to time, as was done in January 2017.

Table of Contents

POTENTIAL PAYMENTS UPON TERMINATION OR CHANGE IN CONTROL ON DECEMBER 31, 2016

The following are estimated payments that would be provided to each of our named executive officers in the event of termination of the named executive officer's employment assuming a termination date of December 31, 2016.

Name	Base Salary (\$)	Incentive Award (\$)	Ao Vesti	Value of ccelerated ing of Equity vards(1) (\$)
Bruce D. Hansen(2)				
Termination without cause or for good reason as a result of a change of control	\$ 1,100,000	\$ 1,412,500	\$	68,481
Termination without cause unrelated to change of control	275,000			
Lee M. Shumway(3)				
Termination without cause or for good reason as a result of a change of control	551,500	137,875		32,471
Termination without cause unrelated to change of control	\$ 137,875	\$	\$	
Robert I. Pennington(3)				
Termination without cause or for good reason as a result of a change of control	594,000	148,500		79,366
Termination without cause unrelated to change of control	148,500			
R. Scott Roswell(3)				
Termination without cause or for good reason as a result of a change of control	531,500	132,875		32,471
Termination without cause unrelated to change of control	132,875			

- Amounts are based upon our closing stock price of \$0.25 per share on December 31, 2016. Amount includes the value of accelerated vesting of stock awards; accelerated vesting of SARs and accelerated vesting of stock options, to the extent the option exercise price exceeded the closing stock price of our common stock on December 31, 2016. The amounts do not include potential exercise of vested stock options. See the "Outstanding Equity Awards at December 31, 2016" table for information regarding vested stock options.
- Includes a change of control payment equal to two times his base salary, 100% of his annual target incentive payment, and payment of his \$1,000,000 major financing award to the extent not already paid. In the event of his termination without cause he is entitled to a payment equal to two times his base salary. In the event of his termination for good reason he is entitled to a payment equal to one times his base salary.
- In the event of his termination without cause as a result of a change of control, or one year following the closing of the change of control, or election of termination for "good reason" he is entitled to two years of his base salary, 100% of his annual target incentive payment, and vesting of all of his outstanding stock awards.

DIRECTOR COMPENSATION

The following table lists compensation information for fiscal 2016 for our directors and our secretary who were not employees. Mr. Hansen, who is also our Chief Executive Officer, does not receive any separate compensation for his service as a director. Mr. Hansen's compensation is fully reflected in the Summary Compensation Table and, as appropriate, in the other tables above.

Table of Contents

On the recommendation of the Compensation Committee, at its June 16, 2011 meeting the Board approved guidelines for share ownership for directors. The current guideline amount is equal to a multiplier of four times each director's individual retainer from the Company. The Board also set a target of five years for each director to reach his/her ownership guideline level. As of December 31, 2016, all non-employee Directors had reached their target ownership, except for Mr. Zhang who was appointed to the Board in December 2015. Effective September 7, 2013, we implemented a cost reduction and personnel retention program, which was maintained throughout 2016 and continues into 2017, which included reductions in base cash compensation for members of the Board of Directors, as well as our executive officers and senior management employees.

Columns required by SEC rules are omitted where there is no amount to report.

	Fees Earned or Paid in		Stock Awards(1)		Total
Name	(Cash (\$)		(\$)	(\$)
Ricardo M. Campoy	\$	79,500	\$	5,000	\$ 84,500
Mark A. Lettes		53,250		5,000	58,250
Gary A. Loving		40,500		5,000	45,500
Gregory P. Raih		47,250		5,000	52,250
Nelson F. Chen(2)		14,700		5,000	19,700
Tong Zhang		33,000		5,000	38,000
Michael K. Branstetter(3)	\$	15,000	\$	2,000	\$ 17,000

- These amounts do not represent the actual amounts paid to or realized by these individuals. These amounts represent the aggregate grant date fair value for grants during the fiscal year, computed in accordance with applicable accounting rules (FASB ASC Topic 718), excluding the amount of estimated forfeitures. For information regarding the assumptions used to calculate the grant date fair value, see Note 9 to the Consolidated Financial Statements contained in our Annual Report on Form 10-K for the year ended December 31, 2016. As of December 31, 2016, the aggregate number of shares of our common stock underlying outstanding option awards and the number of shares of restricted stock for each non-employee director and our secretary was zero.
- (2) As discussed below, Mr. Chen's term as a director expired on June 8, 2016 at the 2016 Annual Meeting.
- (3) Michael K. Branstetter serves as our secretary.

Table of Contents

Director and Secretary Compensation Program

The following table describes the payments to be made by us under our director and secretary compensation program:

Director	
Annual Retainer	\$40,000 total paid quarterly in arrears*
Board Meeting Fee	\$1,000 paid quarterly in arrears*
Audit Committee Chair	\$10,000 total paid quarterly in arrears*
Other Committee Chairs	\$5,000 total paid quarterly in arrears*
Board Chair Annual Retainer	\$80,000 total paid quarterly in arrears(1)*
Committee Meeting Fee	\$1,000 paid quarterly in arrears*
Sign-on Equity	20,000 shares(2)
Annual Equity	25,000 shares(3)
Resignation Equity	5,000 shares(4)
Secretary	
Annual Retainer	\$20,000 total paid quarterly in arrears*
Annual Equity	10,000 shares(3)

- During 2016, all cash compensation for Directors remained reduced by 25% from the amounts listed above as part of our cash conservation program implemented in September 2013.
- (1)

 Board Chair annual retainer is paid to the Board Chair in lieu of the annual retainer paid to other directors and is cash only.
- (2) Represents the number of full-value, fully vested shares of common stock granted upon election to the Board.
- (3)

 Represents the number of full-value, fully vested shares of common stock granted annually on the first business day after January 1.

 New directors receive a pro-rated grant, based upon the time of joining the Board (in addition to the Sign-on Equity award).
- (4)

 Represents the number of full-value, fully vested shares granted upon approval by the Compensation Committee if a director is asked to resign.

COMPENSATION COMMITTEE INTERLOCKS AND INSIDER PARTICIPATION

We do not have any interlocking relationships between any member of our Compensation Committee or Board and any of our executive officers that would require disclosure under the applicable rules promulgated under the U.S. federal securities laws.

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and special reports, proxy statements and other information with the United States Securities and Exchange Commission, or the SEC. Our SEC filings are available to the public from our web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's web site at http://www.generalmoly.com or from the SEC's public reference into and is not made a part of this proxy statement. You may also read and copy any document we file at the SEC's public reference room located at 100 F Street, NE, Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the public reference rooms.

ADDITIONAL STOCKHOLDER INFORMATION

Stockholder Proposals and Recommendations for Director Nominees for the 2018 Annual Meeting

We anticipate that we will hold our 2018 Annual Meeting of Stockholders within 30 days before or after June 8, 2018. If you wish to submit a proposal for inclusion in our proxy materials to be circulated in connection with our 2018 Annual Meeting of Stockholders, you must send the proposal to the Company at the address below. The proposal must be received no later than December 29, 2017 to be considered for inclusion in the Company's proxy statement and form of proxy for that meeting.

For stockholder proposals submitted outside of the process described above, the Company's bylaws require that advance written notice of a stockholder proposal for matters to be brought before an annual stockholders' meeting be received by the Company not less than 90 days or more than 120 days before the first anniversary date of the immediately preceding annual stockholders' meeting. Accordingly, notice of stockholder proposals for the 2018 Annual Meeting must be received by the Company between February 8, 2018 and March 10, 2018. In addition, among other requirements set forth in the SEC's proxy rules, you must have continuously held at least \$2,000 in market value or 1% of our outstanding stock for at least one year by the date you submit the proposal, and you must continue to own such stock through the date of the meeting.

Stockholder proposals and recommendations for director nominees should be sent to General Moly, Inc. Board of Directors, c/o Corporate Secretary, 1726 Cole Blvd., Suite 115 Lakewood, Colorado 80401.

Householding

As permitted by applicable law, we intend to deliver only one copy of certain of our documents, including proxy statements, annual reports and information statements to stockholders residing at the same address, unless such stockholders have notified us of their desire to receive multiple copies thereof. Any such request should be directed to General Moly, Inc. Board of Directors, c/o Corporate Secretary, 1726 Cole Blvd., Suite 115 Lakewood, Colorado 80401 or (303) 928-8599. Upon request, we will promptly deliver a separate copy. Stockholders who currently receive multiple copies of the proxy statement at their address and would like to request householding of their communications should contact their broker.

Annual Report

The Company's Annual Report on Form 10-K (excluding exhibits) for the year ended December 31, 2016 was previously provided to all stockholders. An additional copy, including exhibits, will be furnished without charge to any stockholder by writing to the Corporate Secretary at the address above. The Company's Form 10-K may also be accessed at the Company's website at www.generalmoly.com, or at SEC's website at www.g

Other Matters

As of the date of this proxy statement, the Board is not aware of any matters that will be presented for action at the Special Meeting other than those described above. However, if other

Table of Contents

matters are properly brought before the Special Meeting, the proxies will be voted on those matters at the discretion of the proxy holders.

By Order of the Board of Directors,

Bruce D. Hansen

Chief Executive Officer and Chief Financial Officer

Lakewood, Colorado November 7, 2017

	Annex A
	Execution Version
INVESTMENT AND SECURITIES PURCHASE AGREEMENT	
BETWEEN	
GENERAL MOLY, INC.	
AND	
AMER INTERNATIONAL GROUP CO., LTD.	
April 17, 2015	
A-i	

Table of Contents

TABLE OF CONTENTS

ARTICLE	E I ISSUANCE OF SHARES; CLOSING	Page <u>A-1</u>
1.1	Purchase and Sale of Offered Securities; Closing Payment; Expense Reimbursement Account	
		<u>A-1</u>
<u>1.2</u>	Arrangement Fee	<u>A-2</u>
1.3	Closing	<u>A-2</u>
1.4	Closing Deliveries	<u>A-2</u>
	E II REPRESENTATIONS AND WARRANTIES OF THE COMPANY	
		<u>A-3</u>
<u>2.1</u>	Organization and Standing	110
		<u>A-3</u>
<u>2.2</u>	Capitalization	<u>A-3</u>
<u>2.3</u>	<u>Authorization; Enforceability</u>	<u>A-4</u>
<u>2.4</u>	No Violation; Consents.	<u>A-5</u>
<u>2.5</u>	Compliance with Laws.	<u>A-5</u>
<u>2.6</u>	SEC Reports; Financial Condition.	<u>A-6</u>
<u>2.7</u>	Absence of Changes	<u>A-7</u>
<u>2.8</u>	Books and Records	<u>A-8</u>
2.9	Securities Laws	<u>A-8</u>
2.10	No Default	<u>A-8</u>
2.11	Intellectual Property	<u>A-8</u>
2.12	No Litigation	<u>A-8</u>
2.13	Material Contracts	<u>A-8</u>
2.14	Permits Permits	<u>A-10</u>
2.15	<u>Subsidiaries</u>	A-10
2.16	Related Party Transactions	<u>A-10</u>
<u>2.17</u>	Securities Compliance	A-10
<u>2.17</u> <u>2.18</u>	Environmental Matters	<u>A-10</u> <u>A-10</u>
2.19	Taxes	A-10
<u>2.20</u>	Section 203 of the DGCL	<u>A-11</u>
<u>2.21</u>	Employees Fig. 1. Prof. (1)	<u>A-11</u>
<u>2.22</u>	Employee Benefits Park Park Park Park Park Park Park Park	<u>A-11</u>
2.23	Real Property	<u>A-11</u>
2.24	Tangible Assets	<u>A-12</u>
<u>2.25</u>	<u>Insurance Policies</u>	<u>A-12</u>
<u>2.26</u>	Brokers Fees	<u>A-12</u>
ARTICLE	E III REPRESENTATIONS, WARRANTIES AND COVENANTS OF PURCHASER	
<u>3.1</u>	Authorization; Enforceability; No Violations	<u>A-13</u>
<u>J.1</u>	Authorization, Emolecability, No violations	A-13
2.2	Committee Act Democratational Locanda	
<u>3.2</u>	Securities Act Representations; Legends Investment Decision by Purchaser	<u>A-13</u>
3.3	Compliance with Laws	<u>A-14</u>
3.4		<u>A-15</u>
3.5	Consents ENVERGETT ATION PROUTS	<u>A-15</u>
ARTICLE	E IV REGISTRATION RIGHTS	۸ 15
<u>4.1</u>	Shelf Registration	<u>A-15</u>
***	Maria Avegation VII	<u>A-15</u>
<u>4.2</u>	Company Obligations	A-15
<u>4.2</u> <u>4.3</u>	Registration Statement Effectiveness; Piggyback Registrations	A-15 A-16
<u>4.4</u>	Suspension A-ii	<u>A-18</u>

Table of Contents

		Page		
<u>4.5</u>	Current Public Information	<u>A-18</u>		
ARTICLE	E V CONDUCT OF BUSINESS PENDING THE CLOSING	<u>.</u>		
		<u>A-18</u>		
<u>5.1</u>	<u>General</u>			
		<u>A-18</u>		
<u>5.2</u>	Pre-Closing Covenants	<u>A-18</u>		
	E VI ADDITIONAL AGREEMENTS			
		<u>A-21</u>		
<u>6.1</u>	Agreement to Call Stockholder Meeting			
		<u>A-21</u>		
6.2	Proxy Statement; Other Commission Filings	<u>A-21</u>		
6.2 6.3 6.4 6.5 6.6 6.7 6.8	Listing Applications	A-22		
6.4	Assistance in Procuring Loan; Purchaser Option	<u>A-22</u>		
6.5	Consents	<u>A-22</u>		
6.6	Covenants Concerning the Parties.	<u>A-23</u>		
6.7	No Solicitation	<u>A-23</u>		
6.0	Change of Recommendation.	<u>A-23</u> <u>A-24</u>		
6.9	Certain Tax Covenants	<u>A-24</u> <u>A-25</u>		
6.10	Access to Information	<u>A-25</u>		
6.11	Further Assurances	<u>A-25</u>		
6.12	Publicity	<u>A-25</u>		
6.13	Board Representation	<u>A-25</u>		
ARTICLE	E VII CONDITIONS			
		<u>A-26</u>		
<u>7.1</u>	Closing			
		<u>A-26</u>		
ARTICLE	E VIII TERMINATION			
		<u>A-27</u>		
<u>8.1</u>	<u>Termination</u>			
		<u>A-27</u>		
8.2	Effect of Termination	<u>A-29</u>		
<u>8.3</u>	Payment of Company Break Fee	<u>A-29</u>		
ARTICLE	E IX MISCELLANEOUS			
		<u>A-29</u>		
<u>9.1</u>	<u>Notices</u>			
		<u>A-29</u>		
9.2	Entire Agreement	<u>A-30</u>		
9.3	<u>Amendments</u>	<u>A-31</u>		
9.3 9.4 9.5	Assignment	<u>A-31</u>		
<u>9.5</u>	Benefit	<u>A-31</u>		
9.6 9.7 9.8 9.9	Specific Performance	<u>A-31</u>		
9.7	Governing Law; Language	<u>A-31</u>		
9.8	Waiver of Jury Trial	<u>A-31</u>		
9.9	Severability	<u>A-31</u>		
9.10	Headings and Captions	<u>A-31</u>		
9.11	Certain Terms	<u>A-31</u>		
9.12	No Waiver of Rights, Powers and Remedies	<u>A-32</u>		
9.13	Fees and Expenses	<u>A-32</u>		
9.14	Counterparts	<u>A-32</u>		
<u>9.15</u>	Rules of Construction	<u>A-32</u>		
<u>9.16</u>	Dispute Resolution	A-33		
<u>,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,</u>		A-iii		

Table of Contents

Schedules Schedule 1 Certain Definitions Schedule 2 General Moly, Inc. Loan Summary of Principal Terms and Conditions Schedule 3 Regulatory Approvals Schedule 4 Dispute Resolution Exhibits Exhibit A Molybdenum Supply Agreement Term Sheet Exhibit B Form of Stockholder Agreement Exhibit C Form of Warrant Exhibit D Expense Reimbursement Agreement Term Sheet A-iv

INVESTMENT AND SECURITIES PURCHASE AGREEMENT

THIS INVESTMENT AND SECURITIES PURCHASE AGREEMENT (this "Agreement"), dated April 17, 2015 (the "Signing Date"), is between General Moly, Inc., a Delaware corporation (the "Company"), and Amer International Group Co. Ltd., a limited liability company organized under the laws of the People's Republic of China ("Purchaser"). The Company and Purchaser shall each be referred to herein as a "Party" and collectively as the "Parties". Capitalized terms used in this Agreement but not defined herein shall have the meanings ascribed to such terms in Schedule 1.

RECITALS

- A. The Company desires to issue and sell to Purchaser and Purchaser desires to acquire from the Company, on the terms and subject to the conditions in this Agreement, (1) 40,000,000 shares (the "*Offered Shares*") of the Company's common stock, par value \$0.001 per share (the "*Common Stock*") at the Per Share Price, and (2) warrants to purchase 80,000,000 shares of Common Stock (the "*Offered Warrants*") exercisable at the Per Share Price (the Offered Shares, together with the Offered Warrants, the "*Offered Securities*").
- B. The Offered Securities will be issued upon the satisfaction of the conditions set forth in *Section 7.1*. Upon the Loan Execution, the Offered Securities will constitute approximately 51% of the Company's fully diluted shares of Common Stock.
- C. Purchaser has agreed to endeavor with the Company to procure and support a loan from one or more Prime Chinese Banks to the Company to fund the Company's share of costs (including financing costs) related to the development of the Mt. Hope Project, or approximately US\$700 Million (the "*Loan*"), on substantially similar key terms and conditions as those outlined on *Schedule 2*, which Loan will be guaranteed by Purchaser.
- D. The Company desires to grant Purchaser an option, exercisable simultaneously with the Loan Execution, to enter into with the Company a molybdenum supply agreement for the purchase by Purchaser or an Affiliate of Purchaser of molybdenum produced at the Mount Hope Project (the "*Molybdenum Supply Agreement*"), substantially in accordance with the terms and conditions attached hereto as *Exhibit A*.
- E. At the Closing, the Company and Purchaser will enter into a stockholder agreement with respect to certain matters relating to the acquisition and disposition of the Offered Securities (and other shares of Common Stock, if any, owned by Purchaser or any of its Affiliates) and governance of the Company (the "Stockholder Agreement") substantially in the form attached hereto as Exhibit B.

In consideration of the mutual covenants contained in this Agreement, the Parties, intending to be legally bound, agree as follows:

AGREEMENT

ARTICLE I ISSUANCE OF SHARES; CLOSING

- 1.1 Purchase and Sale of Offered Securities; Closing Payment; Expense Reimbursement Account.
 - (a) Offer and Sale. On the terms and subject to the conditions of this Agreement and in reliance upon the representations and warranties contained herein, Purchaser shall purchase from the Company, and the Company shall issue and sell to Purchaser the Offered Securities.
 - (b) Closing Payment. In exchange for the Company's issuance and sale of the Offered Securities to Purchaser on the Closing Date, Purchaser shall pay to the Company in cash an amount equal to (i) the Stock Purchase Price, minus (ii) the Reimbursed Purchaser Expenses (such

Table of Contents

difference, the "Closing Payment"). In addition to making the Closing Payment, the Company shall pay Purchaser the Arrangement Fee following Closing pursuant to Section 1.2.

- (c) Reimbursement Account. On the Closing Date, the Company shall deposit \$3,000,000 into an account at a bank to be mutually agreed upon by the Company and Purchaser. Purchaser and the Company shall jointly hold such account pursuant to an expense reimbursement agreement (the "Expense Reimbursement Agreement"), substantially in accordance with the terms and conditions attached hereto as Exhibit D, to be entered into at Closing between Purchaser and the Company. Under the Expense Reimbursement Agreement, funds will be released to Purchaser, the Company or third parties to reimburse such parties for reasonable expenses incurred in connection with the procurement of the Loan under Section 6.4(a) (such expenses, collectively, the "Loan Procurement Expenses").
- **1.2** Arrangement Fee. Upon the initial draw of the Loan, the Company shall deliver or cause to be delivered to Purchaser, by wire transfer of immediately available funds in Dollars, (a) a fee equal to seventy-five one-hundredths of one percent (0.75%) of the committed amount under the Loan (the "Arrangement Fee"), minus (b) the Reimbursed Purchaser Expenses, minus (c) the Loan Procurement Expenses.
- **1.3** *Closing.* Subject to the satisfaction or waiver of the conditions set forth in *Section 7.1*, the completion of the purchase and sale of the Offered Shares (the "*Closing*") shall occur at 10:00 a.m. local time at the offices of Bryan Cave LLP, Denver, Colorado, on the second (2nd) Business Day after the satisfaction or waiver of the conditions set forth in *Section 7.1* (other than those that by their terms are to be satisfied or waived at the Closing), or at such other location, date and time as may be mutually agreed upon by the Company and Purchaser. The date of the Closing is referred to herein as the "*Closing Date*."

1.4 Closing Deliveries.

- (a) Company Deliveries. At the Closing, the Company shall deliver or cause to be delivered to Purchaser:
 - (i) a certificate or certificates representing the Offered Shares;
 - (ii) a receipt for the Closing Payment;
 - (iii) a duly executed warrant agreement (the "*Warrant Agreement*"), substantially in the form attached hereto as *Exhibit C*, evidencing the Offered Warrants, registered in the name of Purchaser;
 - (iv) a duly executed counterpart of the Stockholder Agreement;
 - (v) a duly executed counterpart of the Expense Reimbursement Agreement;
 - (vi) duly executed resignations effective as of the Closing Date from the directors on the Board pursuant to *Section 6.13*;
 - (vii) a good standing certificate (or its equivalent) for the Company issued by the Secretary of State of the State of Delaware and of such other applicable jurisdictions where the Company is qualified or licensed to do business or own, lease or operate property making such qualification or licensing necessary, dated as of a date within three (3) Business Days prior to the Closing Date;
 - (viii) copies, certified by the Secretary of the Company, of resolutions of the Board authorizing the execution and delivery of this Agreement and the other Transaction Documents, and in each case, such resolutions shall be in full force and effect and not revoked:

Table of Contents

- (ix) the documents, instruments and writings required to be delivered to Purchaser by the Company pursuant to Section 7.1(c):
 - (x) the APERAM Consent; and
- (xi) such other previously undelivered documents reasonably requested by Purchaser to be delivered by the Company to Purchaser at or prior to the Closing in connection with this Agreement or the other Transaction Documents to which the Company is a party.
- (b) Purchaser Deliveries. At the Closing, Purchaser shall deliver or cause to be delivered to the Company:
 - (i) the Closing Payment by wire transfer in immediately available funds in Dollars to an account specified by the Company in writing no less than three (3) Business Days prior to the Closing;
 - (ii) a duly executed counterpart of the Warrant Agreement;
 - (iii) a duly executed counterpart of the Stockholder Agreement;
 - (iv) a duly executed counterpart of the Expense Reimbursement Agreement;
 - (v) a written letter of intent, in substantially customary form, from a Prime Chinese Bank to the Company, indicating that such bank has formally endorsed the Loan and with the other banks (if any) will make the Loan on the key terms and conditions substantially set forth on *Schedule 2*:
 - (vi) the documents, instruments and writings required to be delivered to the Company by Purchaser pursuant to *Section 7.1(b)*; and
 - (vii) such other previously undelivered documents reasonably requested by the Company to be delivered by Purchaser to the Company at or prior to the Closing in connection with this Agreement or the other Transaction Documents to which Purchaser is a party.

ARTICLE II REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the disclosure schedule (the "*Disclosure Schedule*") delivered by the Company to Purchaser on or prior to the execution of this Agreement, the Company hereby represents and warrants to Purchaser on the Signing Date and the Closing Date as follows:

- **2.1** Organization and Standing. Each of the Company and its Subsidiaries is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization and has all requisite power and authority necessary for it to own or lease its properties and assets and to carry on its business as it is now being conducted (and, to the extent described therein, as described in the SEC Reports). Each of the Company and its Subsidiaries is duly qualified to transact business and is in good standing in each jurisdiction in which the character of the properties owned or leased by it or the nature of its businesses makes such qualification necessary, except where any failure to so qualify or be in good standing would not have a Material Adverse Effect. The copies of the Company's certificate of incorporation and bylaws that are listed as exhibits to the Company's Annual Report on Form 10-K for the year ended December 31, 2014 (the "10-K") are complete and correct copies thereof.
- **2.2** Capitalization. The authorized capital stock of the Company consists of 200,000,000 shares of Common Stock and 10,000,000 shares of preferred stock, par value \$0.001 per share ("Preferred Stock"). As of March 31, 2015, (i) 93,493,979 shares of Common Stock were issued and outstanding, and (ii) no shares of Common Stock were held in the treasury of the Company or by any Subsidiary of the Company. As of March 31, 2015, 257,779 shares of Common Stock were issuable (and such number

Table of Contents

was reserved for issuance) upon exercise of outstanding stock options granted pursuant to the Company's equity incentive plans filed with the SEC Reports (the "Plans"), restricted stock units covering 2,359,776 shares of Common Stock were issued under the Company's Plans; 9,535,000 shares of Common Stock were issuable (and such number was reserved for issuance) upon exercise of outstanding warrants to purchase Common Stock (the "Warrants") and up to 8,135,000 shares of Common Stock were issuable (and such number was reserved for issuance) upon conversion of convertible promissory notes (the "Convertible Notes"). As of the Signing Date, the Company had outstanding SARS with respect to not more than 1,803,146 shares of Common Stock as to which not more than zero shares of Common Stock would be issuable based on a price of Common Stock of \$0.50 per share. Since March 31, 2015, the Company has not issued any shares of its capital stock, or securities convertible into or exchangeable or exercisable for such capital stock, other than those shares of capital stock reserved for issuance as set forth in this Section 2.2. As of the Signing Date, no shares of Preferred Stock are issued and outstanding. The Company has no stock option, incentive or similar plan other than the Plans. All of the outstanding shares of capital stock of the Company have been duly and validly authorized and issued, and are fully paid and nonassessable. Except for the preemptive rights held by APERAM under the APERAM Securities Agreement, if any, all of the shares of Common stock subject to issuance under the Plans, Warrants, Convertible Notes and SARS, upon issuance prior to the Closing Date on the terms and conditions specified in the instruments pursuant to which they are issuable, will be duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights. The Offered Securities have been duly and validly authorized and when issued, sold and delivered by the Company in accordance with this Agreement, will be validly issued, fully paid and nonassessable. Except as set forth in this Section 2.2, there are no outstanding options, warrants, conversion rights, subscription rights, preemptive rights, rights of first refusal or other rights or agreements of any nature outstanding to subscribe for or to purchase any shares of Common Stock or any other securities of the Company of any kind binding on the Company. Except for the preemptive rights held by APERAM under the APERAM Securities Agreement, if any, the issuance by the Company of the Offered Securities is not subject to any preemptive rights, rights of first refusal or other similar limitation or any other claim, Lien, charge, encumbrance or security interest applicable to the assets of the Company, except those which have been waived. There are no restrictions upon the voting or transfer of any shares of Common Stock pursuant to the Company's certificate of incorporation or bylaws. There are no agreements or other obligations (contingent or otherwise) that may require the Company to repurchase or otherwise acquire any shares of Common Stock.

2.3 Authorization; Enforceability. The Company has the corporate power and authority to execute, deliver and perform this Agreement and the other Transaction Documents to which it is a party and has taken all necessary corporate action to authorize the execution, delivery and performance by it of, and the consummation of the transactions contemplated by, this Agreement and such other Transaction Documents, except Stockholder Approval. No other corporate proceeding on the part of the Company is necessary for the valid execution and delivery by the Company of this Agreement and the other Transaction Documents to which it is a party, and the performance and consummation by the Company of the transactions contemplated by this Agreement and such other Transaction Documents to be performed by the Company, except the Stockholder Approval or as has been obtained or waived. The Company has duly executed and delivered this Agreement and, when executed and delivered by it, will have duly executed and delivered the other Transaction Documents to which it is a party. Assuming the due execution and delivery of this Agreement by Purchaser, this Agreement constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar Laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or at law). Assuming the due execution and delivery of the Transaction Documents to which the Company is a party (other than this Agreement) by Purchaser, each of such

Table of Contents

Transaction Documents, when executed and delivered by the Company, will constitute a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar Laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or at law).

2.4 No Violation; Consents.

- (a) The execution, delivery and performance by the Company of this Agreement and the other Transaction Documents and the consummation of the transactions contemplated hereby and thereby to be performed by the Company do not and will not (i) assuming that all consents, approvals, authorizations and other actions described in *Section 2.4(b)* have been obtained and all filings and obligations described in *Section 2.4(b)* have been made, conflict with, violate or contravene the applicable provisions of any Law of any court or any federal or state government or political subdivision thereof or any agency or other entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government (a "*Governmental Authority*") to or by which the Company or any of its Subsidiaries or any of its or their respective assets is bound, (ii) violate, result in a breach of or constitute (with due notice or lapse of time or both) a default or give rise to an event of acceleration under, or give to others any right of termination, amendment or cancellation of, or give to others a right to require any payment to be made under, any contract, lease, license, permit, loan or credit agreement, mortgage, security agreement, trust indenture or other agreement or instrument to which the Company is a party or by which it or any of its Subsidiaries is bound or to which any of their respective assets is subject, nor result in the creation or imposition of any Lien, security interest, charge or encumbrance of any kind upon any of the assets or capital stock of the Company or any of its Subsidiaries, or (iii) conflict with or violate any provision of the Organizational Documents of the Company or any of its Subsidiaries, except in the case of each of clauses (i) and (ii) above as would not have a Material Adverse Effect.
- (b) No consent, approval, authorization or order of, or filing or registration with, any Governmental Authority or other Person is required to be obtained or made by the Company for the execution, delivery and performance of this Agreement and the other Transaction Documents or the consummation of any of the transactions contemplated hereby or thereby, except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act and state securities or "blue sky" laws, for any filings required to be made under the rules and regulations of the NYSE MKT and the Toronto Stock Exchange and for the Stockholder Approval, and (ii) where the failure to obtain such consent, approval, authorization or order or to make such filing or registration would not have a Material Adverse Effect.

2.5 Compliance with Laws.

- (a) Since January 1, 2012, (i) each of the Company and its Subsidiaries has complied, and is currently in compliance, with all Laws applicable to the Company or its Subsidiaries or to the operation of their respective businesses or to any assets owned or used by any of the respective businesses of the Company and its Subsidiaries, except where the failure to be in compliance would not reasonably be expected to have a Material Adverse Effect, and (ii) neither the Company nor any of its Subsidiaries has received any written notice that any investigation or review by any Governmental Authority with respect to the respective businesses of the Company and its Subsidiaries or to any assets owned or used by the respective businesses of the Company or its Subsidiaries is pending or that such investigation or review is contemplated.
- (b) The Company and its Subsidiaries (i) have been in compliance in all material respects with the United States Foreign Corrupt Practices Act of 1977 (the "FCPA") and any other

Table of Contents

applicable United States and foreign anti-corruption Laws, and (ii) since January 1, 2012, have not, to the knowledge of the Company, been investigated by any Governmental Authority with respect to, or been given written notice by a Governmental Authority or any other Person of, any actual or alleged violation by the Company or its Subsidiaries of the FCPA or any other applicable United States or foreign anti-corruption Laws.

- (c) To the Company's knowledge, since January 1, 2012, none of the Company or its Subsidiaries has, directly or indirectly through its representatives or any Person authorized to act on its behalf (including any distributor, agent, sales intermediary or other third party), offered, promised, paid, authorized or given, money or anything of value to any Person for the purpose of: (i) unlawfully influencing any act or decision of any Government Official, (ii) inducing any Government Official to do or omit to do an act in violation of a lawful duty, (iii) securing any improper advantage, or (iv) unlawfully inducing any Government Official to influence the act or decision of a government or government instrumentality, in order to obtain or retain business, or direct business to, any Person or entity, in any way.
- (d) To the Company's knowledge, since January 1, 2012, none of the Company or its Subsidiaries has had a customer or supplier or other business relationship with, is a party to any Contract with, or has engaged in any transaction with, any Person that is the subject of any international economic or trade sanction administered or enforced by the Office of Foreign Assets Control of the United States Department of the Treasury, the United Nations Security Council, the European Union, Her Majesty's Treasury, the United Kingdom Export Control Organization or other relevant sanctions authority.

2.6 SEC Reports; Financial Condition.

- (a) The Company has filed all forms, reports and documents required to be filed by the Company with the SEC since December 31, 2011 (the "SEC Reports"). The SEC Reports (including any financial statements filed as a part thereof or incorporated by reference therein) (i) at the time filed (or if amended or superseded by a filing prior to the Signing Date, then on the date of such subsequent filing), complied in all material respects with the applicable requirements of the Securities Act and the Exchange Act, as the case may be, and (ii) did not, at the time they were filed (or if amended or superseded by a filing prior to the date of this Agreement, then on the date of such subsequent filing), contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements in such Company SEC Reports, in the light of the circumstances under which they were made, not misleading.
- (b) The audited and unaudited consolidated financial statements of the Company and the related notes thereto contained in the SEC Reports (the "Company Financial Statements") present fairly the financial position of the Company and its Subsidiaries at such date and the results of operations of the Company and its Subsidiaries for the periods set forth therein; provided, however, that the unaudited financial statements are subject to normal year-end adjustments. The Company Financial Statements, including the related notes thereto, have been prepared in accordance with generally accepted accounting principles in the United States as in effect for the periods covered thereby.
- (c) Except and to the extent set forth on the consolidated balance sheet of the Company and its Subsidiaries as at December 31, 2014 included in the 10-K, neither the Company nor any of its Subsidiaries has any liability or obligation of any nature, except for (i) liabilities or obligations incurred in the ordinary course of business consistent with past practice since December 31, 2014, (ii) liabilities that are not otherwise required to be disclosed in the Company Financial Statements, (iii) liabilities incurred in compliance with the Eureka Budget, or (iv) liabilities that, individually or the aggregate, would exceed One Million Dollars (\$1,000,000).

Table of Contents

- **2.7** *Absence of Changes.* Except as set forth in the SEC Reports, since December 31, 2014, there has not been any change, effect, event, occurrence, state of facts or development which, individually or in the aggregate, has resulted in, or would reasonably be expected to result in, a Material Adverse Effect. Since December 31, 2014, each of the Company and its Subsidiaries has operated its business only in the ordinary course of business consistent with past practice and there has not been by or with respect to the Company or its Subsidiaries:
 - (a) any declaration, setting aside, or payment of any dividend or other distribution (whether payable in cash, stock, property or a combination thereof) with respect to the capital stock of the Company or its Subsidiaries (other than dividends paid by a wholly-owned Subsidiary of the Company to the Company or to any other wholly-owned Subsidiary of the Company) or entry into any agreement with respect to the voting of the capital stock of the Company or its Subsidiaries;
 - (b) (i) any increase in the compensation or benefits payable or to become payable to the directors, officers or employees of the Company or its Subsidiaries (except for increases in accordance with past practices in salaries or wages of employees of the Company or any of its Subsidiaries which are not across-the-board increases), (ii) any grant of rights to severance or termination pay to, or entry into any employment or severance agreement with, any director, officer or other employee of the Company or any of its Subsidiaries, or establishment, adoption, entry into or amendment of any collective bargaining, bonus, profit sharing, thrift, compensation, stock option, restricted stock, pension, retirement, deferred compensation, employment, termination, severance or other plan, agreement, trust, fund, policy or arrangement for the benefit of any director, officer or employee, or (iii) the taking of any affirmative action to amend or waive any performance or vesting criteria or accelerate vesting, exercisability or funding under any Employee Benefit Plan;
 - (c) (i) any acquisition (including, without limitation, by merger, consolidation, or acquisition of stock or assets) of any interest in any Person or any division thereof or any assets, other than acquisitions of assets in the ordinary course of business consistent with past practice and any other acquisitions for consideration that is individually not in excess of Two Hundred Fifty Thousand Dollars (\$250,000), or in the aggregate, not in excess of Five Hundred Thousand Dollars (\$500,000) for the Company and the Company Subsidiaries taken as a whole, (ii) any incurrence of any indebtedness for borrowed money or issuance of any debt securities or assumption, guarantee or endorsement, or otherwise as an accommodation assumption of responsibility for, the obligations of any Person (other than a wholly-owned Subsidiary of the Company) for borrowed money in a principal amount not, in the aggregate, in excess of One Million Dollars (\$1,000,000) for the Company and its Subsidiaries taken as a whole, (iii) any termination, cancelation or written request for any material change in, or agreement to any material change in, any Material Contract other than in the ordinary course of business consistent with past practice, (iv) other than as set forth in the Eureka Budget, the making or authorization of any capital expenditures that are, in the aggregate, in excess of Five Hundred Thousand Dollars (\$500,000) for the Company and its Subsidiaries taken as a whole;
 - (d) any change in accounting policies or procedures, other than in the ordinary course of business consistent with past practice or except as required by GAAP or by a Governmental Authority;
 - (e) any waiver, release, assignment, settlement or compromise of any material claims, or any material litigation or arbitration;
 - (f) any material tax election or settlement or compromise of any material liability for Taxes;
 - (g) any write up, write down or write off of the book value of any assets, individually or in the aggregate, for the Company and its Subsidiaries taken as a whole, in excess of One Million

Table of Contents

Dollars (\$1,000,000) except for depreciation and amortization in accordance with GAAP consistently applied; or

- (h) any (i) sale, pledge, disposition, transfer, lease, license, guarantee or encumbrance, or authorization of the sale, pledge, disposition, transfer, lease, license, guarantee or encumbrance, of any material property or assets (including Intellectual Property Rights) of the Company or any of its Subsidiaries, except for the sale or purchase of goods in the ordinary course of business consistent with past practice, or (ii) entry into any material commitment or material transaction outside the ordinary course of business consistent with past practice;
- (i) any agreements or commitments, whether oral or in writing, to take any action described in clauses (a) through (h) of this Section 2.7.
- **2.8** *Books and Records.* The books and records of the Company and its Subsidiaries have been maintained in accordance with the customary business practices of the Company and its Subsidiaries and in all material respects with Law. The Company has made available to Purchaser copies of the minute books of the Company in the Company's possession.
- **2.9** Securities Laws. All notices, filings, registrations, or qualifications under state securities or "blue sky" laws, that are required in connection with the offer, issuance, sale and delivery of the Offered Securities pursuant to this Agreement, have been, or will be, completed by the Company.
- **2.10** *No Default.* Subject to Stockholder Approval, the Company and its Subsidiaries are not, and, immediately after the consummation of the transactions contemplated hereby and by the other Transaction Documents to be performed by the Company, will not be, in default of (whether upon the passage of time, the giving of notice or both), (a) any term of its Organizational Documents, (b) any provision of any equity security issued by the Company, or of any agreement, instrument or other undertaking to which the Company or its Subsidiaries is a party or by which it or any of its properties or assets is bound, or (c) the applicable provisions of any Law of any Governmental Authority to or by which the Company or any of its Subsidiaries or any of its assets is bound, which default, in the cause of clauses (b) and (c) above, would either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 2.11 Intellectual Property. The Company and its Subsidiaries have all material patents, licenses, copyrights and trademarks that are needed to conduct the business of the Company and its Subsidiaries as it is now being conducted (the "Intellectual Property Rights"). To the Company's knowledge the Intellectual Property Rights that the Company (or any of its Subsidiaries) owns are valid and enforceable. To the Company's knowledge the use of such Intellectual Property Rights by the Company (or any of its Subsidiaries) does not infringe upon or conflict with any license, copyright or trademark of any third party, and neither the Company nor any of its Subsidiaries has received written notice of any such infringement or conflict other than with respect to alleged infringements or conflicts. The Company has no knowledge of any infringement of its Intellectual Property Rights by any third party.
- **2.12** *No Litigation*. Except as disclosed in the SEC Reports, no Action against the Company or any of its Subsidiaries is pending, or, to the Company's knowledge, threatened or contemplated that, if determined adversely, would, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **2.13** *Material Contracts*. Except as listed or described on the SEC Reports, as of the Signing Date, neither the Company nor any of its Subsidiaries is a party to or bound by any Contract of any of the types described below:
 - (a) any consulting agreement or employment agreement that provides for annual compensation to a Person exceeding Two Hundred Thousand Dollars (\$200,000) per year and

Table of Contents

which cannot be terminated by the Company or Subsidiary party thereto without penalty on notice of sixty (60) days or less;

- (b) any Contract for capital expenditures or the acquisition of fixed assets in excess of One Million Dollars (\$1,000,000);
- (c) any Contract for the purchase, maintenance or acquisition of materials, supplies, merchandise, equipment, parts or other property or services (other than an ongoing license for, or for support or maintenance of, Software) which will extend over a period of more than twelve (12) months or require remaining aggregate future payments in excess of One Million Dollars (\$1,000,000);
- (d) any Contract that restricts the right of the Company or any Subsidiary to engage in any line of business, compete with any Person or provide any service to any Person in any geographic area;
- (e) any lease pertaining to any leased real or personal property that provides for a future liability in excess of Two Hundred Fifty Thousand Dollars (\$250,000) per year;
- (f) any Contract relating to the acquisition or disposition of any business, material asset or real property requiring aggregate payments in the future in excess of Five Hundred Thousand Dollars (\$500,000) or which contains any material continuing obligations on the part of the Company or Subsidiary party thereto;
- (g) any Contract relating to the borrowing of money, or the guaranty of another Person's borrowing of money or other obligation, including all notes, mortgages, indentures and other obligations, guarantees of performance, agreements and instruments for or relating to any lending or borrowing, including any indebtedness, in excess of Five Hundred Thousand Dollars (\$500,000);
- (h) any Contract granting any Person a Lien on any assets or properties of the Company or its Subsidiaries, other than Permitted Liens, where the underlying liability is in excess of Five Hundred Thousand Dollars (\$500,000);
- (i) any Contract relating to the development, ownership, licensing or use of any intellectual property rights material to the business of the Company or any Subsidiary other than non-exclusive, end-user licenses for commercially available prepackaged Software with license, maintenance, support and other fees of less than One Hundred Thousand Dollars (\$100,000) per year;
 - (j) any bargaining agreement or other Contract with any labor organization, union, association or works council;
 - (k) any Contract with any Governmental Authority; or
 - (l) any partnership, joint venture or other similar agreements or arrangements.

Copies of each Material Contract have been made available to Purchaser by the Company or are available in the SEC Reports, and such copies are correct and complete as of the Signing Date. Each Material Contract is in full force and effect, and represents a valid and binding obligation of the Company or one of its Subsidiaries, as applicable, and, to the Company's knowledge, the other parties thereto, enforceable against the Company or one of its Subsidiaries, as applicable, in accordance with its terms, except as enforcement thereof may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar Laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or at law). Neither the Company, any Subsidiary nor, to the Company's knowledge, any other party to any Material Contract is in material breach of or material

Table of Contents

default under any Material Contract. Neither the Company nor any Subsidiary has received any written notice of termination of, or dispute under, any Material Contract.

- **2.14** *Permits*. Each of the Company and its Subsidiaries is in possession of all franchises, grants, authorizations, licenses, permits, easements, variances, exemptions, consents, certificates, approvals and orders necessary to own and lease its properties and to carry on its business as it is now being conducted (collectively, the "*Company Permits*"), and, except as disclosed in the SEC Reports, all such Company Permits are valid, and in full force and effect, and there is no action pending or, to the knowledge of the Company, threatened, regarding suspension or cancellation of any of the Company Permits, except for such Company Permits which the failure to possess or to be valid or in full force and effect, or of which the cancellation or suspension would not, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Neither the Company nor any of its Subsidiaries is in conflict with, or in default or violation of, any of the Company Permits, which conflict, default or violation would, either individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
 - 2.15 Subsidiaries. As of the Signing Date, the Company has no subsidiaries other than those set forth in the SEC Reports.
- **2.16** *Related Party Transactions.* None of the officers, directors, employees or shareholders of the Company is presently a party to any transaction with the Company or any of its Subsidiaries (other than for services as employees, officers and directors), including any Contract providing for the furnishing of services to or by, providing for rental of real or personal property to or from, or the advances of money or otherwise requiring payments to or from any such officer, director, employee or shareholder or, to the knowledge of the Company, any corporation, partnership, trust or other entity in which any such officer, director, employee or shareholder has a substantial interest or is an officer, director, trustee or partner.
- **2.17** Securities Compliance. The Common Stock is registered pursuant to Section 12(b) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and listed on NYSE MKT and the Toronto Stock Exchange. The Company is in material compliance with all NYSE MKT and Toronto Stock Exchange requirements, and the Company has not been contacted by NYSE MKT or the Toronto Stock Exchange, either orally or in writing, concerning any violations or any potential removal of the Common Stock from NYSE MKT or the Toronto Stock Exchange.

2.18 Environmental Matters.

- (a) None of the Company nor any of its Subsidiaries is in violation, in any material respect, of any Environmental Law, and the Company has no knowledge of any event or condition that exists or has occurred that is reasonably likely to result in any material violation of any Environmental Law;
- (b) there are no claims against the Company or any of its Subsidiaries arising under any Environmental Law or concerning the Release of or exposure of persons to any Hazardous Materials; and
- (c) each of the Company and its Subsidiaries possesses all material Environmental Permits required to operate their respective businesses.

2.19 Taxes.

(a) Each of the Company and its Subsidiaries has timely filed or caused to be filed all Federal Tax Returns and all material foreign, state and local Tax Returns required to have been filed by it, each such Tax Return is true, correct and complete in all material respects and all Taxes required to be paid by the Company and its Subsidiaries with respect to the periods covered by such returns or otherwise due have been paid, except any such Tax the validity or amount of which

Table of Contents

is being contested in good faith by appropriate proceedings and as to which the Company has set aside on its books adequate reserves with respect thereto in accordance with generally accepted accounting principles in the United States.

- (b) Neither the Company nor any of its Subsidiaries has received any Tax assessment, written notice of audit, written notice of proposed adjustment or written deficiency notice from any taxing authority, and to the knowledge of the Company, no basis exists for any such Tax assessment, adjustment or deficiency notice.
- (c) To the knowledge of the Company, the Company and its subsidiaries have withheld and paid (or have caused to be withheld and paid on their behalf) all Taxes required to have been withheld and paid in connection with amounts paid or owing to any third party.
- (d) Neither the Company nor any of its Subsidiaries has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency.
- **2.20** Section 203 of the DGCL. Assuming the accuracy of the representations and warranties of Purchaser set forth in Section 3.2(c), the Board has taken all actions necessary or advisable to ensure that Section 203 of the General Corporation Law of the State of Delaware (the "DGCL") does not apply to any of the transactions contemplated by this Agreement (including the purchase of the Offered Securities hereunder).
- **2.21** *Employees*. Neither the Company nor any of its Subsidiaries is a party to or bound by any collective bargaining agreement, nor has any of them experienced any strike or material grievance, claim of unfair labor practices, or other collective bargaining dispute within the past three years. Neither the Company nor any of its Subsidiaries has committed any material unfair labor practice and the Company has no knowledge of any organizational effort presently being made or threatened by or on behalf of any labor union with respect to employees of the Company or any of its Subsidiaries. The Company and its Subsidiaries have complied in all material respects with all applicable Laws governing employment or otherwise relating to the employees of the Company and its Subsidiaries including all applicable Laws relating to labor relations, equal employment opportunity and nondiscrimination, wages and hours, immigration and occupational safety and health.
- **2.22** Employee Benefits. Neither the Company nor any of its Subsidiaries, nor any ERISA Affiliate maintains, sponsors, contributes to, has any obligation to contribute to, or has any liability under or with respect to any: (a) "multiemployer plan" within the meaning of Section 3(37) of ERISA; (b) Pension Plan that is subject to Title IV of ERISA or Section 412 of the Code; or (c) Employee Benefit Plan providing health or life insurance or other welfare-type benefits for current or future retired or terminated employees (or any spouse or other dependent thereof) of the Company or any of its Subsidiaries, except as required by Section 601 of ERISA, Section 4980B of the Code or analogous state Law.

2.23 Real Property.

(a) Except as set forth in the SEC Reports, the Company does not own any real property with a book value in excess of Five Hundred Thousand Dollars (\$500,000). Section 2.23(a) of the Disclosure Schedule contains a list of all the addresses or general location of all real property leased or primarily used by the Company, in each case as of the Signing Date providing for aggregate annual rental payments in excess of One Hundred Thousand Dollars (\$100,000) (the "Leases"). With respect to all real property owned or leased or primarily used by the Company with a book value in excess of Five Hundred Thousand Dollars (\$500,000) (the "Real Property"), the Company has quiet possession thereof, and, with respect to leased Real Property, has valid leasehold interests providing rights to use such Real Property, free and clear of all Liens other than Permitted Liens, but subject to the applicable lease agreements. The Company has not

Table of Contents

received written notice of any pending or threatened condemnation proceeding, or of any sale or other disposition in lieu of condemnation, affecting any of the Real Property.

- (b) There are no leases, subleases, licenses, concessions or other agreements granting to any party or parties other than the Company the right of use or occupancy of any portion of, or any interest in, any of the Real Property, and there are no outstanding options or rights of first refusal to purchase any of the Real Property in favor of the Company, and to the knowledge of the Company, there are no outstanding options or rights of first refusal to purchase any of the Real Property in favor of any third party. No Real Property is used for any material purpose other than the conduct of the Company's business.
- (c) The Company has made available to Purchaser a true, correct and complete copy of each of the Leases not otherwise included in the SEC Reports, and (i) each Lease is legal, valid, binding and enforceable against the Company and to the knowledge of the Company, against the other parties thereto, except as may be limited by applicable bankruptcy, insolvency or similar legal requirements affecting creditors' rights generally or by general equitable principles, (ii) neither the Company nor, to the knowledge of the Company, any other party to any Lease, has waived any material term or condition thereof, and all covenants under any Lease to be performed by the Company have been performed in all material respects, and to the knowledge of the Company, all covenants under any Lease to be performed by any other party to any Lease, have been performed in all material respects, (iii) neither the Company, nor, to the knowledge of the Company, any other party to any Lease, is in breach or default under such Lease in any material respect, and (iv) the Company has not collaterally assigned or granted any security interest in any Lease or any interest therein.
- **2.24** *Tangible Assets.* The Company and its Subsidiaries have good and valid title to or good and valid leasehold interests in all material items of tangible properties and assets owned or leased by the Company or such Subsidiary and reflected in the Company Financial Statements (the "*Tangible Assets*"), free and clear of all Liens, other than Permitted Liens. The Tangible Assets are in good operating condition (normal wear and tear excepted), and are fit in all material respects for use in the ordinary course of business.
- **2.25** *Insurance Policies*. The Company has made available to Purchaser copies of the declaration pages of all material Insurance Policies, and such copies are correct and complete and have not been amended. Each of the Insurance Policies is in full force and effect. There has been no material claim made under an Insurance Policy at any time during the twelve (12) months ending on the Signing Date. None of the Company or any of its Subsidiaries has received written notice under any Insurance Policy denying or disputing any claim (or coverage with respect thereto) made by the Company or its Subsidiaries regarding the termination, cancellation or material amendment of, or material premium increase with respect to, any Insurance Policy, in each case, at any time during the twelve (12) months ending on the Signing Date.
- **2.26** *Brokers Fees.* No broker, investment banker, financial advisor, finder or similar intermediary has acted for or on behalf of, or is entitled to any brokers', finders or similar fee or other commission from the Company or any of its Affiliates in connection with this Agreement or the transactions contemplated hereby.

Table of Contents

ARTICLE III REPRESENTATIONS, WARRANTIES AND COVENANTS OF PURCHASER

Purchaser hereby acknowledges, represents, warrants and agrees as follows:

3.1 Authorization; Enforceability; No Violations.

- (a) Purchaser is duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and has all requisite power and authority to execute, deliver and perform the terms and provisions of this Agreement and the other Transaction Documents to which it is a party and has taken all necessary action to authorize the execution, delivery and performance by it of, and the consummation of the transactions contemplated by, this Agreement and such other Transaction Documents.
- (b) The execution, delivery and performance by Purchaser of this Agreement and the other Transaction Documents to which it is a party and the consummation by Purchaser of the transactions contemplated hereby or thereby to be performed by it do not and will not violate any provision of (i) its Organizational Documents, or (ii) any law, statute, rule, regulation, order, writ, injunction, judgment or decree to which it is subject, except in the case of clause (ii), as would not prevent or materially delay the consummation of the transactions contemplated by this Agreement or the other Transaction Documents. Purchaser has duly executed and delivered this Agreement. Assuming the due execution and delivery hereof by the Company, this Agreement constitutes the legal, valid and binding obligation of Purchaser enforceable against Purchaser in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or at law). Assuming the due execution and delivery of the other Transaction Documents (other than this Agreement) by the other parties thereto, each of such Transaction Documents, when executed and delivered by Purchaser will constitute a legal, valid and binding obligation of Purchaser, enforceable against Purchaser in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether enforcement is sought in a proceeding in equity or at law).

3.2 Securities Act Representations; Legends.

- (a) Purchaser understands and agrees that: (i) the offering and sale of the Offered Securities to be issued and sold hereunder are intended to be exempt from the registration requirements of the Securities Act of 1933, as amended (the "Securities Act"), (ii) the initial offer and sale of the Offered Securities issuable hereunder have not been registered under the Securities Act or any other applicable securities laws and such securities only may be transferred or otherwise resold in accordance with the provisions of Regulation S or Rule 144 under the Securities Act, pursuant to an effective registration statement under the Securities Act and any other applicable securities laws or if an exemption from such registration requirements is available, and (iii) the Company is required to register any resale of the Offered Securities, if any, under the Securities Act and any other applicable securities laws only to the extent provided in this Agreement.
- (b) Purchaser represents that the Offered Securities to be acquired by Purchaser pursuant to this Agreement are being acquired for its own account and not with a view to, or for sale in connection with, any distribution thereof or in violation of the Securities Act or any other securities laws that may be applicable.
- (c) Purchaser represents that, prior to the consummation of the transactions contemplated by this Agreement or the other Transaction Documents to which it is a party, it is not an Affiliate of

Table of Contents

the Company. Neither Purchaser nor any of its "affiliates" or "associates" (as defined in Section 203 of the DGCL) is, nor at any time during the past three (3) years has been, an "interested stockholder" of the Company (as defined in Section 203 of the DGCL). Purchaser does not own directly or indirectly, and has not at any time during the past three years owned, beneficially or otherwise, any shares of Common Stock.

- (d) Purchaser acknowledges that no representations have been made to Purchaser by or on behalf of the Company in connection with the offering and sale of the Offered Securities hereunder other than those as set forth herein, and Purchaser represents that it is not subscribing for the Offered Securities as a result of, or in response to, any advertisement, article, notice or other communication published in any newspaper, magazine or similar media or broadcast over television or radio, or presented at any seminar or meeting.
- (e) Purchaser has had the opportunity to read the SEC Reports and has been afforded the opportunity to ask questions of the Company. Purchaser understands that its investment in the Offered Securities is speculative and involves a high degree of risk. Purchaser acknowledges that it has carefully evaluated the merits and risks of such an investment, including the risk factors set forth in the SEC Reports.
- (f) Purchaser acknowledges that the Offered Securities will be endorsed with restrictive legends, substantially in the following form, that prohibit their transfer except in accordance therewith: "THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "1933 ACT"), OR ANY STATE SECURITIES LAWS. THESE SECURITIES MAY NOT BE OFFERED, SOLD, PLEDGED, HYPOTHECATED OR OTHERWISE TRANSFERRED IN THE ABSENCE OF (I) SUCH REGISTRATION OR (II) AN EXEMPTION THEREFROM AND, IF REQUESTED BY THE COMPANY, AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY TO THE EFFECT THAT SUCH REGISTRATION IS NOT REQUIRED OR (III) IN ACCORDANCE WITH THE PROVISIONS OF REGULATION S UNDER THE 1933 ACT" and, if applicable, "THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A STOCKHOLDER AGREEMENT DATED AS OF [], 2015, COPIES OF WHICH ARE AVAILABLE FROM GENERAL MOLY, INC. UPON REQUEST, AND ANY SALE, PLEDGE, HYPOTHECATION, TRANSFER, ASSIGNMENT OR OTHER DISPOSITION OF SUCH SECURITIES IS SUBJECT TO SUCH STOCKHOLDER AGREEMENT". Purchaser agrees that the Offered Securities are only transferable on the books of the Company in accordance with, and that the Company will refuse to register any transfer of the Offered Securities not made in accordance with, the restrictions set forth in restrictive legends to which they are subject, and the Stockholder Agreement, if applicable.
- (g) Purchaser is (i) an "accredited investor" within the meaning of Rules 501(a)(1), (a)(2), (a)(3), (a)(7) or (a)(8) under the Securities Act, or (ii) a "qualified institutional buyer" as defined in Rule 144A under the Securities Act.
- (h) Purchaser either alone or with the assistance of its professional advisors, is a sophisticated investor and has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of an investment in the Offered Securities and of making an informed investment decision and understands and has fully considered for purposes of this investment the risk of loss of all monies invested in the Company.
- 3.3 Investment Decision by Purchaser. Purchaser understands that nothing in this Agreement or any other materials presented to it in connection with the purchase and sale of the Offered Securities constitutes legal, tax or investment advice. Purchaser has consulted such legal, tax and investment advisors as it, in its sole discretion, has deemed necessary or appropriate in connection with its purchase of the Offered Securities.

Table of Contents

- 3.4 Compliance with Laws. Purchaser (i) has been in compliance in all material respects with the FCPA and any other applicable United States and foreign anti-corruption Laws, and (ii) since January 1, 2012, has not, to the knowledge of Purchaser, been investigated by any Governmental Authority with respect to, or been given written notice by a Governmental Authority or any other Person of, any actual or alleged violation by the Company or its Subsidiaries of the FCPA or any other applicable United States or foreign anti-corruption Laws.
- **3.5** *Consents.* Subject to the accuracy of the Company's representations and warranties herein, no material consent, approval, authorization or order of, or filing or registration with, any Governmental Authority or other Person is required to be obtained or made by Purchaser for the execution, delivery and performance of this Agreement or the consummation of any of the transactions contemplated hereby except for the approval of the Governmental Authorities set forth on *Schedule 3*.

ARTICLE IV REGISTRATION RIGHTS

- **4.1** Shelf Registration. On or before the date that is nine months following the Closing Date, the Company shall file a registration statement with the SEC to effect the registration of the Registrable Securities under the Securities Act (such registration statement and the prospectus included therein being referred to as the "Registration Statement") for a public offering of Common Stock then beneficially owned by Purchaser or any of its Affiliates or issuable to Purchaser or any of its Affiliates upon exercise of any option, warrant or other security convertible into or exercisable for Common Stock (the "Registrable Securities"). Such offering shall be made on a continuous basis pursuant to Rule 415 under the Securities Act ("Rule 415"). If Rule 415 limits the number of Registrable Securities permitted to be registered on a Registration Statement otherwise required to be filed by the Company hereunder, the Company shall promptly file an additional Registration Statement covering any Registrable Securities excluded from such prior Registration Statement. The Company shall also use reasonable best efforts to cause such Registrable Securities to be qualified in such jurisdictions as Purchaser may reasonably request.
 - **4.2** Company Obligations. In connection with the Registration Statement, the Company shall:
 - (a) prepare and file with the SEC such amendments and supplements to the Registration Statement and the prospectus used in connection with the Registration Statement, and such documents and reports to be incorporated by reference into the Registration Statement, as necessary to comply with the provisions of the Securities Act with respect to the disposition of the Registrable Securities;
 - (b) furnish to Purchaser such number of copies of Registration Statements and prospectuses and other documents incident thereto, including any amendment of or supplement to the prospectus, as Purchaser may from time to time reasonably request;
 - (c) promptly furnish to Purchaser copies of any comments that the SEC provides in writing to the Company pertaining to a Registration Statement, and any responses thereto from the Company to the SEC;
 - (d) promptly provide notice to Purchaser when a Registration Statement or any post-effective amendment thereto the same has become effective:
 - (e) use its reasonable best efforts to qualify the Registrable Securities for offer and sale under such other securities or blue sky laws of such jurisdictions in the United States as Purchaser reasonably requests;

Table of Contents

- (f) use its reasonable best efforts to cause all such Registrable Securities to be listed on NYSE MKT or the Toronto Stock Exchange or any other applicable securities exchange or quoted on each inter-dealer quotation system on which the Common Stock is then listed or quoted;
- (g) pay all expenses incurred in connection with such registration, including registration and filing fees with the SEC, reasonable fees and expenses of counsel and other advisors to Purchaser and the Company, printers' and accountants' fees, fees and expenses of compliance with securities or blue sky laws and fees and expenses incurred in connection with the listing or quotation of the Registrable Securities; *provided*, *however*, that any underwriting discounts, underwriting commissions, or underwriting fees attributable to the sale of the Registrable Securities shall be borne by Purchaser;
- (h) enter into customary agreements (including underwriting agreements in customary form) if requested by Purchaser, including representations and warranties by the Company and other terms and provisions that are customarily contained in underwriting agreements generally with respect to secondary distributions, including customary lock up provisions, indemnification and contribution provisions in favor of the underwriters and customary agreements as to the provision of opinions of counsel and accountants' letters; and
- (i) otherwise cooperate with Purchaser, any underwriters, the SEC and other regulatory agencies and take all actions and execute and deliver or cause to be executed and delivered all documents necessary to effect the registration of any Registrable Securities.

4.3 Registration Statement Effectiveness; Piggyback Registrations.

- (a) The Company shall use reasonable best efforts to have the Registration Statement declared effective under the Securities Act as promptly as practicable after filing thereof with the SEC. The Company shall use reasonable best efforts to cause the Registration Statement to continue to be effective until the date that Purchaser has either disposed of or has the ability to dispose of all Registrable Securities without any volume or manner of sale restrictions pursuant to Rule 144 of the Securities Act ("Effective Period"), and, during such period, to cause the Registration Statement and the prospectus contained therein to be updated as reasonably deemed necessary by the Company or required by the Securities Act or the Exchange Act to enable Purchaser to resell the Registrable Securities.
- (b) If at any time during the Effective Period, the Company shall determine to prepare and file with the SEC a registration statement relating to an offering for its own account or the account of others under the Securities Act of any of its equity securities (other than on Form S-4 or Form S-8 (each as promulgated under the Securities Act) or their then equivalents relating to equity securities to be issued solely in connection with any acquisition of any entity or business or equity securities issuable in connection with the stock option or other employee benefit plans) then, not fewer than twenty (20) Business Days prior to the effective date of such registration statement, the Company shall send to Purchaser a written notice of such determination and, if within ten (10) Business Days after the date of such notice, Purchaser shall so request in writing, the Company shall include in such registration statement all or any part of such Registrable Securities Purchaser requests to be registered, *provided* that Purchaser agrees to the same terms and conditions regarding method of sale applicable to the securities otherwise being sold through such registration. If such registration statement relates to an underwritten public offering and the underwriter of such proposed offering advises the Company and Purchaser that, in its opinion, the number of securities requested to be included in the registration statement (including securities to be sold by Purchaser or any other security holder) exceeds the number which can be sold in such offering within an acceptable price range, then the Company shall include in such registration statement first the registrable securities required to be registered pursuant to a request under the

Table of Contents

APERAM Securities Agreement, second the Registrable Securities Purchaser proposes to register, and third any securities the Company or any other security holder proposes to register.

- (c) Promptly upon any registration statement filed pursuant to this *Section 4.3* being declared effective by the SEC, the Company will file a related form of final prospectus pursuant to Rule 424(b) promulgated under the Securities Act.
- (d) Purchaser agrees to indemnify (to the fullest extent permitted by applicable law) the Company, its officers, directors, employees and agents and each underwriter and selling broker, if any, and each person, if any, who controls the Company (within the meaning of the Securities Act), against liabilities, losses, claims, damages, actions or expenses (including, in each case, under the Securities Act or the Exchange Act) arising by reason of any statement contained in a registration statement (including any Registration Statement), or any amendment or supplement thereto, that Purchaser provided to the Company in writing explicitly for use in such registration statement, being actually or allegedly false or misleading or actually or allegedly omitting to state a material fact necessary to be stated in order that the statements made in such registration statement, in the circumstances in which they are made, not be misleading; provided that in no event will the aggregate amount Purchaser be required to pay pursuant to such indemnification obligations exceed the greater of the aggregate purchase price paid by Purchaser hereunder and the amount of the net proceeds received by Purchaser upon the sale of the Registrable Securities giving rise to such indemnification obligation. The Company hereby agrees to indemnify (to the fullest extent permitted by applicable law) Purchaser, its officers, directors, employees and agents and each underwriter and selling broker, if any, and each person, if any, who controls Purchaser (within the meaning of the Securities Act) against liabilities, losses, claims, damages, actions or expenses (including, in each case, under the Securities Act or the Exchange Act) arising by reason of (i) any statement (other than a statement provided by Purchaser as described above) in or incorporated by reference in a registration statement (including any Registration Statement), or any amendment or supplement thereto, being actually or allegedly false or misleading or actually or allegedly omitting to state a material fact necessary to be stated in order that the statements made in or incorporated by reference in such registration statement, in the circumstances in which they are made, not be misleading, or (ii) any actual or alleged violation by the Company of the Securities Act, the Exchange Act, any state securities laws or any rule or regulation promulgated under the Securities Act, the Exchange Act or any state securities laws in connection with a registration statement.
- (e) To the extent a claim for indemnification under this *Section 4.3* is unavailable (by reason of public policy or otherwise) or insufficient to hold harmless an indemnified party in respect of any losses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, in such proportion as is appropriate to reflect the relative fault of the indemnifying party and indemnified party as well as any other relevant equitable considerations. The relative fault of such indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission of a material fact, was taken or made by, or relates to information supplied by, such indemnifying party or indemnified party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such action, statement or omission. The amount paid or payable by a party as a result of any losses shall be deemed to include, subject to the limitations set forth herein, any reasonable attorneys' or other reasonable fees or expenses incurred by such party in connection with any proceeding to the extent such party would have been indemnified for such fees or expenses if the indemnification provided for herein had been available to such party in accordance with its terms.

Table of Contents

- (f) The Parties hereby acknowledge that they are sophisticated business persons who were represented by counsel during the negotiations regarding the provisions hereof, including the provisions of this *Section 4.3*, and are fully informed regarding such provisions.
- **4.4** Suspension. Upon delivery of a notice (a "Suspension Notice") to Purchaser, the Company may suspend the use of any Registration Statement if: (a) in the good faith and reasonable judgment of the Board, after consultation with counsel, such suspension is necessary to delay disclosure of material non-public information that would be seriously detrimental to the Company, and the Board concludes, as a result, that it is in the best interest of the Company to suspend use of the Registration Statement at such time, and (b) the Company furnishes to Purchaser a certificate signed by the Chief Executive Officer of the Company stating that in the good faith judgment of the Board, it would be seriously detrimental to the Company to disclose such material non-public information and that it is, therefore, in the best interest of the Company to suspend availability of the Registration Statement at such time; provided, however, that (i) the Company shall have the right to suspend use of the Registration Statement for a period (a "Blackout Period") of not more than (A) twenty (20) consecutive trading days, and (B) an aggregate of forty-five (45) days during any twelve (12) month period, (ii) the Company shall not defer its obligation in this manner more than two times during any 12-month period, and (iii) the Effective Period shall be extended for the amount of time that the Registration Statement is unavailable due to such a deferral. Upon receipt of a Suspension Notice, Purchaser shall discontinue disposition of Registrable Securities pursuant to a Registration Statement until such Blackout Period has ended. The Company shall be permitted to enter stop transfer instructions with the Company's transfer agent with respect to the Registrable Securities during any Blackout Period.
- **4.5** *Current Public Information*. As long as Purchaser owns any Registrable Securities that are not otherwise eligible for sale as contemplated by Rule 144 under the Securities Act, the Company shall use reasonable best efforts to file all required reports with the SEC, or otherwise make available "adequate current public information" about itself, within the meaning of Rule 144(c) under the Securities Act, to potentially make available to Purchaser the benefits of certain rules and regulations of the SEC which may permit the sale of the Registrable Securities without registration.

ARTICLE V CONDUCT OF BUSINESS PENDING THE CLOSING

- **5.1** *General.* The Company agrees that, between the Signing Date and the earlier of (x) the Closing and (y) the expiration or termination of this Agreement (the "*Pre-Closing Period*"), except as specifically permitted by any other provision of this Agreement:
 - (a) the business of the Company and its Subsidiaries shall be conducted in, and the Company and its Subsidiaries shall not take any action except in, the ordinary course of business consistent with past practice; and
 - (b) subject to *Section 5.2(f)*, the Company shall use its reasonable best efforts to keep available the services of the officers and key employees of the Company and its Subsidiaries and to preserve relationships with vendors with which the Company or any of its Subsidiaries has significant business relationships.
- **5.2** *Pre-Closing Covenants*. Except as specifically permitted by any other provision of this Agreement or as set forth in the Disclosure Schedule, the Company shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, do, or agree to do, any of the following during the Pre-Closing Period without the prior written consent of Purchaser:
 - (a) amend or otherwise change its certificate of incorporation or by-laws or equivalent organizational documents;

Table of Contents

- (b) (i) issue, sell, pledge, dispose of, grant, transfer, encumber, or authorize the issuance, sale, pledge, disposition, grant, transfer or encumbrance of any shares of capital stock of, or other equity interests in, the Company or any of its Subsidiaries of any class, or securities convertible or exchangeable or exercisable for any shares of such capital stock or other equity interests, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or other equity interests or such convertible or exchangeable securities, or any other ownership interest (including, without limitation, any such interest represented by contract right) of the Company or any of its Subsidiaries, other than (A) the issuance of shares of Common Stock upon the exercise of options, warrants and convertible notes outstanding as of the date hereof in accordance with their terms and (B) issuances of securities in connection with existing contractual preemptive rights, *if any*, under Section 6.2(a) of the APERAM Securities Agreement; *provided*, *however*, that in the case of (B) the number of Offered Shares and Offered Warrants shall increase on a *pro rata* basis so that Purchaser will receive Offered Securities constituting the same percentage of shares of Common Stock on a fully diluted basis that Purchaser would have received had such issuances not occurred, (ii) sell, pledge, dispose of, transfer, lease, license, guarantee or encumber, or authorize the sale, pledge, disposition, transfer, lease, license, guarantee or encumbrance of, any material property or assets (including Intellectual Property Rights) of the Company or any of its Subsidiaries, except pursuant to existing contracts or commitments or the sale or purchase of goods in the ordinary course of business consistent with past practice, or (iii) enter into any material commitment or material transaction outside the ordinary course of business consistent with past practice:
- (c) other than dividends or distributions from Eureka Moly to the Company that are permitted under the Eureka Moly LLC Agreement and approved by the manager of Eureka Moly, declare, set aside, make or pay any dividend or other distribution (whether payable in cash, stock, property or a combination thereof) with respect to any of its capital stock (other than dividends paid by a wholly-owned Subsidiary of the Company to the Company or to any other wholly-owned Subsidiary of the Company) or enter into any agreement with respect to the voting of its capital stock;
- (d) reclassify, combine, split, subdivide or redeem, purchase or otherwise acquire, directly or indirectly, any of its capital stock, other equity interests or other securities;
- (e) other than pursuant to the Eureka Budget, (i) acquire (including, without limitation, by merger, consolidation, or acquisition of stock or assets) any interest in any person or any division thereof or any assets, other than acquisitions of assets in the ordinary course of business consistent with past practice and any other acquisitions for consideration that is individually not in excess of Two Hundred and Fifty Thousand Dollars (\$250,000), or in the aggregate, not in excess of Two Hundred and Fifty Thousand Dollars (\$250,000) for the Company and the Company Subsidiaries taken as a whole, (ii) incur any indebtedness for borrowed money or issue any debt securities or assume, guarantee or endorse, or otherwise as an accommodation become responsible for, the obligations of any person (other than a wholly-owned Subsidiary of the Company) for borrowed money in a principal amount, in the aggregate, in excess of Two Hundred and Fifty Thousand Dollars (\$250,000) for the Company and its Subsidiaries taken as a whole, (iii) terminate, cancel or request any material change in, or agree to any material change in, any Material Contract other than in the ordinary course of business consistent with past practice, (iv) make or authorize any capital expenditures that are, in the aggregate, in excess of Two Hundred and Fifty Thousand Dollars (\$250,000) for the Company and its Subsidiaries taken as a whole, or (v) enter into or amend any contract, agreement, commitment or arrangement that, if fully performed, would not be permitted under this *Section 5.2(e)*;
- (f) except as required by contractual commitments or corporate policies with respect to severance or termination pay in existence on the date of this Agreement and disclosed in the SEC

Table of Contents

Reports or Section 5.2(f) of the Disclosure Schedule: (i) materially increase the compensation or benefits payable or to become payable to its directors, officers or employees (except for increases in accordance with past practices in salaries or wages of employees of the Company or any of its Subsidiaries which are not across-the-board increases), (ii) grant any rights to severance or termination pay to, or enter into any employment or severance agreement with, any director, officer or other employee of the Company or any of its Subsidiaries, or establish, adopt, enter into or amend any collective bargaining, bonus, profit sharing, thrift, compensation, stock option, restricted stock, pension, retirement, deferred compensation, employment, termination, severance or other plan, agreement, trust, fund, policy or arrangement for the benefit of any director, officer or employee, except to the extent required by applicable Law or the terms of a collective bargaining agreement in existence on the date of this Agreement, or (iii) take any affirmative action to amend or waive any performance or vesting criteria or accelerate vesting, exercisability or funding under any Employee Benefit Plan;

- (g) (i) pre-pay any long-term debt, except in the ordinary course of business in an amount not to exceed Two Hundred and Fifty Thousand Dollars (\$250,000) in the aggregate for the Company and its Subsidiaries taken as a whole, or pay, discharge or satisfy any claims, liabilities or obligations (absolute, accrued, contingent or otherwise), except in the ordinary course of business consistent with past practice and in accordance with their terms, (ii) accelerate or delay collection of notes or accounts receivable in advance of or beyond their regular due dates or the dates when the same would have been collected in the ordinary course of business consistent with past practice, (iii) accelerate payment of any account payable in advance of its due date other than in the ordinary course of business consistent with past practice, or (iv) vary the Company's inventory practices in any material respect from the Company's past practices;
- (h) make any change in accounting policies or procedures, other than in the ordinary course of business consistent with past practice or except as required by GAAP or by a Governmental Authority;
 - (i) waive, release, assign, settle or compromise any material claims, or any material litigation or arbitration;
 - (j) make any material tax election or settle or compromise any material liability for Taxes;
- (k) modify, amend or terminate, or waive, release or assign any material rights or claims with respect to any confidentiality or standstill agreement to which the Company is a party;
- (l) write up, write down or write off the book value of any assets, individually or in the aggregate, for the Company and its Subsidiaries taken as a whole, in excess of Two Hundred and Fifty Thousand Dollars (\$250,000), except for depreciation and amortization in accordance with GAAP consistently applied;
- (m) except as permitted by, and in accordance with, *Sections 6.1* or *6.8*, take any action to exempt or make not subject to (i) the provisions of Section 203 of the DGCL, or (ii) any other state takeover law or state law that purports to limit or restrict business combinations or the ability to acquire or vote shares; or
 - (n) authorize or enter into any agreement or otherwise make any commitment to do any of the foregoing.

ARTICLE VI ADDITIONAL AGREEMENTS

6.1 Agreement to Call Stockholder Meeting. The Company shall call a meeting of its stockholders (the "Stockholders Meeting") to be held as promptly as practicable for the purpose of considering and voting upon (a) the issuance of the Offered Securities, (b) the amendment of the Company's Certificate of Incorporation to increase the number of authorized shares of Common Stock up to 1,000,000,000, (c) provide for, at the election of the Board, a reclassification of Common Stock to effect a reverse stock split, and (d) each other matter required to be approved by such stockholders in connection with the this Agreement, the other Transaction Documents and transactions contemplated hereby and thereby (collectively, the "Stockholder Approval").

The Company will, through its Board, subject to its fiduciary obligations, recommend that its stockholders approve the Stockholder Approval. The Company shall use reasonable best efforts to solicit proxies in favor of the Stockholder Approval and otherwise to secure the required vote of its stockholders; *provided*, *however*, the Company may, through its Board, withdraw, change, amend, modify or qualify its recommendation that its stockholders approve the Stockholder Approval (a "*Change of Recommendation*") if the Company has complied with the provisions of *Section 6.7* and *6.8* and the applicable provisions of *Section 8.3*.

6.2 Proxy Statement; Other Commission Filings.

- (a) As soon as reasonably practicable after the execution of this Agreement, but in any event within thirty (30) days following the Signing Date, the Company shall file with the SEC a preliminary proxy statement (the "Proxy Statement") for the Stockholders Meeting which may be combined with the Company's annual meeting of stockholders. The Proxy Statement shall be in form and substance reasonably satisfactory to the Parties. The Company shall respond promptly to any comments of the SEC and shall use reasonable best efforts to cause the Proxy Statement to be cleared by the SEC as promptly as practicable after such filing after which the Company shall promptly mail the definitive Proxy Statement to its stockholders. The Company will notify Purchaser promptly of the receipt of any comments from the SEC or its staff or any other government officials and of any request by the SEC or its staff or such other government officials for amendments or supplements to the Proxy Statement or any filing incorporated therein or for additional information, and will supply Purchaser with copies of all correspondence between it and any of its representatives, on the one hand, and the SEC or its staff or any other government officials on the other hand, with respect to the Proxy Statement and the transactions contemplated by this Agreement. The Company shall provide Purchaser with a reasonable opportunity to review and comment on drafts of the Proxy Statement (including each amendment or supplement thereto) and all responses to requests, if any, for additional information by and replies to comments of the SEC, prior to filing such with, or sending such to, the SEC. Whenever any Party becomes aware of any event that is required to be set forth in an amendment or supplement to the Proxy Statement or any other filing with the SEC in connection with this Agreement, or the transactions contemplated hereby or thereby, such party shall promptly inform the other parties of such occurrence. The Company shall promptly prepare and, with Purchaser's prior approval (which shall not be unreasonably withheld or delayed), file with the SEC any such amendment or supplement and, following clearance thereof, if applicable, mail such amendment or supplement to its stockholders. To the extent information regarding Purchaser is required for the preparation of the Proxy Statement, Purchaser shall promptly provide such information to the Company upon request.
- (b) Until consummation of the transactions contemplated by this Agreement or earlier termination of this Agreement, the Company shall timely file all reports, registration statements, proxy or information statements and other documents required to be filed by it with the SEC (collectively, the "*Other Filings*"), each of which filings shall comply with all applicable

Table of Contents

requirements of the Securities Act and the Exchange Act and the rules and regulations of the SEC under each such act and other applicable Laws. The Company shall promptly notify Purchaser and its counsel of all filings with the SEC made by the Company prior to the consummation of the transactions contemplated by this Agreement or earlier termination of this Agreement.

- (c) The Company agrees that none of the information included or incorporated by reference (i) in the Proxy Statement and any amendment or supplement thereto, at the date of mailing to shareholders and at the time of the Stockholders Meeting, and (ii) in any Other Filings, at the time of filing and at any distribution or dissemination thereof, will contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading; *provided*, *however*, that the Company's covenants in clauses (i) and (ii) above will not apply to statements or omissions included in the Proxy Statement or any Other Filings based upon information furnished in writing to the Company by Purchaser specifically for use therein. The Company agrees that the Proxy Statement, and any amendments or supplements thereto, when filed by the Company with the SEC, or when distributed or otherwise disseminated to the Company's shareholders, as applicable, will comply with the applicable requirements of the Exchange Act and the rules and regulations of the SEC under such act and other applicable Laws.
- (d) The information supplied by Purchaser for the purpose of inclusion in the Proxy Statement shall not, on the date the Proxy Statement is first mailed to the Company's stockholders, and at the time of the Stockholders Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.
- **6.3** Listing Applications. The Company shall apply to list the Offered Securities for trading on the NYSE MKT and will use reasonable best efforts to cause such listing to be effective as of the Closing, subject to official notice of issuance.

6.4 Assistance in Procuring Loan; Purchaser Option.

- (a) Purchaser shall use its reasonable best efforts to assist the Company to procure from one or more Prime Chinese Banks, the Loan for the Company in accordance with the terms and conditions set forth on *Schedule 2*, including a guarantee of the Loan as required by such Prime Chinese Bank or Banks.
- (b) The Company hereby grants Purchaser an option to enter into the Molybdenum Supply Agreement with the Company simultaneously with the Loan Execution. Purchaser may exercise such option by providing the Company with written notice thereof at any time from the Signing Date until the day prior to the Loan Execution.

6.5 Consents.

- (a) Each of the Company and Purchaser shall use reasonable best efforts to take or to cause to be taken all actions, and to do, or to cause to be done, all things necessary, proper or advisable to obtain all required consents, approvals and waivers for the consummation of the transactions contemplated by this Agreement and the other Transaction Documents, including, without limitation, obtaining the consents and approvals from any third parties or Governmental Authorities (including those identified in *Schedule 3*), including giving all required notices and making all required filings with respect thereto.
 - (b) The Company shall use reasonable best efforts to obtain the APERAM Consent.

Table of Contents

6.6 Covenants Concerning the Parties.

- (a) Each Party shall be obligated to furnish prompt written notice of each of the following to the other Party: (i) the occurrence of any material breach or default by such Party under this Agreement; (ii) the filing or commencement of any action, suit or proceeding by or before and Governmental Authority against or affecting such Party that, if adversely determined, would reasonably be expected to result in any condition to the obligations of any party to effect the transactions contemplated by this Agreement not to be satisfied, or otherwise materially delay or make unlikely the Closing or the Loan Execution, (iii) any other development, review, request, requirement or proceeding that would reasonably be expected to result in any condition to the obligations of any party to effect the transactions contemplated by this Agreement not to be satisfied, or otherwise materially delay or make unlikely the Closing or the Loan Execution. Each notice delivered under this Section shall be accompanied by a statement of an officer of such setting forth the details of the event or development requiring such notice; *provided*, *however*, that that delivery of any notice under this Section 6.6(a) shall not cure any breach of any representation or warranty requiring disclosure of such matter prior to the date of this Agreement or otherwise limit or affect the remedies available hereunder to the Party receiving such notice.
- (b) Each Party shall act in good faith and use reasonable best efforts to facilitate the completion of the transactions contemplated under this Agreement and the other Transaction Documents on the terms and conditions set forth in this Agreement and the other Transaction Documents.
- (c) Purchaser acknowledges that some of the information disclosed by the Company pursuant to this Agreement will be confidential information or material non-public information of the Company. Purchaser shall keep all such information confidential in accordance with the provisions of the NDA Agreement.
 - (d) Each Party shall materially comply with all Laws, including the FCPA and other applicable anti-corruption Laws.

6.7 No Solicitation.

(a) None of the Company or any of its Subsidiaries shall, directly or indirectly, take (and the Company shall not authorize or permit any of its officers, directors, employees, accountants, consultants, legal counsel, advisors, agents and other representatives or, to the extent within the Company's control, other Affiliates to take) any action to (i) encourage (including by way of furnishing non-public information), solicit, initiate or facilitate any Alternative Proposal, (ii) enter into any agreement with respect to any Alternative Proposal or enter into any agreement, arrangement or understanding requiring it to abandon, terminate or fail to consummate the sale of the Offered Securities or any other transaction contemplated by this Agreement, or (iii) participate in any discussions or negotiations with, or furnish any information to, any person in connection with, or take any other action to facilitate any inquiries or the making of any proposal that constitutes, or could reasonably be expected to lead to, any Alternative Proposal; provided, however, that if, at any time prior to obtaining the Stockholder Approval, the Board determines in good faith, after consultation with outside counsel, that it would otherwise constitute a breach of the directors' fiduciary duties to stockholders, the Company may, in response to a Superior Proposal and subject to the Company's compliance with Sections 6.1, 6.7 and 6.8, (x) furnish information with respect to the Company to the person making such Superior Proposal pursuant to a customary confidentiality agreement, the benefits of the terms of which are no more favorable to the other party to such confidentiality agreement than those in place with Purchaser, and (y) participate in discussions with the person making such Superior Proposal (including discussions with such person upon receipt of the Alternative Proposal that may be deemed necessary by the Board to determine whether such Alternative Proposal constitutes a Superior Proposal). Upon

Table of Contents

execution of this Agreement, the Company shall cease immediately and cause to be terminated any and all existing discussions or negotiations with any parties conducted heretofore with respect to an Alternative Proposal and promptly request that all confidential information with respect thereto furnished on behalf of the Company be returned.

- (b) The Company shall, as promptly as practicable (and in no event later than thirty-six (36) hours after receipt thereof), advise Purchaser of any inquiry received by it relating to any potential Alternative Proposal and of the material terms of any proposal or inquiry, including the identity of the person and its affiliates making the same, that it may receive in respect of any such potential Alternative Proposal, or of any information requested from it or of any negotiations or discussions being sought to be initiated with it, shall furnish to Purchaser a copy of any such proposal or inquiry, if it is in writing, or a written summary of any such proposal or inquiry, if it is not in writing, and shall keep Purchaser fully informed on a prompt basis with respect to any developments with respect to the foregoing.
- (c) Except in accordance with *Sections 6.1*, *6.7* and *6.8*, neither the Board nor any committee thereof shall (i) undertake a Change of Recommendation, or (ii) cause the Company to enter into any letter of intent, agreement in principle, acquisition agreement or other similar agreement related to any Alternative Proposal.

6.8 Change of Recommendation.

- (a) Notwithstanding anything in *Sections 6.1* and *6.7* to the contrary, at any time prior to the receipt of the Stockholder Approval, the Board may make a Change of Recommendation following receipt of an unsolicited bona fide written proposal or offer (an "*Offer*"), by a Person or group (as defined in Section 13(d) of the Securities Exchange Act), including any amendment or modification to any existing Offer, with respect to an acquisition of beneficial ownership by such Person or group of (i) at least twenty-five percent (25%) of the assets of, equity interests in, or businesses of, the Company (whether pursuant to a single or multi-step transaction or series of related transactions), or (ii) a merger, consolidation, recapitalization or other transaction that results in the issuance, disposition or sale of twenty-five percent (25%) or more of the voting power of the Company (an "*Alternative Proposal*"), which the Board determines, in the exercise of its fiduciary duties, is a Superior Proposal, in each case, if the Board has determined in good faith after consultation with the Company's outside legal counsel that the failure to take such action would constitute a breach of the fiduciary duties of the members of the Board under applicable Delaware Law and the Company first complies with *Section 6.8(b)*.
- (b) Prior to the Company taking any action permitted under *Section 6.8(a)*, the Company shall provide Purchaser with seven (7) Business Days' prior written notice (it being understood and agreed that any material amendment to the applicable Alternative Proposal shall require a new notice and an additional five (5) Business Day period) advising Purchaser that the Board intends to take such action and providing to Purchaser a notice of such Offer (containing the principal terms and conditions of the Offer), and during such seven (7) Business Day period (or subsequent five (5) business day period), (i) the Company shall negotiate, and cause its representatives to negotiate, with Purchaser and its representatives in good faith (to the extent Purchaser wishes to negotiate) to enable Purchaser to determine whether to propose revisions to the terms of this Agreement or any other Transaction Document such that such Alternative Proposal would no longer constitute a Superior Proposal, and (ii) the Company shall consider in good faith any proposal by Purchaser to amend the terms and conditions of this Agreement or any other Transaction Document such that such Alternative Proposal would no longer constitute a Company Superior Proposal.
- (c) Nothing contained in this Agreement shall prohibit the Company or the Board from (i) disclosing to the Company's stockholders a position contemplated by Rules 14d-9 and 14e-2(a)

Table of Contents

promulgated under the Exchange Act, or (ii) making any disclosure to its stockholders if the Board has reasonably determined in good faith after consultation with the Company's outside legal counsel that the failure to do so would constitute a breach of the fiduciary duties of the members of the Board under applicable Delaware Law; *provided* that this *Section* 6.8(c) shall not permit the Board to make a Change of Recommendation except in compliance with *Section* 6.8(a) and *Section* 6.8(b).

- (d) All information provided to Purchaser under this *Section 6.8* shall be held in confidence in accordance with the terms of the NDA Agreement (whether or not then in effect).
- **6.9** *Certain Tax Covenants*. Notwithstanding any provision in this Agreement to the contrary, the Company will deduct and withhold any and all amounts required to be withheld and paid to any taxing authority in respect of any payments to be made to Purchaser (including any amounts paid by the Company in respect of the Arrangement Fee or the Company Break Fee). In no event will the Company pay any "additional amounts" or "gross-up" payments to any party under this Agreement in order to compensate such party for any reduction in the net after-tax proceeds it receives as a result of any amounts deducted and withheld by the Company.
- **6.10** Access to Information. Except as required pursuant to any confidentiality agreement or similar agreement or arrangement to which the Company or any of its Subsidiaries is a party (which such person shall use its reasonable best efforts to cause the counterparty to waive), from the Signing Date to the Closing Date, the Company shall, and shall cause each of its Subsidiaries and representatives to: (a) provide to Purchaser and its representatives access at reasonable times upon prior notice to the officers, employees, agents, properties, offices and other facilities of such party and its subsidiaries and to the books and records thereof, and (b) furnish promptly such information concerning the business, properties, contracts, assets, liabilities, personnel and other aspects of such party and its subsidiaries as Purchaser or its representatives may reasonably request. No investigation conducted pursuant to this Section 6.9 shall affect or be deemed to modify or limit any representation or warranty made in this Agreement or any other Transaction Document.
- **6.11** Further Assurances. In case at any time after the Closing any further action is necessary or desirable to carry out the purposes of this Agreement, the Company and Purchaser will take such further action as the other party may reasonably request, all at the sole cost and expense of the requesting Party.
- 6.12 *Publicity*. Upon execution of this Agreement, the Parties shall issue a mutually agreed press release concerning this Agreement and the transactions contemplated hereby. Other than such joint press release, no Party shall, nor shall such Party permit its Affiliates to, issue any press release or public announcement concerning this Agreement, the other Transaction Documents or the transactions contemplated hereby or thereby without obtaining the prior written approval of the other Party, which approval will not be unreasonably withheld or delayed. Notwithstanding the foregoing, each of the Parties may issue such press release or public announcement and make such filings with any Governmental Authority (including the SEC) if, in the reasonable judgment of counsel to such Party, such disclosure or filing is otherwise required by applicable laws or by the applicable rules of any stock exchange on which such Party lists its securities; *provided* that the disclosing Party shall use its reasonable best efforts to consult with the other Party with respect to the text thereof if possible under applicable laws and by the applicable rules of such stock exchanges.
- **6.13** *Board Representation.* The Company shall provide, as of the Closing, the resignation and releases of the directors designated by the Company in conjunction with the appointment of the Purchaser Nominees. The Company agrees to use reasonable best efforts to cause the Board, at the Closing, to be comprised of eight (8) individuals, of whom two (2) individuals shall be nominated by Purchaser consistent with the terms of the Stockholder Agreement (the "*Purchaser Nominees*"), subject to requirements of applicable Law and under the rules of the NYSE MKT.

ARTICLE VII CONDITIONS

7.1 Closing.

- (a) Conditions to the Obligations of Each Party. The obligations of each Party to effect the transactions relating to the Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following conditions:
 - (i) *No Order*. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any law (whether temporary, preliminary or permanent) which is then in effect and has the effect of making the transactions contemplated by this Agreement or the other Transaction Documents illegal or otherwise restricting, preventing or prohibiting consummation of the transactions contemplated by this Agreement or the other Transaction Documents.
 - (ii) Stockholder Approval. The Company shall have received the Stockholder Approval.
 - (iii) Required Approvals. Purchaser shall have received the approvals set forth in Schedule 3.
- (b) Conditions to the Obligations of the Company. The obligations of the Company to effect the transactions relating to the Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Representations and Warranties. Each of the representations and warranties of Purchaser contained in this Agreement that are qualified by materiality or Material Adverse Effect shall be true and correct as of the Signing Date and as of the Closing as though made on and as of the Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date), and each of the representations and warranties which is not so qualified shall be true and correct in all material respects as of the Signing Date and as of the Closing as though made on and as of the Closing (except that those representations and warranties which address matters only as of a particular date need only remain true and correct in all material respects as of such date).
 - (ii) Agreements and Covenants. Purchaser shall have performed in all material respects, all obligations and complied with, in all material respects, its agreements and covenants to be performed or complied with by it under this Agreement on or prior to the Closing.
 - (iii) Closing Deliveries. Purchaser shall have made all of the deliveries contemplated by Section 1.4(b).
 - (iv) Officer's Certificate. Purchaser shall have delivered to the Company a certificate, dated the Closing Date, signed by an officer of Purchaser, certifying as to the satisfaction of the conditions specified in Sections 7.1(b)(i) and (ii).
- (c) Conditions to the Obligations of Purchaser. The obligations of Purchaser to effect the transactions relating to the Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Representations and Warranties. Each of the representations and warranties of the Company contained in this Agreement that are qualified by materiality or Material Adverse Effect shall be true and correct as of the Signing Date and as of the Closing as though made on and as of the Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date) and each of the representations and warranties which is not so qualified shall be true and correct in all material respects as of the Signing Date and as of the Closing as though made on and as of

Table of Contents

the Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date); *provided*, *however*, that with respect to *Sections 2.4(b)* and *2.17*, all references to the Toronto Stock Exchange shall not apply if the Company has delisted, or is in the process of delisting from, the Toronto Stock Exchange.

- (ii) Agreements and Covenants. Each of the Company and each of its Subsidiaries shall have performed, in all material respects, all obligations and complied with, in all material respects, its agreements and covenants to be performed or complied with by it under this Agreement on or prior to the Closing.
- (iii) Stockholder Approval. Purchaser shall have received a true and complete copy of the resolutions of the stockholders of the Company adopting the Stockholder Approval, certified by the Secretary or an Assistant Secretary of the Company.
 - (iv) APERAM Consent. The Company shall have obtained the APERAM Consent.
- (v) NYSE MKT Listing. The Company shall have filed an application for the listing of the Offered Securities with the NYSE MKT and shall have received notification from NYSE MKT that the Offered Securities have been approved for listing, subject to official notice of issuance.
 - (vi) Closing Deliveries. The Company shall have made all of the deliveries contemplated by Section 1.4(a).
- (vii) Officer's Certificate. The Company shall have delivered to Purchaser a certificate, dated the Closing Date, signed by an officer of the Company, certifying as to the satisfaction of the conditions specified in Sections 7.1(c)(i), (ii) and (viii).
- (viii) No Material Adverse Effect. Since the date of this Agreement, there shall not have occurred any Material Adverse Effect, or any change, event, condition, state of facts or development that may, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (ix) Resignation of Directors; Re-Constitution of the Board. The directors of the Company listed on Section 6.13 of the Disclosure Schedule shall have submitted their resignations in writing to the Company with copies to Purchaser. Such resignations shall be effective as of the Closing. The Purchaser Nominees shall have been appointed or elected as two (2) of the eight (8) members of the Board effective as of the Closing Date.

ARTICLE VIII TERMINATION

- **8.1** *Termination*. This Agreement may be terminated and the transactions contemplated by this Agreement may be abandoned at any time prior to the Closing:
 - (a) By mutual written consent of Purchaser and the Company; or
 - (b) By either Purchaser or the Company if:
 - (i) the Closing shall not have occurred on or before December 31, 2015;
 - (ii) any Governmental Authority shall have enacted, issued, promulgated, enforced or entered any order, decree, judgment, injunction or ruling or have issued a denial of approval which is then in effect and is final and nonappealable and has the effect of making consummation of the transactions contemplated by this Agreement or the other Transaction

Table of Contents

Documents illegal or otherwise preventing or prohibiting consummation of the transactions contemplated by this Agreement or the other Transaction Documents; or

(iii) the Stockholder Approval shall not have been obtained at a duly held meeting of stockholders or at any adjournment thereof.

(c) By the Company if:

- (i) (A) any of the representations or warranties of Purchaser that is qualified as to materiality or material adverse effect shall have become untrue, or any of the representations or warranties of Purchaser that is not so qualified shall have become untrue in any material respect, or Purchaser shall have breached or failed to perform or comply in any material respect with any of its covenants or agreements in this Agreement, and (B) any such misrepresentation or breach cannot be cured or has not been cured within twenty (20) days after the giving of written notice by the Company to Purchaser specifying such breach;
- (ii) (A) Purchaser makes a general assignment for the benefit of its creditors, (B) Purchaser commences a voluntary case or proceeding under any applicable bankruptcy, insolvency or reorganization law seeking to be adjudicated a bankrupt or insolvent, (C) Purchaser consents to the entry of an order for relief in respect of Purchaser in an involuntary case or proceeding under any applicable bankruptcy, insolvency or reorganization law, or (D) a court of competent jurisdiction enters a decision, decree or other order, which decision, decree or other order remains unstayed and in effect for sixty (60) days, under any law relating to bankruptcy, insolvency or reorganization that is for relief against Purchaser in an involuntary case, that appoints a custodian of Purchaser or for a substantial part of its property or that orders the winding up or liquidation of Purchaser; or
- (iii) if, prior to obtaining the Stockholder Approval, (A) the Company is not in material breach of any of the terms of this Agreement, and (B) if permitted by, and in compliance with, *Sections 6.1, 6.7* and *6.8*, the Board authorizes the Company to enter into a binding written agreement concerning a transaction that constitutes a Superior Proposal.

(d) By Purchaser if:

- (i) (A) any of the representations or warranties of the Company that is qualified as to materiality or Material Adverse Effect shall have become untrue, or any of the representations or warranties of the Company that is not so qualified shall have become untrue in any material respect, or the Company shall have breached or failed to perform or comply in any material respect with any of its covenants or agreements in this Agreement, and (B) any such misrepresentation or breach cannot be cured or has not been cured within twenty (20) days after the giving of written notice by Purchaser to the Company specifying such breach;
- (ii) there shall have occurred any Material Adverse Effect, or any change, event, condition, state of facts or development that may, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect and such Material Adverse Effect is not cured within twenty (20) days after written notice thereof;
- (iii) (A) The Company, Eureka Moly or Nevada Moly makes a general assignment for the benefit of its creditors, (B) the Company, Eureka Moly or Nevada Moly commences a voluntary case or proceeding under any applicable bankruptcy, insolvency or reorganization law seeking to be adjudicated a bankrupt or insolvent, (C) the Company, Eureka Moly or Nevada Moly consents to the entry of an order for relief in respect of the Company, Eureka Moly or Nevada Moly in an involuntary case or proceeding under any applicable bankruptcy, insolvency or reorganization law, or (D) a court of competent jurisdiction enters a decision, decree or other order, which decision, decree or other order remains unstayed and in effect

Table of Contents

for sixty (60) days, under any law relating to bankruptcy, insolvency or reorganization that is for relief against the Company or Eureka Moly or Nevada Moly in an involuntary case, that appoints a custodian of the Company or Eureka Moly or Nevada Moly or for a substantial part of its property or that orders the winding up or liquidation of the Company or Eureka Moly or Nevada Moly; or

- (iv) the Board shall have effected a Change of Recommendation in accordance with Sections 6.1 and 6.8.
- **8.2** Effect of Termination. In the event of the termination of this Agreement pursuant to Section 8.1, this Agreement shall forthwith become void, and there shall be no liability or obligation on the part of any Party, except with respect to (i) Section 6.12, this Section 8.2 and Article IX and (ii) any liabilities or damages incurred or suffered by a party (subject to the provisions of Schedule 4) as a result of the material breach by the other party of any of its representations, warranties, covenants or other agreements set forth in this Agreement or other Transaction Documents; provided that, in circumstances where the Company Break Fee is payable under Section 8.3 such fee shall be the sole remedy with respect to specified basis for termination, and in no event shall the Company be obligated to pay the Company Break Fee on more than one (1) occasion.

8.3 Payment of Company Break Fee.

- (a) If this Agreement is terminated pursuant to Section 8.1(c)(iii) or Section 8.1(d)(iv), the Company shall pay Purchaser the Company Break Fee upon such termination, in the case of termination by the Company, or within two (2) Business Days after such termination, in the case of termination by Purchaser.
- (b) If this Agreement is terminated pursuant to *Section 8.1(b)(iii)*, the Company shall pay Purchaser the Reimbursed Purchaser Expenses upon such termination; *provided*, *however*, that if within twelve (12) months of such termination, the Company enters into or consummates a definitive agreement with respect to an Alternative Proposal, the Company shall pay Purchaser within two (2) Business Days following the execution or consummation of such definitive agreement an amount equal to (i) Company Break Fee, *minus* (ii) the Reimbursed Purchaser Expenses.
- (c) If this Agreement is terminated pursuant to $Section \ 8.1(b)(i)$ or $Section \ 8.1(d)(i)$ and (i) an Alternative Proposal shall have been made to the Company or its stockholders, or any Person shall have publicly announced an intention to make an Alternative Proposal with respect to the Company prior to such termination, and (ii) within twelve (12) months of such termination, the Company enters into or consummates a definitive agreement with respect to an Alternative Proposal, the Company shall pay Purchaser the Company Break Fee within two (2) Business Days following the execution or consummation of such definitive agreement.

ARTICLE IX MISCELLANEOUS

9.1 *Notices*. All notices, requests, consents and other communications hereunder shall be in writing, shall be addressed to the receiving party's address set forth below or to such other address as a party may designate by notice under this *Section 9.1*, and shall be either (a) delivered by hand,

Table of Contents

(b) made by telecopy or facsimile transmission, or (c) sent by Federal Express, DHL, UPS or another internationally recognized delivery service.

If to Purchaser: Amer International Group Co., Ltd

29/F, Block A, East Pacific International Center

7888th Shennan Boulevard Shenzhen 518040, China Attention: Tong Zhang Facsimile: +86.755.2711.8899

With copies to: ZHONG LUN LAW FIRM

36-37/F,SK Tower, 6A Jianguomenwai Avenue Chaoyang District, Beijing 100022, P.R.China

Attention: Jun CHENG Facsimile: +86 10 6568 1838

Latham & Watkins

18th Floor, One Exchange Square 8 Connaught Place, Central

Hong Kong

Attention: David M. Blumental

Allen C. Wang

Facsimile: +852.2912.2600

If to the Company: General Moly, Inc.

1726 Cole Blvd.

Suite 115

Lakewood, CO 80401

U.S.A.

Attention: Chief Executive Officer Facsimile: +1 (303) 928-8598

With a copy to: Bryan Cave LLP

1700 Lincoln Street

Suite 4100

Denver, CO 80203-4541

U.S.A.

Attention: Charles D. Maguire, Jr. Facsimile: +1 (303) 866-0200

All notices, requests, consents and other communications hereunder shall be deemed to have been given and received (i) if by hand, at the time of the delivery thereof to the receiving party at the address of such party set forth above, (ii) if by telecopy or facsimile transmission, on the day that receipt thereof has been acknowledged by electronic confirmation or otherwise, or (iii) if sent by internationally recognized delivery service, on the day of actual receipt.

9.2 Entire Agreement. This Agreement and the other Transaction Documents, including exhibits or other documents referred to herein and therein, embody the entire agreement and understanding between the Parties with respect to the subject matter hereof and supersedes all prior oral or written agreements and understandings relating to the subject matter hereof. No statement, representation, warranty, covenant or agreement not expressly set forth in this Agreement shall affect, or be used to interpret, change or restrict the express terms and provisions of this Agreement.

Table of Contents

- **9.3** *Amendments*. The terms and provisions of the Agreement may be modified, amended or waived, or consent for the departure from such terms and provisions may be granted, only by written consent of the Company and Purchaser. Each such waiver or consent shall be effective only in the specific instance and for the purpose for which it was given, and shall not constitute a continuing waiver or consent.
- **9.4** Assignment. Neither Party may assign its rights or delegate its obligations under this Agreement without the express prior written consent of the other Party, *provided*, however, that Purchaser may assign any of its rights and obligations hereunder, in whole or in part, to any Affiliate of Purchaser without obtaining the consent of the Company and any such assignment shall not relieve Purchaser of its obligations hereunder.
- **9.5** *Benefit.* All statements, representations, warranties, covenants and agreements in this Agreement shall be binding on the Parties and shall inure to the benefit of the respective successors and permitted assigns of each Party. Nothing in this Agreement shall be construed to create any rights or obligations except between the Parties, and no person or entity shall be regarded as a third party beneficiary of this Agreement subject to *Sections 4.3(d)* and *(e)*.
- **9.6** *Specific Performance*. The Parties agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with the terms hereof and that the Parties shall be entitled to specific performance of the terms hereof, in addition to any other remedy at law or in equity.
- **9.7** *Governing Law; Language*. This Agreement and the rights and obligations of the Parties hereunder shall be governed by, and construed in accordance with, the laws of the State of Delaware, without giving effect to the conflict of law principles thereof that would cause the application of the laws of any jurisdiction other than the State of Delaware. This Agreement has been negotiated and executed by the Parties in English. In the event any translation of this Agreement is prepared for convenience or any other purpose, the provisions of the English version shall govern.
- **9.8** *Waiver of Jury Trial.* Each of the Parties hereby waives to the fullest extent permitted by applicable law any right it may have to a trial by jury with respect to any litigation directly or indirectly arising out of, under or in connection with this Agreement or the other Transaction Documents or the transactions contemplated hereby or thereby.
- **9.9** Severability. In the event that any court of competent jurisdiction shall determine that any provision, or any portion thereof, contained in this Agreement shall be unenforceable in any respect, then such provision shall be deemed limited to the extent that such court deems it enforceable, and as so limited shall remain in full force and effect. In the event that such court shall deem any such provision, or portion thereof, wholly unenforceable, the remaining provisions of this Agreement shall nevertheless remain in full force and effect.
- **9.10** *Headings and Captions*. The headings and captions of the various subdivisions of this Agreement are for convenience of reference only and shall in no way modify or affect the meaning or constructions of any of the terms or provisions hereof.

9.11 Certain Terms.

(a) Unless the context of this Agreement otherwise clearly requires, (i) references to the plural include the singular, and references to the singular include the plural, (ii) references to one gender include the other gender, (iii) the words "include," "includes" and "including" do not limit the preceding terms or words and shall be deemed to be followed by the words "without limitation," (iv) the terms "hereof," "herein," "hereunder," "hereto" and similar terms in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement, (v) references to a day, without the explicit qualification of "business" refers to a

Table of Contents

calendar day, (vi) references to a month, quarter, year or such other subdivision, without the explicit qualification of "fiscal", refers to a calendar month, quarter, year or other such subdivision, respectively, (vii) all references to "the date hereof," "the date of this Agreement" or similar terms (but excluding references to the date of execution hereof) refer to the Signing Date, notwithstanding that the Parties may have executed this Agreement on a later date, and (viii) references to any Person include such Person's respective successors, assigns, transferees, lessees, heirs, executors and administrators, whether by merger, consolidation, amalgamation, reorganization, sale of assets or otherwise.

- (b) Unless otherwise set forth herein, references in this Agreement to (i) any document, instrument or agreement (including this Agreement) include and incorporate all exhibits, schedules and other attachments thereto, as amended, modified or supplemented, and (ii) a particular Law referenced herein means such Law as amended, modified, supplemented or succeeded. When a reference is made in this Agreement to Articles, Sections or any other subdivision, such reference is to an Article, a Section or other subdivision of this Agreement, unless otherwise indicated. When a reference is made in this Agreement to a party or parties, such reference is to parties to this Agreement, unless otherwise indicated. Unless otherwise specified, all references to "\$" shall be deemed to be references to the lawful currency of the United States.
- (c) In this Agreement, any reference to the Company's knowledge, and comparable terms including "know," "known," "aware" or "awareness," of a particular fact or other matter means the actual knowledge of the officers of the Company or what they could reasonably be expected to have known in performing their duties in the offices in which they serve or had such individuals conducted a reasonable inquiry under the applicable circumstances. The executive officers of the Company as of the Signing Date are Bruce D. Hansen, Chief Executive Officer, David A. Chaput, Chief Financial Officer, Michael K. Branstetter, Secretary, R. Scott Roswell, Corporate Counsel VP Human Resources, Lee M. Shumway, Treasurer, and Robert I. Pennington, Chief Operating Officer.
- **9.12** *No Waiver of Rights, Powers and Remedies.* No failure or delay by a Party in exercising any right, power or remedy under this Agreement, and no course of dealing between the Parties, shall operate as a waiver of any such right, power or remedy of the party. No single or partial exercise of any right, power or remedy under this Agreement by a Party, nor any abandonment or discontinuance of steps to enforce any such right, power or remedy, shall preclude such party from other or further exercise thereof or the exercise of any other right, power or remedy hereunder. The election of any remedy by a Party shall not constitute a waiver of the right of such party to pursue other available remedies. No notice to or demand on a Party not expressly required under this Agreement shall entitle the Party receiving such notice or demand to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the Party giving such notice or demand to any other or further action in any circumstances without such notice or demand.
- **9.13** *Fees and Expenses.* Except as otherwise set forth in this Agreement or the other Transaction Documents, each of the Parties shall pay its own fees and expenses (including the fees of any attorneys, accountants, appraisers or others engaged by such Party) in connection with this Agreement and the transactions contemplated hereby whether or not the transactions contemplated hereby are consummated.
- **9.14** *Counterparts*. This Agreement may be executed in counterparts (including by facsimile, "PDF" or similar means of electronic communication), each of which shall be deemed an original and all of which together shall constitute one agreement.
- **9.15** *Rules of Construction.* The Parties agree that they have been represented by counsel during the negotiation, preparation and execution of this Agreement and, therefore, waive the application of

Table of Contents

any Law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the party drafting such agreement or document.

9.16 *Dispute Resolution*. All disputes between the Parties arising out of, relating to or in connection with this Agreement (a "*Dispute*") and not otherwise settled by agreement between the Parties shall be exclusively and finally settled in accordance with *Schedule 4*; *provided*, *however*, that any dispute arising out of the subject matter covered by *Sections 6.1*, 6.7 and 6.8 (including, without limitation, the exercise by the Board of its fiduciary duties and the Board effecting a Change of Recommendation) and *Section 8.3* shall be brought and determined exclusively in the Court of Chancery of the State of Delaware or, if under applicable Law exclusive jurisdiction over such matter is vested in the federal courts, any court of the United States located in the State of Delaware, and each of the Parties hereby irrevocably submits with regard to any such action or proceeding for itself and in respect to its property, generally and unconditionally, to the exclusive jurisdiction of the aforesaid courts and agrees that it will not bring any legal action or proceeding with respect to the provisions of this Agreement set forth above or for recognition and enforcement of any judgment in respect thereof in any court other than the aforesaid courts.

[Signature page follows]

Table of Contents

The Company and Purchaser have executed this Investment and Securities Purchase Agreement as of the Signing Date.

GENERAL MOLY, INC.

By: /s/ BRUCE D. HANSEN

Name: Bruce D. Hansen
Title: Chief Executive Officer

AMER INTERNATIONAL GROUP CO., LTD.

By: /s/ WENYIN WANG

Name: Wenyin Wang Title: *Chairman*

[SIGNATURE PAGE OF INVESTMENT AND SECURITIES PURCHASE AGREEMENT]

Table of Contents

Schedule 1 (to Investment and Securities Purchase Agreement)

Certain Definitions

For purposes of the Agreement (including this Schedule 1) the following terms and variations thereof have the meanings specified or referred to in this *Schedule 1*:

"10-K": As defined in Section 2.6(a).

"1933 ACT": As defined in Section 3.2(f).

"Action": Any suit, claim, complaint, charge, investigation, audit or examination, citation, legal proceeding, administrative enforcement proceeding, arbitration proceeding or subpoena of any kind or nature whatsoever, in each case by or before any Governmental Authority.

"Affiliate": As defined in Rule 405 under the Securities Act, provided none of the Company and its Subsidiaries shall be deemed an affiliate of Purchaser and neither Purchaser, POS-Minerals, nor any of its respective Affiliates shall be deemed to be an affiliate of the Company or any of its Subsidiaries.

"Agreement": As defined in the Preamble.

"Alternative Proposal": As defined in Section 6.8(a).

"APERAM": APERAM, an entity incorporated in the Grand Duchy of Luxembourg.

"APERAM Securities Agreement" the Securities Purchase Agreement dated November 19, 2007, between the Company and ArcelorMittal S.A., which has been assigned to APERAM.

"APERAM Consent": An executed letter or other document of APERAM setting forth (a) APERAM's intention with respect to the exercise of its rights under Section 6.2(a) of the APERAM Securities Agreement and (b) APERAM's confirmation that (i) it either (A) waives its rights under Section 6.3(a) of the APERAM Securities Agreement or (B) agrees that such rights are not applicable to the transactions contemplated by this Agreement and (ii) it either (X) approves of the transactions contemplated by this Agreement pursuant to Section 6.3(b) of the APERAM Securities Agreement or (Y) agrees that such rights are not applicable to the transactions contemplated by this Agreement.

"Arrangement Fee": As defined in Section 1.2.

"Blackout Period": As defined in Section 4.4.

"Board": The Board of Directors of the Company.

"Business Day": A day other than a Saturday, Sunday or other day on which commercial banks in New York City or Beijing, China are authorized or required by Law to close.

"Change of Recommendation" As defined in Section 6.1.

"Closing": As defined in Section 1.3.

"Closing Date": As defined in Section 1.3.

"Closing Payment": As defined in Section 1.1(b).

"Code": The Internal Revenue Code of 1986, as amended.

"Common Stock": As defined in Recital A.

"Company": As defined in the Preamble.

Table of Contents

"Company Break Fee": An amount equal to five percent (5%) of the Stock Purchase Price, payable in cash.

"Company Financial Statements": As defined in Section 2.6(b).

"Company Permits": As defined in Section 2.14.

"Contract": Any agreement, arrangement, understanding, note, mortgage, indenture, lease, deed of trust, license, plan, instrument or other contract.

"Convertible Notes": As defined in Section 2.2.

"DGCL": As defined in Section 2.20.

"Disclosure Schedule": As defined in Article II.

"Dispute": As defined in Section 9.16.

"Dispute Negotiation Notice": As defined in Schedule 4.

"Dollars": The lawful currency of the United States.

"Effective Period": As defined in Section 4.3(a).

"Employee Benefit Plan": Any "employee benefit plan" as defined in Section 3(3) of ERISA.

"Environmental Law": Any Law relating to the protection of the environment or the exposure of persons to, or remediation of, any Hazardous Materials, in each case as in effect as of the Signing Date.

"Environmental Permits": Any permits, licenses, certifications, authorizations or any other approvals issued by any Governmental Authority relating to the operation of the business of the Company and its Subsidiaries pursuant to any Environmental Law.

"Equity Securities": Common Stock and any other securities issued by the Company representing equity interests in the Company.

"ERISA": The Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate": Each entity that is treated as a single employer with the Company for purposes of Section 414(b), (c), (m) or (o) of the Code.

"Eureka Budget": The operating budget and business plan of Eureka Moly, as set forth in the Disclosure Schedule.

"Eureka Moly": Eureka Moly, LLC, a Delaware limited liability company.

"Eureka Moly LLC Agreement": The Amended and Restated Limited Liability Company Agreement of Eureka Moly, dated February 26, 2008, by and between Nevada Moly LLC and POS-Minerals Corporation, as amended.

"Exchange Act": As defined in Section 2.17.

"Expense Reimbursement Account": As defined in Section 1.1(c).

"Expense Reimbursement Agreement": As defined in Section 1.1(c).

"FCPA": As defined in Section 2.5(b).

"GAAP": Generally accepted accounting principles as applied in the United States.

"Governmental Authority": As defined in Section 2.4(a).

Table of Contents

"Government Official": (a) Any official, officer, employee, representative or any person acting in an official capacity for or on behalf of any Governmental Authority, (b) any political party or party official or candidate for political office, (c) any public international organization or any department or agency thereof, or (d) any Person or other entity owned in whole or in part, or controlled by any Person described in the foregoing clauses (a), (b) or (c) of this definition.

"Hazardous Material": "Hazardous substances," as defined by the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq.; "hazardous wastes," as defined by the Resource Conservation and Recovery Act, 42 U.S.C. § 6901 et seq.; petroleum or petroleum products; radioactive material, including, without limitation, any source, special nuclear, or by-product material, as defined in 42 U.S.C. §2011 et seq.; asbestos in any form or condition; toxic mold; polychlorinated biphenyls; and any other material, chemical, substance or waste regulated under any Environmental Law.

"Insurance Policies": All insurance policies maintained by the Company and its Subsidiaries, including, but not limited to, general liability, product liability, comprehensive general liability and umbrella insurance policies.

"ICC": As defined in Schedule 4.

"Intellectual Property Rights": As defined in Section 2.11.

"Law": Any applicable U.S. federal, state or local or any foreign (including the People's Republic of China) statute, code, ordinance, decree, rule, regulation or general principle of common or civil law or equity.

"Leases": As defined in Section 2.23.

"LIBOR": The rate per annum quoted by Bloomberg, or any other comparable services based upon quotes from the London Interbank Offered Rate from the British Bankers Association as quoted for U.S. Dollars by for determining the one (1) month LIBOR rate. The Index is to be strictly interpreted and is not intended to serve any other purpose other than providing an index to determine the interest rate used herein.

"Lien": Any mortgage, pledge, hypothecation, hypothec, right of others, claim, security interest, encumbrance, lease, sublease, license, occupancy agreement, adverse claim or interest, easement, right-of-way, levy, covenant, encroachment, burden, deed of trust, title defect, conditional or contingent sale agreement, title retention agreement, voting trust agreement, interest, equity, option, lien, right of first refusal, charge or other restrictions or limitations of any nature whatsoever, other than restrictions on the offer and sale of securities under U.S. Federal and state securities Laws.

"Loan": As defined in Recital C.

"Loan Execution": The entry into the Loan by the parties thereto.

"Loan Procurement Expenses": As defined in Section 1.1(c).

"Material Adverse Effect": Any event, circumstance, change or effect that, individually or in the aggregate, (a) has a material adverse effect on the business, assets, operations, properties or condition (financial or otherwise) of the Company and its Subsidiaries, taken as a whole; provided, however, that no event, circumstance, change or effect arising from the following, either alone or in combination, shall be taken into account when determining whether a Material Adverse Effect has occurred or is reasonably likely to occur: (i) any changes in general United States or global economic conditions, (ii) changes in the industries in which the Company operates, (iii) any change in GAAP or interpretation thereof, (iv) the execution, delivery, pendency or public announcement of this Agreement or the consummation of the transactions contemplated hereby, (v) acts of terrorism or armed hostilities,

Table of Contents

including pursuant to an act of war (whether or not declared), (vi) any change in the market price or trading volume of Common Stock, in any case taken by itself (it being understood that any event, circumstance, change or effect giving rise or contributing to such failure that is not otherwise excluded from the definition of "Material Adverse Effect" may be taken into account), or (vii) any failure by the Company to meet any internal or published projections or forecasts of revenues, earnings or other financial performance, in any case in and of itself (provided that any event, circumstance, change or effect giving rise or contributing to such failure that is not otherwise excluded from the definition of "Material Adverse Effect" may be taken into account), except, in the case of clauses (i) - (iii), to the extent the Company and its Subsidiaries, taken as a whole, are disproportionately impacted thereby relative to other entities operating in the same industry or industries in which the Company and its Subsidiaries operate (in which case the incremental disproportionate impact or impacts may be taken into account in determining whether a "Material Adverse Effect" has occurred), or (b) prevents or materially delays the consummation of the transactions contemplated by this Agreement and the other Transaction Documents; provided, however, that in any event, a Material Adverse Effect shall be deemed to occur if the damages incurred by Purchaser in connection with a breach by the Company of one or more of its representations and warranties in this Agreement exceeds either \$2,000,000 individually or \$5,000,000 in the aggregate.

```
"Material Contracts": Contracts of the type that are described in clauses (a) through (1) of Section 2.13.
```

"Molybdenum Supply Agreement": As defined in Recital E.

"Mount Hope Project": The primary molybdenum property located in Eureka County, Nevada, U.S.A.

"NDA Agreement": The Mutual Nondisclosure Agreement between Purchaser and Company executed by the Company on March 13, 2015.

"Nevada Moly": Nevada Moly, LLC, a Delaware limited liability company.

"New York City": New York, New York, U.S.A.

"Offer": As defined in Section 6.8(a).

"Offered Securities": As defined in Recital A.

"Offered Shares": As defined in Recital A.

"Offered Warrants": As defined in Recital A.

"Organizational Documents": (a) With respect to a corporation, the certificate or articles of incorporation and bylaws, or the certificate of incorporation and memorandum and articles of association or any equivalent formation or governing documents, (b) with respect to any other Person, any charter or similar document or instrument adopted or filed in connection with the creation, formation, governance or organization of a Person, including any limited partnership agreement for any limited partnership and any operating agreement or limited liability company agreement for any limited liability company, and (c) any amendment to any of the foregoing.

"Other Filings": As defined in Section 6.2(b).

"Party": As defined in the Preamble.

"Parties": As defined in the Preamble.

"Pension Plan": As defined in Section 3(2) of ERISA.

Table of Contents

"Permitted Liens": (a) Liens for (i) Taxes not yet due and payable as of the Closing Date and (ii) Taxes which are being contested in good faith and for which a reserve, determined in accordance with GAAP, has been established in the Company Financial Statements, (b) local, state and federal laws, ordinances or governmental regulations, including but not limited to, building and zoning laws, ordinances and regulations, now or hereafter in effect relating to any real property, (c) covenants, conditions and restrictions of record not violated by the existing use and improvements, and public and utility easements, (d) statutory Liens of landlords for amounts not yet due and payable, and (e) Liens of carriers, warehousemen, mechanics and materialmen incurred in the ordinary course of business for amounts not yet due and payable.

"Per Share Price": The VWAP of the shares of Common Stock for the ninety (90) days prior to the date of this Agreement.

"Person": Any individual, firm, corporation, partnership, limited liability company, trust, joint venture, or other entity.

"Plans": As defined in Section 2.2.

"POS-Minerals": POS-Minerals Corporation, a Delaware corporation.

"Pre-Closing Period": As defined in Section 5.1.

"Preferred Stock": As defined in Section 2.2.

"Prime Chinese Bank": One of the following: (a) China Development Bank, (b) the Export-Import Bank of China, (c) Bank of China, (d) China Construction Bank, (e) Industrial and Commercial Bank of China, and (f) Agricultural Bank of China.

"Proxy Statement": As defined in Section 6.2(a).

"Purchaser": As defined in the Preamble.

"Purchaser Nominees": As defined in Section 6.13.

"Real Property": As defined in Section 2.23.

"Reimbursed Purchaser Expenses": Fifty percent (50%) of all fees and expenses of law firms, investment banking firms, accountants, experts, consultants and advisors engaged by Purchaser in connection with this Agreement, the other Transaction Documents and the transactions contemplated hereby and thereby, including in connection with performing due diligence, negotiating and documenting this Agreement and the other Transaction Documents; provided, however, that the aggregate amount of Reimbursed Purchaser Expenses shall not exceed \$150,000.

"Registrable Securities": As defined in Section 4.1.

"Registration Statement": As defined in Section 4.1.

"Release": Any disposing, placing, releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, disposing, migration or dumping of any Hazardous Materials.

"*Rights*": Securities of the Company exercisable, convertible or exchangeable for or into Equity Securities (with or without consideration) or that carry any right to subscribe for or acquire Equity Securities; *provided*, *however*, Rights shall exclude SARs.

"Rules": As defined in Schedule 5.

"Rule 415": As defined in Section 4.1.

Table of Contents

"SAR": A stock appreciation right issued by the Company under its Plans.

"SEC": Securities and Exchange Commission.

"SEC Reports": As defined in Section 2.6(a).

"Securities Act": As defined in Section 3.2(a).

"Signing Date": As defined in the Preamble.

"Software": Any computer software of any kind and in any form (including source code and executable code), and all related documentation.

"Stock Purchase Price": The product of (i) the Offered Shares, multiplied by (ii) the Per Share Price.

"Stockholder Agreement": As defined in Recital F.

"Stockholder Approval": As defined in Section 6.1.

"Stockholders Meeting": As defined in Section 6.1.

"Subsidiary" or "Subsidiaries" of Purchaser, the Company or any other person means any corporation, partnership, joint venture or other legal entity of which Purchaser, the Company or such other person, as the case may be (either alone or through or together with any other subsidiary), owns, directly or indirectly, a majority of the stock or other equity interests the holders of which are generally entitled to vote for the election of the board of directors or other governing body of such corporation or other legal entity.

"Superior Proposal": Any Alternative Proposal made by a third party which was not solicited in violation of Sections 6.7 and 6.8, that the Board determines in its good faith judgment (after consultation with the Company's outside legal and financial advisors) is more favorable to the Company's stockholders from a financial point of view than the transactions contemplated by this Agreement (including any changes to the terms of this Agreement proposed by Purchaser in response to such Alternative Proposal or otherwise), after considering, among other things, the financial, legal and regulatory aspects of such proposal, whether the Person making the proposal has the financial wherewithal or the ability to obtain through responsible sources the financial wherewithal to consummate the proposal and whether the Alternative Proposal is reasonably likely to be completed as proposed on a timely basis.

"Suspension Notice": As defined in Section 4.4.

"Tangible Assets": As defined in Section 2.24.

"Tax" or "Taxes:" All forms of taxation or duties imposed, or required to be collected or withheld, including without limitation any United States federal, state or local, or non-United States, income, gross receipts, franchise, estimated, alternative minimum, add-on minimum, sales, use, transfer, registration, value added, excise, natural resources, severance, stamp, withholding, occupation, premium, windfall profit, environmental, customs, duties, real property, personal property, capital stock, net worth, intangibles, social security, unemployment, disability, payroll, license, employee or other tax or similar levy, of any kind whatsoever, together with any interest, penalties or additions to tax in respect of the foregoing and any transferee liability in respect of the foregoing payable by reason of contract, assumption, transferee liability, operation of Law, Section 1.1502-6(a) of the Treasury Regulations (or any predecessor or successor thereof or any analogous or similar provision under Law) or otherwise.

"Tax Return(s)": Any return, declaration, report, claim for refund, information return or other document (including any related or supporting estimates, elections, schedules, statements or

Table of Contents

information) filed or required to be filed in connection with the determination, assessment or collection of any Tax or the administration of any Law relating to any Tax, and where permitted or required, combined or consolidated returns for any group of entities.

"Trading Price": On any trading day, the daily volume weighted average price for the Common Stock on NYSE MKT during such trading day beginning at 9:30:01 a.m., New York City time (or such other official open of trading established by NYSE MKT) and ending at 4:00 p.m., New York City time (or such other official close of trading established by NYSE MKT) as reported by Bloomberg Financial Services through its "Volume at Price" function.

"*Transaction Documents*": This Agreement, the Stockholder Agreement, the Warrant Agreement, the Expense Reimbursement Agreement and the Molybdenum Supply Agreement.

"Tribunal": As defined in Schedule 4.

"United States" or "U.S." or "U.S.A.": The United States of America.

"*US\$*" or "\$": Dollars.

"Warrant Agreement": As defined in Section 1.4(a)(iii).

"Warrants": As defined in Section 2.2.

"VWAP": For any period of measurement, the arithmetic average (rounded to the nearest whole cent) of the Trading Prices of shares of Common Stock for each consecutive business day during the period of measurement on NYSE MKT immediately preceding the date in question.

Table of Contents

Schedule 2

(to Investment and Securities Purchase Agreement)

General Moly, Inc. Loan Summary of Principal Terms and Conditions

Borrower: General Moly Inc. (the "Company")

Guarantor Amer International Group Co. Ltd. ("Purchaser"), as required by Lender.

Lender: One or more Prime Chinese Banks.

Amount: An amount necessary to fund the Company's share of costs (including financing costs) related to the development of the

Mt. Hope Project, or approximately US \$700 Million.

Maturity: 12¹/₄ years from first drawdown.

Purpose: To fund the Company's share of costs (including financing costs) related to development of the Mt. Hope Project.

Availability: Loan Agreement to be signed no later than the date that is the two year anniversary of the Closing Date, available for

drawdown from that date until 30 months thereafter.

Repayment: Semi-annual principal repayments, with the first repayment 30 months after signing of the Loan Agreement. The first

two principal installments will be US \$1 million each, whilst the Mt. Hope Project ramps up to full production;

subsequently installments will be equal.

Front End Fee: Not greater than RMB 30 million.

Commitment Fee: Not greater that 1% pa on the undrawn Loan amount.

Interest Rate: 6 month LIBOR plus a spread not greater that 4%.

Security: Pledge of the assets of the Company and Nevada Moly, its wholly owned subsidiary which holds an 80% interest in

Eureka Moly.

Security will not include any pledge of the assets of Eureka Moly.

Prepayments: Voluntary at any time without penalty.

Mandatory 40% of the Company's share of project cash flow after payment of all project related costs (including operating, lease, financing, marketing and distribution, administration, taxes, hedging) and funding of reserve accounts

and permitted capital expenditures under the Loan Agreement.

Hanlong Shares: The Company shall use reasonable best efforts to purchase from the Export-Import Bank of China ("CEXIM") the

Hanlong Loan, together with all collateral pledged to secure the Hanlong Loan (including the Hanlong Shares) and all other rights of CEXIM under the Hanlong Loan. The purchase price paid by the Company to CEXIM shall be a secured subordinated promissory note made by the Company payable to CEXIM in the original principal amount equal to the

outstanding balance of, and accrued and unpaid interest under, the Hanlong Loan.

Table of Contents

"**Hanlong Loan**": The loan made by CEXIM to Hanlong in 2010 secured by, among other things, a pledge of the Hanlong Shares.

"Hanlong Shares": The 11,843,341 shares of Common Stock held of record by Hanlong (USA) Mining Investment, Inc. and pledged to CEXIM to secure repayment of the Hanlong Loan.

Other Provisions:

Covenants, events of default, and other provisions to be similar to those previously agreed with China Development

Table of Contents

Schedule 3 (to Investment and Securities Purchase Agreement)

Required Approvals

- 1 Approval from or registration with the National Development Reform Commission of the People's Republic of China.
- 2. Approval from or registration with the Ministry of Commerce of the People's Republic of China.
- 3. Approval from or registration with the State Administration of Foreign Exchange of the People's Republic of China.

Table of Contents

Schedule 4 (to Investment and Securities Purchase Agreement)

Dispute Resolution

- (a) All Disputes not otherwise settled by agreement between the Parties shall be finally settled by binding arbitration under the Rules of Arbitration (the "*Rules*") of the International Chamber of Commerce (the "*ICC*") by arbitrators appointed in accordance with the Rules then in effect, except to the extent the Rules conflict with the provisions of this *Schedule 4*, in which event the provisions of this *Schedule 4* shall control. The Parties specifically agree that any time before the Tribunal has been appointed, a party may seek a preliminary injunction or other interim relief before a court of competent jurisdiction to the extent necessary to preserve the status quo or to preserve a party's ability to obtain meaningful relief pending the outcome of the arbitration under this *Schedule 4*.
- (b) In the event of any Dispute and before arbitration may be commenced, upon notice by any Party to the other Party to such Dispute (the "Dispute Negotiation Notice"), such Dispute must immediately be referred to one representative of the executive management of the Company designated by the Company and one representative of the executive management of Purchaser designated by Purchaser, who must be authorized to settle the Dispute. Such representatives must promptly meet in a good faith effort to resolve the Dispute. If the representatives so designated do not resolve the Dispute within ten (10) Business Days after the delivery of the Dispute Negotiation Notice, the Dispute will be exclusively and finally resolved by binding arbitration as described in this Schedule 4.
- (c) The arbitration shall be conducted before a panel of three arbitrators, each of whom must be fluent in English and be neutral and independent of the parties to the Dispute (the "*Tribunal*"). The claimant shall appoint one arbitrator and the respondent must appoint one arbitrator, as provided in the Rules. The third arbitrator shall be selected by the two arbitrators so appointed; *provided* that if the two arbitrators so appointed fail to select the third arbitrator within thirty (30) days after the date on which the second of such two arbitrators is appointed, then the third arbitrator shall be appointed by the ICC Court. The third arbitrator, regardless of how selected, shall chair the Tribunal.
- (d) The place of arbitration shall be Hong Kong SAR. The arbitration shall be conducted in English; *provided* that (i) any party thereto, at its cost, may provide for the translation of the proceedings into a language other than English, (ii) any party thereto may elect to submit documents or other information to the Tribunal in English, and (iii) any witness whose native language is not English may elect to give testimony in English or in such witness's native language, with simultaneous interpretation into English if such testimony is given in such native language. If simultaneous interpretation is so made, the interpreter will be appointed by the Tribunal. Each party to any arbitration or its legal counsel may also hire an interpreter at such party's own expense, and may participate in the examination and cross-examination of witnesses at any hearing.
- (e) Unless the Tribunal orders an earlier date or the parties to the arbitration otherwise agree, not less than thirty (30) days before the beginning of the evidentiary hearing, each party to the arbitration shall submit to the other parties to the arbitration the documents that it may use at the hearing and a list of the witnesses whom such party may call at the hearing.
- (f) The Tribunal shall have no authority to award any indirect, incidental, special, consequential, or punitive damages, and each Party irrevocably waives its right to recover any such damages.
- (g) The decision of the Tribunal will be final and binding. Any award made in the arbitration will be enforceable in any court of competent jurisdiction, including without limitation in any jurisdiction where one or more of the parties is domiciled or has assets. For purposes of an action for recognition or enforcement of the award, each party irrevocably waives any objection that it might have to personal jurisdiction in the courts of a jurisdiction where one or more of the parties is domiciled or has assets.

Schedule	4
A-45	

Table of Contents

(h) Notwithstanding the pendency of any arbitration, the obligations of the Parties under this Agreement will remain in full force and effect; *provided* that no Party will be considered in default under this Agreement (except for defaults for the payment of money) during the pendency of an arbitration specifically relating to the default. The non-prevailing party in any arbitration shall pay all costs and expenses in connection with such arbitration, unless the Tribunal determines otherwise.

The parties shall use their reasonable best efforts to encourage the Tribunal to enter a final award resolving the Dispute within 90 days from the appointment of the Tribunal. Notwithstanding any provision to the contrary in this *Schedule 4*, the parties to any arbitration under this *Schedule 4* may agree at any time to discontinue and terminate such arbitration.

Execution Version

AMENDMENT NO. 1 TO INVESTMENT AND SECURITIES PURCHASE AGREEMENT

THIS AMENDMENT NO. 1 TO INVESTMENT AND SECURITIES PURCHASE AGREEMENT, dated November 2, 2015 (this "Amendment"), is between General Moly, Inc., a Delaware corporation, (the "Company"), and Amer International Group Co., Ltd., a limited liability company organized under the laws of the People's Republic of China ("Purchaser"). Capitalized terms used, but not defined herein, shall have the meanings assigned to them in the Agreement, as defined below.

RECITALS

- A. The Company and Purchaser are parties to the Investment and Securities Purchase Agreement, dated April 17, 2015 (the "Agreement"); and
- B. The parties desire to amend the Agreement to, among other things, provide for the Offered Securities to be issued in three separate tranches, to make related changes to the board representation provisions of the Agreement, and to modify specified conditions precedent in the Agreement.

Accordingly, in consideration of the mutual covenants contained in this Amendment, the parties intending to be legally bound agree as follows.

AGREEMENT

- 1. Recital A of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "A. The Company desires to issue and sell to Purchaser and Purchaser desires to acquire from the Company, on the terms and subject to the conditions in this Agreement, (1) 40,000,000 shares (the "*Offered Shares*") of the Company's common stock, par value \$0.001 per share (the "*Common Stock*") at an aggregate purchase price of \$20,000,000, and (2) warrants to purchase 80,000,000 shares of Common Stock (the "*Offered Warrants*") exercisable at the Per Share Price (the Offered Shares, together with the Offered Warrants, the "*Offered Securities*")."
- 2. Recital B of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "B. The Offered Securities will be issued in three tranches: (1) the first tranche ("*Tranche 1*") will consist of 13,333,333 Offered Shares (the "*Tranche 1 Shares*") and 80,000,000 Offered Warrants (together with the Tranche 1 Shares, the "*Tranche 1 Securities*"), (2) the second tranche ("*Tranche 2*") will consist of 12,000,000 Offered Shares (the "*Tranche 2 Securities*"); and (3) the third tranche ("*Tranche 3*") will consist of 14,666,667 Offered Shares (the "*Tranche 3 Securities*"). The Tranche 1 Securities will be issued upon the satisfaction of the conditions set forth in *Section 7.1*, the Tranche 2 Securities will be issued upon the satisfaction of the conditions set forth in *Section 7.2* and the Tranche 3 Securities will be issued upon the satisfaction of the conditions set forth in *Section 7.3*. Upon the Loan Execution, the Offered Securities will constitute approximately 51% of the Company's fully diluted shares of Common Stock."
- 3. Recital E of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "E. At the Tranche 1 Closing, the Company and Purchaser will enter into a stockholder agreement with respect to certain matters relating to the acquisition and disposition of the Offered Securities (and other shares of Common Stock, if any, owned by Purchaser or any of its Affiliates) and governance of the Company (the "Stockholder Agreement") substantially in the form attached hereto as *Exhibit B*."

Table of Contents

- 4. Section 1.1 of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "1.1 Purchase and Sale of Offered Securities; Closing Payment; Expense Reimbursement Account.
 - (a) Offer and Sale. On the terms and subject to the conditions of this Agreement and in reliance upon the representations and warranties contained herein, Purchaser shall purchase from the Company, and the Company shall issue and sell to Purchaser: (i) the Tranche 1 Securities on the Tranche 1 Closing Date, (ii) the Tranche 2 Securities on the Tranche 2 Closing Date and (iii) the Tranche 3 Securities on the Tranche 3 Closing Date.
 - (b) Closing Payments.
 - (i) In exchange for the Company's issuance and sale of the Tranche 1 Securities to Purchaser on the Tranche 1 Closing Date, Purchaser shall pay to the Company in cash an amount equal to the Stock Purchase Price for the Tranche 1 Shares (the "*Tranche 1 Closing Payment*").
 - (ii) In exchange for the Company's issuance and sale of the Tranche 2 Securities to Purchaser on the Tranche 2 Closing Date, Purchaser shall pay to the Company in cash an amount equal to the Stock Purchase Price for the Tranche 2 Securities (the "*Tranche 2 Closing Payment*").
 - (i) (iii) In exchange for the Company's issuance and sale of the Tranche 3 Securities to Purchaser on the Tranche 3 Closing Date, Purchaser shall pay to the Company in cash an amount equal to (a) the Stock Purchase Price for the Tranche 3 Securities, minus (b) the Reimbursed Purchaser Expenses (such difference, the "*Tranche 3 Closing Payment*" and, collectively with the Tranche 1 Closing Payment and the Tranche 2 Closing Payment, the "*Closing Payments*").
 - (iv) In addition to making the Closing Payments, the Company shall pay Purchaser the Arrangement Fee following the Tranche 1 Closing, the Tranche 2 Closing or the Tranche 3 Closing, as applicable, pursuant to *Section 1.2*.
 - (c) Reimbursement Account.
 - (i) Within three (3) days of the Tranche 1 Closing Date, the Company shall deposit \$2,000,000 from the Tranche 1 Closing Payment into an account (the "Expense Reimbursement Account") at a bank to be mutually agreed upon by the Company and Purchaser. Purchaser and the Company shall jointly hold such account pursuant to an expense reimbursement agreement (the "Expense Reimbursement Agreement"), substantially in accordance with the terms and conditions attached hereto as Exhibit D, to be entered into at the Tranche 1 Closing between Purchaser and the Company. Under the Expense Reimbursement Agreement, funds will be released to Purchaser, the Company or third parties to reimburse such parties for reasonable expenses incurred in connection with the procurement of the Loan under Section 6.4(a) and other mutually agreed upon joint opportunity evaluations and transactions (such expenses, collectively, the "Loan Procurement Expenses").
 - (ii) On the Tranche 2 Closing Date, the Company shall deposit \$1,000,000 from the Tranche 2 Closing Payment into the Expense Reimbursement Account."
- 5. Section 1.3 of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "1.3 Closings.
 - (a) Tranche 1 Closing. Subject to the satisfaction or waiver of the conditions set forth in Section 7.1, the completion of the purchase and sale of the Tranche 1 Securities (the

Table of Contents

- "*Tranche 1 Closing*") shall occur at 10:00 a.m. local time at the offices of Bryan Cave LLP, Denver, Colorado, on the second (2nd) Business Day after the satisfaction or waiver of the conditions set forth in *Section 7.1* (other than those that by their terms are to be satisfied or waived at the Tranche 1 Closing), or at such other location, date and time as may be mutually agreed upon by the Company and Purchaser. The date of the Tranche 1 Closing is referred to herein as the "*Tranche 1 Closing Date*."
- (b) *Tranche 2 Closing*. Subject to the satisfaction or waiver of the conditions set forth in *Section 7.2*, the completion of the purchase and sale of the Tranche 2 Securities (the "*Tranche 2 Closing*") shall occur at 10:00 a.m. local time at the offices of Bryan Cave LLP, Denver, Colorado, on the second (2nd) Business Day after the satisfaction or waiver of the conditions set forth in *Section 7.2* (other than those that by their terms are to be satisfied or waived at the Tranche 2 Closing), or at such other location, date and time as may be mutually agreed upon by the Company and Purchaser. The date of the Tranche 2 Closing is referred to herein as the "*Tranche 2 Closing Date*."
- (c) *Tranche 3 Closing*. Subject to the satisfaction or waiver of the conditions set forth in *Section 7.3*, the completion of the purchase and sale of the Tranche 2 Securities (the "*Tranche 3 Closing*") shall occur at 10:00 a.m. local time at the offices of Bryan Cave LLP, Denver, Colorado, on the second (2nd) Business Day after the satisfaction or waiver of the conditions set forth in *Section 7.3* (other than those that by their terms are to be satisfied or waived at the Tranche 3 Closing), or at such other location, date and time as may be mutually agreed upon by the Company and Purchaser. The date of the Tranche 3 Closing is referred to herein as the "*Tranche 3 Closing Date*.""
- 6. Section 1.4 of the Agreement is hereby deleted in its entirety and replaced with the following:

"1.4 Tranche 1 Closing Deliveries.

- (a) Company Deliveries. At the Tranche 1 Closing, the Company shall deliver or cause to be delivered to Purchaser:
 - (i) a certificate or certificates representing the Tranche 1 Shares;
 - (ii) a receipt for the Tranche 1 Closing Payment;
 - (iii) a duly executed warrant agreement (the "*Warrant Agreement*"), substantially in the form attached hereto as *Exhibit C*, evidencing the Offered Warrants, registered in the name of Purchaser;
 - (iv) a duly executed counterpart of the Stockholder Agreement;
 - (v) a duly executed counterpart of the Expense Reimbursement Agreement;
 - (vi) a duly executed resignation effective on or before the Tranche 1 Closing Date from the director on the Board, if applicable, pursuant to *Section 6.13(a)*;
 - (vii) a good standing certificate (or its equivalent) for the Company issued by the Secretary of State of the State of Delaware and of such other applicable jurisdictions where the Company is qualified or licensed to do business or own, lease or operate property making such qualification or licensing necessary, dated as of a date within thirty (30) days prior to the Tranche 1 Closing Date;
 - (viii) copies, certified by the Secretary of the Company, of resolutions of the Board authorizing the execution and delivery of this Agreement and the other Transaction Documents, and in each case, such resolutions shall be in full force and effect and not revoked;

Table of Contents

- (ix) the documents, instruments and writings required to be delivered to Purchaser by the Company pursuant to Section 7.1(c);
 - (x) [Intentionally deleted]; and
- (xi) such other previously undelivered documents reasonably requested by Purchaser to be delivered by the Company to Purchaser at or prior to the Tranche 1 Closing in connection with this Agreement or the other Transaction Documents to which the Company is a party.
- (b) Purchaser Deliveries. At the Tranche 1 Closing, Purchaser shall deliver or cause to be delivered to the Company:
 - (i) the Tranche 1 Closing Payment by wire transfer in immediately available funds in Dollars to an account specified by the Company in writing no less than three (3) Business Days prior to the Tranche 1 Closing;
 - (ii) a duly executed counterpart of the Warrant Agreement;
 - (iii) a duly executed counterpart of the Stockholder Agreement;
 - (iv) a duly executed counterpart of the Expense Reimbursement Agreement;
 - (v) [Intentionally deleted];
 - (vi) the documents, instruments and writings required to be delivered to the Company by Purchaser pursuant to Section 7.1(b); and
 - (vii) such other previously undelivered documents reasonably requested by the Company to be delivered by Purchaser to the Company at or prior to the Tranche 1 Closing in connection with this Agreement or the other Transaction Documents to which Purchaser is a party."
- 7. A new Section 1.5 shall be added to the Agreement and shall read as follows:

"1.5 Tranche 2 Closing Deliveries.

- (a) Company Deliveries. At the Tranche 2 Closing, the Company shall deliver or cause to be delivered to Purchaser:
 - (i) a certificate or certificates representing the Tranche 2 Securities;
 - (ii) a receipt for the Tranche 2 Closing Payment;
 - (iii) the documents, instruments and writings required to be delivered to Purchaser by the Company pursuant to Section 7.2(c); and
 - (iv) such other previously undelivered documents reasonably requested by Purchaser to be delivered by the Company to Purchaser at or prior to the Tranche 2 Closing in connection with this Agreement or the other Transaction Documents to which the Company is a party.
- (b) Purchaser Deliveries. At the Tranche 2 Closing, Purchaser shall deliver or cause to be delivered to the Company:
 - (i) the Tranche 2 Closing Payment by wire transfer in immediately available funds in Dollars to an account specified by the Company in writing no less than three (3) Business Days prior to the Tranche 2 Closing;
 - (ii) the documents, instruments and writings required to be delivered to the Company by Purchaser pursuant to Section 7.2(b); and

Table of Contents

- (iii) such other previously undelivered documents reasonably requested by the Company to be delivered by Purchaser to the Company at or prior to the Tranche 2 Closing in connection with this Agreement or the other Transaction Documents to which Purchaser is a party."
- 8. A new Section 1.6 shall be added to the Agreement and shall read as follows:

"1.6 Tranche 3 Closing Deliveries.

- (a) Company Deliveries. At the Tranche 3 Closing, the Company shall deliver or cause to be delivered to Purchaser:
 - (i) a certificate or certificates representing the Tranche 3 Securities;
 - (ii) a receipt for the Tranche 3 Closing Payment;
 - (iii) a duly executed resignation effective as of the Tranche 3 Closing Date from the director on the Board, if applicable, pursuant to *Section 6.13(b)*;
 - (iv) the documents, instruments and writings required to be delivered to Purchaser by the Company pursuant to Section 7.3(c); and
 - (v) such other previously undelivered documents reasonably requested by Purchaser to be delivered by the Company to Purchaser at or prior to the Tranche 3 Closing in connection with this Agreement or the other Transaction Documents to which the Company is a party.
- (b) Purchaser Deliveries. At the Tranche 3 Closing, Purchaser shall deliver or cause to be delivered to the Company:
 - (i) the Tranche 3 Closing Payment by wire transfer in immediately available funds in Dollars to an account specified by the Company in writing no less than three (3) Business Days prior to the Tranche 3 Closing;
 - (ii) the documents, instruments and writings required to be delivered to the Company by Purchaser pursuant to Section 7.3(b); and
 - (iii) such other previously undelivered documents reasonably requested by the Company to be delivered by Purchaser to the Company at or prior to the Tranche 3 Closing in connection with this Agreement or the other Transaction Documents to which Purchaser is a party."
- 9. The introductory paragraph to Article II of the Agreement is hereby amended by replacing the phrase "Closing Date" in the third (3rd) line of the paragraph with "Tranche 1 Closing Date."
- 10. Section 2.2 of the Agreement is hereby amended by (a) adding the phrase "As of March 31, 2015," at the beginning of the first (1st) line of the paragraph and (b) replacing the phrase "Closing Date" in the twenty-fourth (24th) line of the paragraph with "Tranche 1 Closing Date."
- 11. Section 4.1 of the Agreement is hereby amended by replacing the phrase "Closing Date" in the second (2^{nd}) line of the paragraph with "Tranche 1 Closing Date."
- 12. The introductory paragraph of Section 5.1 of the Agreement is hereby amended by replacing the word "Closing" in the second (2nd) line of the paragraph with "Tranche 1 Closing."
- 13. Section 6.3 of the Agreement is hereby amended by replacing the word "Closing" in the third (3rd) line of the paragraph with "Tranche 1 Closing."

Table of Contents

- 14. Section 6.5(b) of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "(b) [Intentionally deleted]."
- 15. Section 6.6(a) of the Agreement is hereby amended by (a) replacing the phrase "the Closing" in the seventh (7th) line of the paragraph with "the Tranche 1 Closing, the Tranche 2 Closing or the Tranche 3 Closing" and (b) replacing the phrase "the Closing" in the tenth (10th) line of the paragraph with "the Tranche 1 Closing, the Tranche 2 Closing or the Tranche 3 Closing."
- 16. Section 6.10 of the Agreement is hereby amended by (a) replacing the phrase "Closing Date" in the fourth (4th) line of the paragraph with "Tranche 1 Closing Date" and (b) replacing the phrase "Section 6.9" in the tenth (10th) line of the paragraph with "Section 6.10."
- 17. Section 6.11 of the Agreement is hereby amended by replacing the phrase "the Closing" in the first (1st) line of the paragraph with "the Tranche 1 Closing, the Tranche 2 Closing or the Tranche 3 Closing."
 - 18. Section 6.13 of the Agreement is hereby deleted in its entirety and replaced with the following:

"6.13 Board Representation.

- (a) The Company shall provide, as of the Tranche 1 Closing, the resignation of the director designated by the Company in conjunction with the appointment of the Purchaser Nominee to be appointed at the Tranche 1 Closing, if necessary to reduce the Board to six (6) members. The Company agrees to use reasonable best efforts to cause the Board, at the Tranche 1 Closing, to be comprised of seven (7) individuals, of whom one (1) individual shall be nominated by Purchaser consistent with the terms of the Stockholder Agreement, subject to requirements of applicable Law and under the rules of the NYSE MKT.
- (b) If necessary, the Company shall provide, as of the Tranche 3 Closing, the resignation of the director designated by the Company in conjunction with the appointment of the Purchaser Nominee to be appointed at the Tranche 3 Closing. The Company agrees to use reasonable best efforts to cause the Board, at the Tranche 3 Closing, to be comprised of eight (8) individuals, of whom one (1) individual, in addition to the individual nominated pursuant to Section 6.13(a), shall be nominated by Purchaser consistent with the terms of the Stockholder Agreement (the "*Purchaser Nominees*"), subject to requirements of applicable Law and under the rules of the NYSE MKT."
- 19. Section 7.1 of the Agreement is hereby deleted in its entirety and replaced with the following:

"7.1 Tranche 1 Closing.

- (a) Conditions to the Obligations of Each Party. The obligations of each Party to effect the transactions relating to the Tranche 1 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following conditions:
 - (i) No Order. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any law (whether temporary, preliminary or permanent) which is then in effect and has the effect of making the transactions contemplated by this Agreement or the other Transaction Documents illegal or otherwise restricting, preventing or prohibiting consummation of the transactions contemplated by this Agreement or the other Transaction Documents.
 - (ii) Stockholder Approval. The Company shall have received the Stockholder Approval.

Table of Contents

- (iii) Required Approvals. Purchaser shall have received the approvals set forth in Schedule 3.
- (b) Conditions to the Obligations of the Company. The obligations of the Company to effect the transactions relating to the Tranche 1 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Representations and Warranties. Each of the representations and warranties of Purchaser contained in this Agreement that are qualified by materiality or Material Adverse Effect shall be true and correct as of the Signing Date and as of the Tranche 1 Closing as though made on and as of the Tranche 1 Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date), and each of the representations and warranties which is not so qualified shall be true and correct in all material respects as of the Signing Date and as of the Tranche 1 Closing as though made on and as of the Tranche 1 Closing (except that those representations and warranties which address matters only as of a particular date need only remain true and correct in all material respects as of such date).
 - (ii) Agreements and Covenants. Purchaser shall have performed in all material respects, all obligations and complied with, in all material respects, its agreements and covenants to be performed or complied with by it under this Agreement on or prior to the Tranche 1 Closing.
 - (iii) *Tranche 1 Closing Deliveries*. Purchaser shall have made all of the deliveries contemplated by *Section 1.4(b)*.
 - (iv) Officer's Certificate. Purchaser shall have delivered to the Company a certificate, dated the Tranche 1 Closing Date, signed by an officer of Purchaser, certifying as to the satisfaction of the conditions specified in Sections 7.1(b)(i) and (ii).
- (c) Conditions to the Obligations of Purchaser. The obligations of Purchaser to effect the transactions relating to the Tranche 1 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Representations and Warranties. Each of the representations and warranties of the Company contained in this Agreement that are qualified by materiality or Material Adverse Effect shall be true and correct as of the Signing Date and as of the Tranche 1 Closing as though made on and as of the Tranche 1 Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date) and each of the representations and warranties which is not so qualified shall be true and correct in all material respects as of the Signing Date and as of the Tranche 1 Closing as though made on and as of the Tranche 1 Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date); provided, however, that with respect to Sections 2.4(b) and 2.17, all references to the Toronto Stock Exchange shall not apply if the Company has delisted, or is in the process of delisting from, the Toronto Stock Exchange.
 - (ii) Agreements and Covenants. Each of the Company and each of its Subsidiaries shall have performed, in all material respects, all obligations and complied with, in all material respects, its agreements and covenants to be performed or complied with by it under this Agreement on or prior to the Tranche 1 Closing.

Table of Contents

- (iii) Stockholder Approval. Purchaser shall have received a true and complete copy of the resolutions of the stockholders of the Company adopting the Stockholder Approval, certified by the Secretary or an Assistant Secretary of the Company.
 - (iv) [Intentionally deleted].
- (v) NYSE MKT Listing. The Company shall have filed an application for the listing of the Offered Securities with the NYSE MKT and shall have received notification from NYSE MKT that the Offered Securities have been approved for listing, subject to official notice of issuance.
- (vi) Tranche 1 Closing Deliveries. The Company shall have made all of the deliveries contemplated by Section 1.4(a).
- (vii) Officer's Certificate. The Company shall have delivered to Purchaser a certificate, dated the Tranche 1 Closing Date, signed by an officer of the Company, certifying as to the satisfaction of the conditions specified in Sections 7.1(c)(i), (ii) and (viii).
- (viii) *No Material Adverse Effect.* Since the date of this Agreement, there shall not have occurred any Material Adverse Effect, or any change, event, condition, state of facts or development that may, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (ix) Resignation of Directors; Re-Constitution of the Board. If necessary such that the Board of Directors shall constitute six (6) members immediately prior to the Tranche 1 Closing, the director of the Company identified by the Company to Purchaser shall have submitted his resignation in writing to the Company with copies to Purchaser. Such resignation shall be effective as of or before the Tranche 1 Closing. The Purchaser Nominee to be appointed at the Tranche 1 Closing shall have been appointed or elected as one (1) of the seven (7) members of the Board effective as of the Tranche 1 Closing Date."
- 20. A new Section 7.2 shall be added to the Agreement and shall read as follows:

"7.2 Tranche 2 Closing.

- (a) Conditions to the Obligations of Each Party. The obligations of each Party to effect the transactions relating to the Tranche 2 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following conditions:
 - (i) No Order. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any law (whether temporary, preliminary or permanent) which is then in effect and has the effect of making the transactions contemplated by this Agreement or the other Transaction Documents illegal or otherwise restricting, preventing or prohibiting consummation of the transactions contemplated by this Agreement or the other Transaction Documents.
- (b) Conditions to the Obligations of the Company. The obligations of the Company to effect the transactions relating to the Tranche 2 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Tranche 2 Closing Deliveries. Purchaser shall have made all of the deliveries contemplated by Section 1.5(b).

Table of Contents

- (c) Conditions to the Obligations of Purchaser. The obligations of Purchaser to effect the transactions relating to the Tranche 2 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Tranche 2 Closing Deliveries. The Company shall have made all of the deliveries contemplated by Section 1.5(a).
 - (ii) Officer's Certificate. The Company shall have delivered to Purchaser a certificate, dated the Tranche 2 Closing Date, signed by an officer of the Company, certifying as to the satisfaction of the conditions specified in Sections 7.2(c)(iii), (iv) and (v).
 - (iii) *Price of Molybdenum*. The price of molybdenum oxide shall have been at least \$8.00 per pound as published by Platt's Metals Week for thirty (30) consecutive days.
 - (iv) Water Permits. The Nevada State Water Engineer shall have issued the water permits for the Mt. Hope Project.
 - (v) Stockholder Approval. If required, the Company shall have received stockholder approval for the transactions contemplated by this Agreement, as amended."
- 21. A new Section 7.3 shall be added to the Agreement and shall read as follows:

"7.3 Tranche 3 Closing.

- (a) Conditions to the Obligations of Each Party. The obligations of each Party to effect the transactions relating to the Tranche 3 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following conditions:
 - (i) No Order. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any law (whether temporary, preliminary or permanent) which is then in effect and has the effect of making the transactions contemplated by this Agreement or the other Transaction Documents illegal or otherwise restricting, preventing or prohibiting consummation of the transactions contemplated by this Agreement or the other Transaction Documents.
- (b) Conditions to the Obligations of the Company. The obligations of the Company to effect the transactions relating to the Tranche 3 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Tranche 3 Closing Deliveries. Purchaser shall have made all of the deliveries contemplated by Section 1.6(b).
- (c) Conditions to the Obligations of Purchaser. The obligations of Purchaser to effect the transactions relating to the Tranche 3 Closing contemplated by this Agreement shall be subject to the satisfaction or waiver of the following additional conditions:
 - (i) Tranche 3 Closing Deliveries. The Company shall have made all of the deliveries contemplated by Section 1.6(a).
 - (ii) Officer's Certificate. The Company shall have delivered to Purchaser a certificate, dated the Tranche 3 Closing Date, signed by an officer of the Company, certifying as to the satisfaction of the conditions specified in Section 7.3(c)(iv), (v) and (vi).
 - (iii) Resignation of Directors; Re-Constitution of the Board. If applicable, the director of the Company identified by the Company to Purchaser shall have submitted his resignation in writing to the Company with copies to Purchaser. Such resignation shall be

Table of Contents

effective as of the Tranche 3 Closing. The Purchaser Nominee to be appointed at the Tranche 3 Closing shall have been appointed or elected as one (1) of the eight (8) members of the Board effective as of the Tranche 3 Closing Date.

- (iv) *Price of Molybdenum.* The price of molybdenum oxide shall have been at least \$12.00 per pound as published by Platt's Metals Week for thirty (30) consecutive days.
- (v) Resolution of Litigation. The pending litigation relating to the decision of the Nevada State Water Engineer to grant the water permits for the Mt. Hope Project shall have become final and nonappealable or dismissed with prejudice.
- (vi) Stockholder Approval. If required, the Company shall have received stockholder approval for the transactions contemplated by this Agreement, as amended."
- 22. The introductory paragraph to Section 8.1 of the Agreement is hereby amended by replacing the word "Closing" in the second (2nd) line of the paragraph with "Tranche 3 Closing."
 - 23. Section 8.1(b)(i) of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "(i) the Tranche 1 Closing shall not have occurred on or before December 31, 2015, the Tranche 2 Closing shall not have occurred on or before the date which is two (2) years from the Tranche 1 Closing Date, or the Tranche 3 Closing shall not have occurred on or before the date which is three (3) years from the Tranche 1 Closing Date;"
 - 24. The definition of "Closing" set forth in Schedule 1 to the Agreement is hereby deleted in its entirety and replaced with the following:
 - ""Tranche 1 Closing," "Tranche 2 Closing" and "Tranche 3 Closing": As defined in Section 1.3."
- 25. The definition of "Closing Date" set forth in Schedule 1 to the Agreement is hereby deleted in its entirety and replaced with the following:
 - ""Tranche 1 Closing Date," "Tranche 2 Closing Date" and "Tranche 3 Closing Date": As defined in Section 1.3."
- 26. The definition of "Closing Payment" set forth in Schedule 1 to the Agreement is hereby deleted in its entirety and replaced with the following:
 - ""Closing Payments": As defined in Section 1.1(b)."
- 27. The definition of "Company Break Fee" set forth in Schedule 1 to the Agreement is hereby deleted in its entirety and replaced with the following:
 - ""Company Break Fee": An amount equal to five percent (5%) of the aggregate Stock Purchase Prices for Tranche 1, Tranche 2 and Tranche 3, payable in cash.
- 28. The definition of "Permitted Liens" set forth in Schedule 1 to the Agreement is hereby amended by replacing the phrase "Closing Date" in the first (1st) line of the paragraph with "Tranche 1 Closing Date."
- 29. The definition of "Per Share Price" set forth in Schedule 1 to the Agreement is hereby deleted in its entirety and replaced with the following:
 - ""Per Share Price": (i) With respect to the Tranche 1 Shares, \$0.30; (ii) with respect to the Tranche 2 Shares, \$0.50; (iii) with respect to the Tranche 3 Shares, \$0.6818; and (iv) with respect to the Offered Warrants, the VWAP of the shares of Common Stock for the ninety (90) days prior to the date of this Agreement."

Table of Contents

- 30. The definition of "Stock Purchase Price" set forth in Schedule 1 to the Agreement is hereby deleted in its entirety and replaced with the following:
 - ""Stock Purchase Price": The product of (i) the Tranche 1 Shares, the Tranche 2 Securities or the Tranche 3 Securities, as applicable, multiplied by (ii) the Per Share Price."
- 31. The definition of "Availability" set forth in Schedule 2 to the Agreement is hereby deleted in its entirety and replaced with the following:
- "Availability: Loan Agreement to be signed no later than the date that is the two year anniversary of the Tranche 1 Closing Date, available for drawdown from that date until 30 months thereafter."
- 32. The Form of Stockholder Agreement attached as Exhibit B to the Agreement is hereby deleted in its entirety and replaced with the Form of Stockholder Agreement attached as *Exhibit B* to this Amendment.
 - 33. The definition of "Date" set forth in Exhibit D to the Agreement is hereby deleted in its entirety and replaced with the following:
- "Date: Tranche 1 Closing Date of the Investment and Securities Purchase Agreement (the "Purchase Agreement")."
 - 34. The definition of "Term" set forth in Exhibit D to the Agreement is hereby deleted in its entirety and replaced with the following:
- "Term: Commences on Tranche 1 Closing Date and terminates on the earlier of (a) execution of Loan Agreement with one or more Prime Chinese Banks and (b) two years from the Tranche 1 Closing Date (the "Term")."
- 35. In reference to Section 2.1 of the Agreement, on February 6, 2015, the Company amended its bylaws to add a new Section 3.12 allowing the Board, by affirmative vote of a majority of the directors, to grant a leave of absence to a director for a specified period of time that shall not exceed six months. The amendment gives the Compensation Committee discretion to determine appropriate adjustments to compensation and stock awards with respect to any director granted a leave of absence.

Except as specifically set forth in this Amendment, the Agreement shall remain in full force and effect.

[Signature page follows]

Table of Contents

IN WITNESS WHEREOF, the parties have executed this Amendment to be effective as of the date first above written.

GENERAL MOLY, INC.

By: /s/ BRUCE D. HANSEN

Name: Bruce D. Hansen
Title: Chief Executive Officer

AMER INTERNATIONAL GROUP CO., LTD.

By: /s/ WENYIN WANG

Name: Wenyin Wang

Title:

[Signature Page to Amendment No. 1 to Investment and Securities Purchase Agreement]

AMENDMENT NO. 2 TO INVESTMENT AND SECURITIES PURCHASE AGREEMENT

THIS AMENDMENT NO. 2 TO INVESTMENT AND SECURITIES PURCHASE AGREEMENT, dated August 7, 2017 (this "Second Amendment"), is between General Moly, Inc., a Delaware corporation, (the "Company"), and Amer International Group Co., Ltd., a limited liability company organized under the laws of the People's Republic of China ("Purchaser"). Capitalized terms used, but not defined herein, shall have the meanings assigned to them in the Agreement, as defined below.

RECITALS

- A. The Company and Purchaser are parties to the Investment and Securities Purchase Agreement, dated April 17, 2015, as amended by Amendment No. 1 to Investment and Securities Purchase Agreement on November 2, 2015 (the "Agreement"); and
- B. The parties desire to further amend the Agreement to, among other things, provide for the modification of specified conditions precedent to closing Tranche 2 and Tranche 3, and the definition of "Per Share Price" in the Agreement.

Accordingly, in consideration of the mutual covenants contained in this Amendment, the parties intending to be legally bound agree as follows.

AGREEMENT

- 1. Recital A of the Agreement is hereby amended to read as follows:
 - "A. (1) 40,000,000 shares (the "*Offered Shares*") of the Company's common stock, par value \$0.001 per share (the "*Common Stock*") at the Per Share Price;"
- 2. Recital B of the Agreement is hereby amended to read as follows:
 - "B. (2) the second tranche ("*Tranche 2*") will consist of \$6,000,000 of Offered Shares at the Per Share Price (the "*Tranche 2 Securities*"); and (3) the third tranche ("*Tranche 3*") will consist of \$10,000,000 of Offered Shares at the Per Share Price (the "*Tranche 3 Securities*")."
- 3. Section 1.1(c)(ii) of the Agreement is hereby amended to read as follows:
 - "(ii) On the Tranche 2 Closing Date, the Company shall deposit \$500,000 from the Tranche 2 Closing Payment into the Expense Reimbursement Account. On the Tranche 3 Closing Date, the Company shall deposit \$500,000 from the Tranche 3 Closing Payment into the Expense Reimbursement Account."
- 4. Section 1.3(b) of the Agreement is hereby amended to read as follows:
 - "(b) *Tranche 2 Closing*. Subject to the satisfaction or waiver of the conditions set forth in *Section 7.2*, the completion of the purchase and sale of the Tranche 2 Securities (the "*Tranche 2 Closing*") shall occur at 10:00 a.m. local time at the offices of Bryan Cave LLP, Denver, Colorado, on September 30, 2017, or at such other location, date and time as may be mutually agreed upon by the Company and Purchaser. The date of the Tranche 2 Closing is referred to herein as the "*Tranche 2 Closing Date*."
- 5. Section 7.2(b) of the Agreement is amended by adding a new subsection (ii) as follows:
 - "(ii) Warrant Amendment. The Company shall have received from Purchaser an executed Fourth Amendment to the Warrant Agreement, extending the expiration of the Warrant

Table of Contents

Agreement to the third anniversary of the issuance of the Record of Decision for the Mt. Hope Project."

- 6. Section 7.2(c)(ii) of the Agreement is amended as follows:
 - "(ii) *Officer's Certificate*. The Company shall have delivered to Purchaser a certificate, dated the Tranche 2 Closing Date, signed by an officer of the Company, certifying as to the satisfaction of the conditions specified in *Sections 7.1(c)(iii)* and (v)."
- 7. Section 7.2(c)(iii) of the Agreement is amended as follows:
 - "(iii) [Intentionally deleted]."
- 8. Section 7.2(c)(iv) of the Agreement is amended as follows:
 - "(iv) [Intentionally deleted]."
- 9. Section 7.2(c)(v) of the Agreement is deleted in its entirety and replaced with the following:
 - "(v) Stockholder Approval. If the Company deems that such is required in its reasonable discretion, the Company shall have received stockholder approval for the transactions contemplated by this Agreement, as amended."
- 10. Section 7.2(c) of the Agreement is amended by adding a new subsection (vi) as follows:
 - "(vi) Warrant Amendment. Purchaser shall have received from the Company an executed Fourth Amendment to the Warrant Agreement, extending the expiration of the Warrant Agreement to the third anniversary of the issuance of the Record of Decision for the Mt. Hope Project."
- 11. Section 7.3(c)(ii) of the Agreement is amended as follows:
 - "(ii) Officer's Certificate. The Company shall have delivered to Purchaser a certificate, dated the Tranche 3 Closing Date, signed by an officer of the Company, certifying as to the satisfaction of the conditions specified in Sections 7.1(c)(iv) and (vi)."
- 12. Section 7.3(c)(iv) of the Agreement is amended as follows:
 - "(iv) [Intentionally deleted]."
- 13. Section 7.3(c)(v) of the Agreement is amended as follows:
 - "(v) [Intentionally deleted]."
- 14. Section 7.3(c) of the Agreement is amended with the addition of a new subsection (vii), as follows:
 - "(vii) the earlier of the date upon which 1) the Parties shall have closed a mutually agreed acquisition involving more than 10 million shares of the Company's common stock as consideration for the acquisition; or 2) the Nevada State Engineer shall have issued water permits for the Mt. Hope Project.
- 15. Section 8.1(b)(i) of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "(i) the Tranche 1 Closing shall not have occurred on or before December 31, 2015, the Tranche 2 Closing shall not have occurred on or before September 30, 2017, or the Tranche 3 Closing shall not have occurred on or before the later of March 31, 2018 or 90 days after the occurrence of the earlier of the events described in 7.3(c)(vii);"

Table of Contents

- 16. The definition of "Per Share Price" in the Agreement is amended to read as follows:
 - "(ii) with respect to the Tranche 2 Shares, the volume weighted average price of the Company's Common Stock for the 30-day period ending on the date of execution of this Second Amendment;
 - (iii) with respect to the Tranche 3 Shares, \$0.50;"
- 17. The definition of "Availability" in the Agreement is amended to read as follows:

"Availability: Loan Agreement to be signed no later than the date that is the five (5) year anniversary of the Tranche 1 Closing Date, available for drawdown from that date until 30 months thereafter."

Except as specifically set forth in this Second Amendment, the Agreement shall remain in full force and effect.

[Signature page follows]

Table of Contents

IN WITNESS WHEREOF, the parties have executed this Second Amendment to be effective as of the date first above written.

GENERAL MOLY, INC.

By: /s/ BRUCE D. HANSEN

Name: Bruce D. Hansen
Title: Chief Executive Officer

AMER INTERNATIONAL GROUP CO., LTD.

By: /s/ WENYIN WANG

Name: Wenyin Wang Title: *Chairman*

[Signature Page to Amendment No. 2 to Investment and Securities Purchase Agreement]

AMENDMENT NO. 3 TO INVESTMENT AND SECURITIES PURCHASE AGREEMENT

THIS AMENDMENT NO. 3 TO INVESTMENT AND SECURITIES PURCHASE AGREEMENT, dated September 30, 2017 (this "*Third Amendment*"), is between General Moly, Inc., a Delaware corporation, (the "*Company*"), and Amer International Group Co., Ltd., a limited liability company organized under the laws of the People's Republic of China ("*Purchaser*"). Capitalized terms used, but not defined herein, shall have the meanings assigned to them in the Agreement, as defined below.

RECITALS

- A. The Company and Purchaser are parties to the Investment and Securities Purchase Agreement, dated April 17, 2015, as amended by Amendment No. 1 to Investment and Securities Purchase Agreement on November 2, 2015 and Amendment No. 2 to Investment and Securities Purchase Agreement on August 7, 2017 (the "*Agreement*"); and
 - B. The parties desire to further amend the Agreement to change the date of the Tranche 2 Closing.

Accordingly, in consideration of the mutual covenants contained in this Amendment, the parties intending to be legally bound agree as follows.

AGREEMENT

- 1. Section 1.3(b) of the Agreement is hereby amended to read as follows:
 - "(b) *Tranche 2 Closing*. Subject to the satisfaction or waiver of the conditions set forth in *Section 7.2*, the completion of the purchase and sale of the Tranche 2 Securities (the "*Tranche 2 Closing*") shall occur at 10:00 a.m. local time at the offices of Bryan Cave LLP, Denver, Colorado, on October 16, 2017, or at such other location, date and time as may be mutually agreed upon by the Company and Purchaser. The date of the Tranche 2 Closing is referred to herein as the "*Tranche 2 Closing Date*."
- 2. Section 8.1(b)(i) of the Agreement is hereby deleted in its entirety and replaced with the following:
 - "(i) the Tranche 1 Closing shall not have occurred on or before December 31, 2015, the Tranche 2 Closing shall not have occurred on or before October 16, 2017, or the Tranche 3 Closing shall not have occurred on or before the later of March 31, 2018 or 90 days after the occurrence of the earlier of the events described in 7.3(c)(vii);"

Except as specifically set forth in this Third Amendment, the Agreement shall remain in full force and effect.

[Signature page follows]

Table of Contents

IN WITNESS WHEREOF, the parties have executed this Third Amendment to be effective as of the date first above written.

GENERAL MOLY, INC.

/s/ BRUCE D HANSEN

By:

Name: Bruce D. Hansen
Title: Chief Executive Officer

AMER INTERNATIONAL GROUP CO., LTD.

By: /s/ WENYIN WANG

Name: Wenyin Wang Title: *Chairman*

[Signature Page to Amendment No. 3 to Investment and Securities Purchase Agreement]

Table of Contents

Co-holder (if any) sign above

REVOCABLE PROXY GENERAL MOLY, INC.

THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS

This proxy revokes all prior proxies with respect to the Special Meeting. Receipt of the Notice of Special Meeting and the Proxy Statement relating to the Special Meeting is hereby acknowledged.

The undersigned hereby appoints R. Scott Roswell and Michael K. Branstetter (collectively, the "Proxies"), and each of them, with full power of substitution, as proxies to vote all of the shares of Common Stock of General Moly, Inc. (the "Company") that the undersigned is entitled to vote at the Special Meeting of Stockholders of the Company to be held on December 15, 2017, and any adjournment thereof. Such shares shall be voted as indicated with respect to the proposals listed on this proxy and in the Proxies' discretion on such other matters as may properly come before the meeting or any adjournment thereof. Each of the proposed items below are described in the Proxy Statement that accompanies this revocable proxy, and the descriptions herein are qualified in their entirety by the information set forth in the Proxy Statement.

Proposals The Board of Directors recommends a vote FOR each of Proposals 1 and 2. 1. FOR **ABSTAIN** Approval of the issuance of shares of our common stock in connection with a AGAINST transaction in which the total shares issued represent more than 20% of our outstanding o o common stock, and may be issued at a discount to the greater of book or market value of our common stock. Approval of the ratification of equity awards granted since December 12, 2016. **FOR AGAINST ABSTAIN** o o o In their discretion, upon such other matters as may properly come before the meeting. 3. This proxy, when properly executed, will be voted in the manner directed herein by the undersigned stockholder. If this proxy is properly executed and returned, but no direction is made, this proxy will be voted by the Proxies FOR each of Proposals 1 and 2. Please sign exactly as name appears below. When shares are held by joint tenants, both should sign. When signing as attorney, as executor, administrator, trustee, or guardian, please give full title as such. If a corporation, please sign in full corporate name by President or other authorized officer. If a partnership, please sign in partnership name by authorized person. Please be sure to sign and date this proxy in the spaces below: Date: , 2017 Stockholder sign above Date: , 2017

Detach above card, sign, date and mail in postage paid envelope provided.

GENERAL MOLY, INC.

1726 Cole Blvd., Ste 115.

Lakewood, Colorado 80401

PLEASE MARK, SIGN, DATE AND RETURN THIS PROXY PROMPTLY USING THE ENCLOSED ENVELOPE

IF YOUR ADDRESS HAS CHANGED, PLEASE CORRECT THE ADDRESS IN THE SPACE PROVIDED BELOW AND RETURN THIS PORTION WITH THE PROXY IN THE ENVELOPE PROVIDED.

ont style="DISPLAY: inline; FONT-SIZE: 10pt; FONT-FAMILY: times new roman">

Weighted Average Common Shares Outstanding

139,444,940

126,290,753

10,000

The accompanying notes are an integral part of these financial statements

41

Optex Systems Holdings, Inc. (formerly known as Sustut Exploration, Inc.) Consolidated Statements of Cash Flows

			Success the period October through September	r 15\$epte	
Cash flows from operating activities:					
Net income (loss)	\$	(9,611,972)	\$ (89)	,487) \$	65,332
Adjustments to reconcile net loss to net cash used i operating activities:	n				
Depreciation and amortization		1,103,732	2,161	,486	9,691
(Gain) loss on impairment of intangible assets		8,038,431		-	-
Provision for allowance for inventory valuation		(129,152)	(146	,266)	27,363
Noncash interest expense		19,707	159	,780	9,500
Stock option compensation expense		97,311	39	,528	-
(Increase) decrease in accounts receivable		(572,854)	(397)	,996)	1,049,802
(Increase) decrease in inventory (net of progress billed)		2,253,247	(2,483)	.686)	(863,566)
(Increase) decrease in other current assets		86,352	•	,633	18,541
(Increase) decrease in deferred tax asset (net of				,	23,212
valuation allowance)		(282,319)	(711	,177)	_
Increase (decrease) in accounts payable and accrue	d	(===,==>)	(122	,,	
expenses		(1,831,205)	733.	,453	(186,051)
Increase (decrease) in accrued warranty costs		(56,530)		,470)	_
Increase (decrease) in due to parent		-		-	1,428
Increase (decrease) in accrued estimated loss on					
contracts		9,008	541	,479	(15,304)
Increase (decrease) in income taxes payable		, -		_	-
Total adjustments	\$	8,735,728	\$ (52)	,236) \$	51,404
Net cash (used)/provided by operating activities	\$	(876,244)		,723) \$	116,736
71 31 2					
Cash flows from investing activities:					
Cash received through Optex Texas acquisition	\$	-	\$ 253	,581 \$	-
Purchased of property and equipment		(115,703)		,824)	(13,338)
Net cash (used in) provided by investing activities	\$	(115,703)	\$ 239	,757 \$	(13,338)
Cash flows from financing activities:					
Private placement net of stock issuance cost		-	1,024	.529	-
Proceeds (to) from credit facility (net)		1,106,852	1,02	-	_
Proceeds from loans payable		250,000	(207	,265)	(20,000)
Repayments on loans payable		(250,000)	(207)	-	-
Net cash (used In) provided by financing activities	\$		\$ 817	,264 \$	(20,000)
Net increase (decrease) in cash and cash equivalent	s \$	114,905	\$ 915	,298 \$	83,398

Predecessor

Cash and cash equivalents at beginning of period	915,298	-	170,183
Cash and cash equivalents at end of period	\$ 1,030,203 \$	915,298 \$	253,581

The accompanying notes are an integral part of these financial statements

Optex Systems Holdings, Inc. (formerly known as Sustut Exploration, Inc.) Consolidated Statements of Cash Flows – continued

Predecessor
Successor For the period
SuccessoFor the period October \$5ptember 29, 2008
Twelve months end2008 through September 237ough October 14,
October 3, 2010 2009 2008

Noncash Investing and Financing Activities:					
Optex Delaware (Successor) Purchase of Optex Texas					
(Predecessor)					
Cash Received		-		253,581	-
Accounts Receivable		-		1,404,434	-
Inventory		-		5,383,929	-
Intangibles		-		4,036,790	-
Other Assets		-		632,864	-
Accounts Payable		-		(1,953,833)	-
Other Liabilities		-		(1,868,180)	-
Debt		-		(6,000,000)	-
Goodwill		-		7,110,415	-
Issuance of Stock	\$	-	\$	9,000,000 \$	-
Conversion of Debt to Series A Preferred Stock					
Additional Paid in Capital (6,000,000 Debt Retirement					
plus Accrued Interest of \$159,780)	\$	_	\$	6,159,780 \$	
plus Accided interest of \$139,780)	φ	-	Ф	0,139,780 \$	-
Issuance of Common shares in exchange for Investor					
Relations Services					
Prepaid Expenses (1,030,000 shares issued at \$0.001					
par)	\$	-	\$	226,500 \$	-
Lawrence of Warmente on Dalet Lawrence Coat					
Issuance of Warrants as Debt Issuance Cost	ф	22.000	Φ	¢	
Additional Paid in Capital (1,100,000 warrants)	\$	32,000	Þ	- \$	-
Supplemental cash flow information:					
Cash Paid for Interest	\$	69,631		10,290 \$	_
Cash Paid for Taxes	\$	119,847		488,799 \$	-

The accompanying notes are an integral part of these financial statements

Optex Systems Holdings, Inc. (formerly known as Sustut Exploration, Inc.) Consolidated Statement of Stockholders' Equity

	Common Shares Outstanding	Series A Preferred CommonPref Shares Stockeries	erradreasury Stock A Stoutex Texas	Additional Paid in Capital	Retained S Earnings	Total tockholders Equity
Predecessor Entity	Outstanding	Shares Stockeries	A Stoques Texas	Сарпа	Lamings	Equity
Balance at September 28,	10.000	¢ 164 024	¢ (1.217.400)	\$ 15 24C 202	Ф. <i>(5</i> .010.700). ф	0.202.016
2008	10,000	\$ 164,834	\$ (1,217,400)	\$ 15,246,282	\$ (5,910,700) \$	8,283,016
Net Income					65,332	65,332
Balance at October 14, 2008	10,000	-\$ 164,834 \$	-\$ (1,217,400)	\$ 15,246,282	\$ (5,845,368) \$	8,348,348
Successor Entity Balance at October 15, 2008						
Issuance of Common Stock (1)	113,333,282	-\$ 113,333 \$	_\$ _	\$ 8,886,667	\$\$	9,000,000
Cancellation of Investor Relations Stock	(700,000)) (700)		(104,300)		(105,000)
	(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	((- ,,-		(==,==,
Investor Relations Common Stock	490,000	400		142.520		144,000
Issued	480,000	480		143,520		144,000
Issuance of Common Stock	750,000	750		149,250		150,000
Conversion of 6,000,000 Debt and Interest to Series A preferred shares		1,027	1	6,159,780		6,159,781
				· ,		
Sustut Exploration Reorganization	17,449,991	17,450		170,050		187,500

Stock Option Compensation Expense		_	_	_	_	39,528		_	- 39,528	
Expense						37,320			37,320	
Private Placement Sale										
of Stock	8,131,667	_	8,132	_	_	1,012,647		_	- 1,020,779	
	, ,		,			, ,			, ,	
Accumulated Dividends on Preferred Stock						186,246		(186,246)	-	L
Net Earnings (Loss) from										
continuing								/aa /a=\	/aa .a=v	
operations	_	- —	_	_	_	-	_	(89,487)	(89,487)	
Balance at September 27,								,	* · · · • • · · · · · · · · · · · · · ·	
2009	139,444,940	1,027 \$13	39,445 \$	1 \$	- \$	16,643,388	\$	(275,733)	\$ 16,507,101	
Balance at September 27,	120 444 040	1.007	00 445 · Φ	1 0	Ф	16 642 200	ф	(075 7 22)	Φ 1.6 5 0 5 101	
2009	139,444,940	1,027 \$13	39,445 \$	1 \$	- \$	16,643,388	\$	(275,733)	\$ 16,507,101	
a 101										
Stock Option Compensation										
Expense		_	_	_	_	97,311		_	- 97,311	
Warrants Issued		- —	_	_	_	32,000		_	- 32,000	
Accumulated										
Dividends on										
Preferred Stock						389,551		(389,551)	-	H
Net Earnings										
(Loss) from continuing										
operations	_	- —	_	_	_	_	– (9	,611,972)	(9,611,972)	
Balance at										
October 3, 2010	139,444,940	1,027 \$ 13	39,445 \$	1 \$	\$	17,162,250	\$(10	,277,256)	\$ 7,024,440	

The accompanying notes are an integral part of these financial statements

⁽¹⁾After giving effect to the equivalent number of shares issued to existing Optex shareholders due to the reorganization.

Note 1 - Organization and Operations

On March 30, 2009, Optex Systems Holdings, Inc., (formerly known as Sustut Exploration, Inc.), a Delaware corporation, along with Optex Systems, Inc., a privately held Delaware corporation, which is a wholly-owned subsidiary of Optex Systems Holdings', also known as Successor, entered into a reorganization agreement and plan of reorganization, pursuant to which Optex Systems, Inc. (Delaware) was acquired by Optex Systems Holdings in a share exchange transaction. Optex Systems Holdings became the surviving corporation. At the closing, Optex Systems Holdings changed its name from Sustut Exploration Inc. to Optex Systems Holdings, Inc. and its year end from December 31 to a fiscal year ending on the Sunday nearest September 30.

On October 14, 2008, certain senior secured creditors of Irvine Sensors Corporation, Longview Fund, L.P. and Alpha Capital Anstalt, formed Optex Systems, Inc. (Delaware), which acquired all of the assets and assumed certain liabilities of Optex Systems, Inc., a Texas corporation and wholly-owned subsidiary of Irvine Sensors Corporation, also known as Predecessor, in a transaction that was consummated via purchase at a public auction. Following this asset purchase, Optex Systems, Inc. (Texas) remained a wholly-owned subsidiary of Irvine Sensors Corporation.

In accordance with FASB ASC 805 (Prior authoritative literature: SFAS No. 141(R), "Business Combinations" and EITF 98-3 "Determining Whether a Non-monetary Transaction Involves Receipt of Productive Assets or of a Business") Optex Systems, Inc. (Delaware)'s purchase of substantially all of the assets and assumption of certain liabilities represented the acquisition of a business. FASB ASC 805 outlines the guidance in determining whether a "business" has been acquired in a transaction. For a transferred set of activities and assets to be a business, it must contain all of the inputs and processes necessary for it to continue to conduct normal operations after the transferred set of assets is separated from the transferor, which include the ability to sustain a revenue stream by providing its outputs to customers. Optex Systems, Inc. (Delaware) obtained the inputs and processes necessary for normal operations.

Optex Systems, Inc. (Texas) was a privately held Subchapter "S" Corporation from inception in 1987 until December 30, 2005 when 70% of the issued and outstanding stock was acquired by Irvine Sensors Corporation, and Optex Systems, Inc. (Texas) was automatically converted to a Subchapter "C" Corporation. On December 29, 2006, the remaining 30% equity interest in Optex Systems, Inc. (Texas) was purchased by Irvine Sensors Corporation.

On February 20, 2009, Sileas Corporation., a newly-formed Delaware corporation, owned by present members of Optex Systems Holdings' management, purchased 100% of Longview's equity and debt interest in Optex Systems, Inc. (Delaware), representing 90% of the issued and outstanding common equity interests in Optex Systems, Inc. (Delaware), in a private transaction. See Note 4.

Optex Systems, Inc. (Delaware) operated as a privately-held Delaware corporation until March 30, 2009, when as a result of the reorganization agreement (described above and also in Note 5), it became a wholly-owned subsidiary of Optex Systems Holdings. Sileas is the majority owner (parent) of Optex Systems Holdings owning 73.52% of Optex Systems Holdings. Optex Systems Holdings plans to carry on the business of Optex Systems, Inc. (Delaware) as its sole line of business and all of Optex Systems Holdings' operations are conducted by and through its wholly-owned subsidiary, Optex Systems, Inc. (Delaware). Accordingly, in subsequent periods the financial statements presented will be those of the accounting acquirer. The financial statements of Optex Systems Holdings represent subsidiary statements and do not include the accounts of its majority owner.

Optex Systems Holdings' operations are based in Richardson, Texas in a leased facility comprising 49,100 square feet. As of December 13, 2010, Optex Systems Holdings operated with 89 full-time equivalent employees.

Optex Systems Holdings manufactures optical sighting systems and assemblies, primarily for Department of Defense applications. Its products are installed on a variety of U.S. military land vehicles such as the Abrams and Bradley

fighting vehicles, light armored and advanced security vehicles and have been selected for installation on the Stryker family of vehicles. Optex Systems Holdings also manufactures and delivers numerous periscope configurations, rifle and surveillance sights and night vision optical assemblies. Optex Systems Holdings' products consist primarily of build to customer print products that are delivered both directly to the military and to other defense prime contractors.

In February 2009, Optex Systems Holdings' ISO certification status was upgraded from 9001:2000 to 9001:2008 bringing Optex Systems Holdings into compliance with the new ISO standards rewritten to align with ISO 14001.

Note 2 - Accounting Policies

Basis of Presentation

Principles of Consolidation: The consolidated financial statements include the accounts of Optex Systems Holdings and its wholly-owned subsidiary, Optex Systems, Inc. (Delaware). All significant inter-company balances and transactions have been eliminated in consolidation.

The accompanying financial statements include the results of operations, changes in stockholders equity and statements of cash flows, for the periods from October 15, 2008 through September 27, 2009 and September 28, 2009 through October 3, 2010 and the balance sheets at October 3, 2010 and September 27, 2009 of Optex Systems, Inc. (Delaware), the accounting acquirer in the Sustut reorganization and the Successor in the October 14, 2008 Optex Systems, Inc. (Texas) asset purchase transaction. The accompanying financial statements also include the results of operations, changes in stockholders' equity and cash flows for the period from September 29, 2008 through October 14, 2008 of Optex Systems, Inc. (Texas), Predecessor.

45

Although, Optex Systems, Inc. (Texas) (Predecessor) has been majority owned by various parent companies described in the preceding paragraphs, no accounts of the parent companies or the effects of consolidation with any parent companies have been included in the accompanying financial statements. The Optex Systems, Inc. (Texas) accounts have been presented on the basis of push down accounting in accordance with FASB ASC 805-50-S99 (Prior authoritative literature: Staff Accounting Bulletin No. 54 Application of "Push Down" Basis of Accounting in Financial Statements of Subsidiaries Acquired by Purchase). FASB ASC 805-50-S99 states that the push down basis of accounting should be used in a purchase transaction in which the entity becomes wholly-owned. Under the push down basis of accounting certain transactions incurred by the parent company, which would otherwise be accounted for in the accounts of the parent, are "pushed down" and recorded on the financial statements of the subsidiary. Accordingly, items resulting from the Optex Systems, Inc. (Texas) purchase transaction such as goodwill, debt incurred by the parent to acquire the subsidiary and other costs related to the purchase have been recorded on the financial statements of Optex Systems Holdings.

Upon completing the business combination with Sustut on March 30, 2009, Optex Systems Holdings elected to change its fiscal year to match that of Optex Systems, Inc. (Delaware). Accordingly, all activity of the combined companies was presented as of the quarter's end of the accounting acquirer, which was March 29, 2009.

Although the effective date of the merger was March 30, 2009, all transactions related to the business combination (and only those transactions), with Sustut have been reflected as if they had taken place one day prior (on March 29, 2009) so as to coincide with the accounting acquirer's quarter end of March 29, 2009. See Note 5 for details of the reorganization.

Use of Estimates: The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statement and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from the estimates.

Segment Reporting: Management has determined that Optex Systems Holdings, Inc. is organized, managed and internally reported as one business segment. Segments are determined based on differences in products, internal reporting and how operational decisions are made.

Fiscal Year: Optex's fiscal year ends on the Sunday nearest September 30. Fiscal year 2009 ended on October 3, 2010 and included 53 weeks. Fiscal year 2009 ended on September 27, 2009 and included 52 weeks.

Fair Value of Financial Instruments: FASB ASC 825-10 (Prior authoritative literature: FASB No. 107, "Disclosures about Fair Value of Financial Instruments)," requires disclosure of fair value information about certain financial instruments, including, but not limited to, cash and cash equivalents, accounts receivable, refundable tax credits, prepaid expenses, accounts payable, accrued expenses, notes payable to related parties and convertible debt-related securities. Fair value estimates discussed herein are based upon certain market assumptions and pertinent information available to management as of fiscal years ended October 3, 2010 and September 28, 2008. The carrying value of the balance sheet financial instruments included in Optex Systems, Inc. (Texas)'s consolidated financial statements approximated their fair values.

Cash and Cash Equivalents: For financial statement presentation purposes, Optex considers those short-term, highly liquid investments with original maturities of three months or less to be cash or cash equivalents.

Concentration of Credit Risk: Optex's cash and cash equivalents are on deposit with banks. Only a portion of the cash and cash equivalents would be covered by deposit insurance and the uninsured balances are substantially greater than the insured amounts. Although cash and cash equivalent balances exceed insured deposit amounts, management does

not anticipate non-performance by the banks.

Optex revenues and accounts receivables for fiscal year ended October 3, 2010 are derived from sales to U.S. government agencies (39%), General Dynamics (51%) or other prime government contractors (10%). Optex does not believe that this concentration results in undue credit risk because of the financial strength of the payees.

Accounts Receivable: Optex records its accounts receivable at the original sales invoice amount less liquidations for previously collected advance/progress bills and an allowance for doubtful accounts. An account receivable is considered to be past due if any portion of the receivable balance is outstanding beyond its scheduled due date. On a quarterly basis, Optex evaluates its accounts receivable and establishes an allowance for doubtful accounts, based on its history of past write-offs and collections, and current credit conditions. No interest is accrued on past due accounts receivable. As the customer base is primarily U.S. government and government prime contractors, Optex has concluded that there is no need for an allowance for doubtful accounts for the years ended October 3, 2010 and September 27, 2009. Optex charges uncollectible accounts to bad debt expense in the period as they are first deemed uncollectible. In 2010, there was no bad debt expense associated with uncollectable accounts. In 2009, Optex Systems Holdings recorded \$35,297 in bad debt expense attributable to one customer that went out of business.

46

Inventory: Inventory is recorded at the lower of cost or market, and adjusted as appropriate for decreases in valuation and obsolescence. Adjustments to the valuation and obsolescence reserves are made after analyzing market conditions, current and projected sales activity, inventory costs and inventory balances to determine appropriate reserve levels. Cost is determined using the first-in first-out method. Under arrangements by which progress payments are received against certain contracts, the customer retains a security interest in the undelivered inventory identified with these contracts. Payments received for such undelivered inventory are classified as unliquidated progress payments and deducted from the gross inventory balance. As of October 3, 2010, and September 27, 2009 inventory included:

	As of		As of	
	October 3, 2010		Sep	otember 27, 2009
Raw Materials	\$	4,343,168	\$	7,161,241
Work in Process		2,823,501		4,043,308
Finished Goods		366,110		245,056
Gross Inventory	\$	7,532,779	\$	11,449,605
Less:				
Unliquidated Progress Payments		(1,217,319)		(2,880,898)
Inventory Reserves		(425,674)		(554,826)
Net Inventory	\$	5,889,786	\$	8,013,881

Optex Systems Holdings conducts an annual physical inventory in the fourth quarter of each fiscal year. The accounting records are adjusted to reflect any changes in the physical inventory valuation as compared to the book carrying values based on the results of the physical inventory. In 2010, Optex Systems recognized a loss of (\$919,470) as compared to a net gain in the period ending September 27, 2009 of \$166,179. The inventory loss was primarily attributable to higher than expected manufacturing costs across both our Periscope and Howitzer production lines. The loss is a result of lower production volume in the second half of fiscal year 2010 which impacted material pricing, scrap, and labor efficiencies resulting in cost overruns for these products carried in inventory as compared to the estimates at completion.

Warranty Costs: Some of Optex Systems Holdings' customers require that the company warrant the quality of its products to meet customer requirements and be free of defects for up to fifteen months subsequent to delivery. In the years ended October 3, 2010 and September 27, 2009, Optex Systems Holdings, Inc. recognized income of \$56,530, and \$145,470, respectively, related to improvements in the warranty experience rate for warranties expiring in each of the respective years. Future warranty costs are based on the estimated cost of replacement for expected returns based upon our most recent experience rate of defects as a percentage of warranty covered sales.

Property and Equipment: Property and equipment are recorded at cost. Depreciation is computed using the straight line method over the estimated useful lives of the assets, ranging from three to seven years. Expenditures for renewals and betterments are capitalized. Expenditures for minor items, repairs and maintenance are charged to operations as incurred. Gain or loss upon sale or retirement due to obsolescence is reflected in the operating results in the period the event takes place.

Goodwill and Other Intangible Assets: Goodwill represents the cost of acquired businesses in excess of fair value of the related net assets at acquisition. (See also notes 4 and 11). Optex Systems Holdings does not amortize goodwill, but tests it annually for impairment using a fair value approach during the fiscal fourth quarter and between annual testing periods, if circumstances warrant. The performance of the test involves a two-step process. The first step of the impairment test involves comparing the fair values of the applicable reporting units with their aggregate carrying values, including goodwill. We generally determine the fair value of our reporting units using the income approach methodology of valuation that includes the discounted cash flow method as well as other generally accepted valuation

methodologies, which requires significant judgment by management. If the carrying amount of a reporting unit exceeds the reporting unit's fair value, we perform the second step of the goodwill impairment test to determine the amount of impairment loss. The second step of the goodwill impairment test involves comparing the implied fair value of the affected reporting unit's goodwill with the carrying value of that goodwill. These impairment tests may result in impairment charges that could have a material adverse impact on our results of operations. The goodwill of Optex Systems Holdings was reviewed as of October 3, 2010. The review indicated that goodwill was impaired, as determined based on the projected cash flows over the next three years. The cash flow projections took into effect the expected net sales and corresponding expenses against those sales in the respective years. The impairment loss for goodwill was \$7,110,415. The goodwill was written off as a component of general and administrative operating expenses during fiscal year 2010.

Optex amortizes the cost of other intangibles over their estimated useful lives, unless such lives are deemed indefinite. Amortizable intangible assets are tested for impairment based on undiscounted cash flows and, if impaired, written down to fair value based on either discounted cash flows or appraised values. The identified amortizable intangible assets at October 3, 2010 related to the acquisition of Optex Systems, Inc. (Delaware) from Irvine Sensors as of October 14, 2008 and consisted of customer backlog, with initial useful lives ranging from one to five years. (See note 4 and 11). The identified amortizable intangible assets at September 28, 2008 related to the acquisition of Optex Systems, Inc. (Texas) by Irvine Sensors and consisted of non-competition agreements and customer backlog, with initial useful lives ranging from two to eight years. (See note 4 and 11).

Intangible assets with indefinite lives are tested annually for impairment, during the fiscal fourth quarter and between annual periods, if impairment indicators exist, and are written down to fair value as required. As of October 3, 2010, the intangible assets were reviewed in light of a reduction of expected delivery orders against contracted orders and higher than expected costs on those orders. The review indicated that intangible assets were impaired, as determined based on a projected cash flow analysis of our future operations. The cash flow projections took into effect the expected net sales from the customer backlog as of October 14, 2008 and the corresponding expenses against those sales in the respective years. The impairment loss recorded in 2010 for intangible assets was \$928,016 and was split between cost of goods sold and general and administrative costs in the amount of \$150,534, and \$777,482, respectively. As of October 3, 2010, after impairment, the total balance of unamortized intangible assets and goodwill was zero.

Impairment or Disposal of Long-Lived Assets: Optex Systems Holdings adopted the provisions of FASB ASC 360-10 (Prior authoritative literature FASB No. 144, "Accounting for the Impairment or Disposal of Long-lived Assets .") This standard requires, among other things, that long-lived assets be reviewed for potential impairment whenever events or circumstances indicate that the carrying amounts may not be recoverable. The assessment of possible impairment is based on the ability to recover the carrying value of the asset from the expected future pre-tax cash flows (undiscounted and without interest charges) of the related operations. If these expected cash flows are less than the carrying value of such asset, an impairment loss is recognized for the difference between estimated fair value and carrying value. The primary measure of fair value is based on discounted cash flows. The measurement of impairment requires management to make estimates of these cash flows related to long-lived assets, as well as other fair value determinations.

Revenue Recognition:

Optex Systems Holdings recognizes revenue based on the modified percentage of completion method utilizing the units-of-delivery method, in accordance with FASB ASC 605-35 (Prior authoritative literature: SOP 81-1 "Accounting for Performance of Construction—Type and Certain Production—Type Contracts"):

The units-of-delivery method recognizes as revenue the contract price of units of a basic production product delivered during a period and as the cost of earned revenue the costs allocable to the delivered units; costs allocable to undelivered units are reported in the balance sheet as inventory or work in progress. The method is used in circumstances in which an entity produces units of a basic product under production-type contracts in a continuous or sequential production process to buyers' specifications.

Optex Systems Holdings contracts are fixed price production type contracts whereby a defined order quantity is delivered to the customer during a continuous or sequential production process tailored to the buyer's specifications (build to print). Optex Systems Holdings' deliveries against these contracts generally occur in monthly increments across fixed delivery periods spanning from 3 to 36 months.

Estimated Costs at Completion and Accrued Loss on Contracts: Optex Systems Holdings reviews and reports on the performance of its contracts and production orders against the respective resource plans for such contracts/orders.

These reviews are summarized in the form of estimates at completion. Estimates at completion include Optex Systems Holdings' incurred costs to date against the contract/order plus management's current estimates of remaining amounts for direct labor, material, other direct costs and subcontract support and indirect overhead costs based on the completion status and future contractual requirements for each order. If an estimate at completion indicates a potential overrun (loss) against a fixed price contract/order, management generally seeks to reduce costs and /or revise the program plan in a manner consistent with customer objectives in order to eliminate or minimize any overrun and to secure necessary customer agreement to proposed revisions.

If an estimate at completion indicates a potential overrun against budgeted resources for a fixed price contract/order, management first attempts to implement lower cost solutions to still profitably meet the requirements of the fixed price contract. If such solutions do not appear practicable, management makes a determination whether to seek renegotiation of contract or order requirements from the customer. If neither cost reduction nor renegotiation appears probable, an accrual for the contract loss/overrun is recorded against earnings and the loss is recognized in the first period the loss is identified based on the most recent estimates at completion of the particular contract or product order.

For the fiscal years ended October 3, 2010 and September 27, 2009, estimated loss reserves were \$1,357,068 and \$1,348,060, respectively. During 2010, Optex Systems Holdings realized increases losses against the Howitzer programs of \$1,139,659 of which \$762,864 relates specifically to production issues encountered on one of our Howitzer product lines. Contract losses attributable to program deliveries during the fiscal year 2010 were \$1,130,651 for a net increase of \$9,008 in the ending reserve balance. Increased losses were primarily attributable to manufacturing issues on our U.S. government Howitzers culminating in higher material scrap and labor hours, combined with a reduction in total production volume in 2010 which further impacted production efficiencies across all product lines. Optex Systems Holdings has requested an equitable adjustment on this program due to significant design issues impacting the manufacturability of the product. As there is no guarantee that the request will be granted in part or in full, we realized the entire loss in fiscal year 2010. However, we believe there is a reasonable possibility that we will be able to recover a substantial amount of the incurred loss in fiscal year 2011 pending the outcome of the negotiations.

Government Contracts: Virtually all of Optex Systems Holdings' contracts are prime or subcontracted directly with the Federal government and as such, are subject to Federal Acquisition Regulation (Federal Acquisition Regulation) Subpart 49.5, "Contract Termination Clauses" and more specifically Federal Acquisition Regulation clauses 52.249-2 "Termination for Convenience of the Government (Fixed-Price)", and 49.504 "Termination of fixed-price contracts for default". These clauses are standard clauses on prime military contracts and are generally, "flowed down" to Optex Systems Holdings as subcontractors on other military business. It has been Optex Systems Holdings' experience that the termination for convenience is rarely invoked, except where it has been mutually beneficial for both parties. Optex Systems Holdings is not currently aware of any pending terminations for convenience or default on its existing contracts.

In the event a termination for convenience were to occur, these Federal Acquisition Regulation clause 52.249-2 provides for full recovery of all contractual costs and profits reasonably occurred up to and as a result of the terminated contract. In the event a termination for default were to occur, Optex Systems Holdings could be liable for any excess cost incurred by the government to acquire supplies from another supplier similar to those terminated from Optex Systems Holdings. Optex Systems Holdings would not be liable for any excess costs if the failure to perform the contract arises from causes beyond the control and without the fault or negligence of the company as defined by Federal Acquisition Regulation clause 52.249-8. In addition, the government may require Optex Systems Holdings to transfer title and deliver to the government any completed supplies, partially completed supplies and materials, parts, tools, dies, jigs, fixtures, plans, drawings, information, and contract rights that Optex Systems Holdings has specifically produced or acquired for the terminated portion of this contract. The government shall pay contract price for completed supplies delivered and accepted, and Optex Systems Holdings and the government would negotiate an agreed upon amount of payment for manufacturing materials delivered and accepted and for the protection and preservation of the property. Failure to agree on an amount for manufacturing materials is subject to the Federal Acquisition Regulation Disputes clause 52.233-1.

In some cases, Optex Systems Holdings may receive orders subject to subsequent price negotiation on contracts exceeding the federal government simplified acquisition threshold of \$650,000 prior to October 1, 2010 and \$700,000 subsequent to October 1, 2010. These "undefinitized" contracts are considered firm contracts but as Cost Accounting

Standards Board covered contracts, they are subject to the Truth in Negotiations Act disclosure requirements and downward only price negotiation. As of October 3, 2010 and September 27, 2009, Optex Systems had no booked orders that fell under this criteria. Optex Systems Holdings' experience has been that the historically negotiated price differentials have been immaterial and accordingly, it does not anticipate any significant downward adjustments on these booked orders.

Shipping and Handling Costs: All shipping and handling costs are included as a component of cost of goods sold.

Stock-Based Compensation: In December 2004, FASB issued FASB ASC 718 (Prior authoritative literature: SFAS No. 123R, "Share-Based Payment"). FASB ASC 718 establishes standards for the accounting for transactions in which an entity exchanges its equity instruments for goods or services. It also addresses transactions in which an entity incurs liabilities in exchange for goods or services that are based on the fair value of the entity's equity instruments or that may be settled by the issuance of those equity instruments. FASB ASC 718 focuses primarily on accounting for transactions in which an entity obtains employee services in share-based payment transactions. FASB ASC 718 requires that the compensation cost relating to share-based payment transactions be recognized in the financial statements. That cost will be measured based on the fair value of the equity or liability instruments issued.

Optex Systems Holdings' accounting policy for equity instruments issued to consultants and vendors in exchange for goods and services follows the provisions of FASB ASC 505-50 (Prior authoritative literature: EITF 96-18, "Accounting for Equity Instruments That are Issued to Other Than Employees for Acquiring, or in Conjunction with Selling, Goods or Services" and EITF 00-18, "Accounting Recognition for Certain Transactions Involving Equity Instruments Granted to Other Than Employees"). The measurement date for the fair value of the equity instruments issued is determined at the earlier of (i) the date at which a commitment for performance by the consultant or vendor is reached or (ii) the date at which the consultant or vendor's performance is complete. In the case of equity instruments issued to consultants, the fair value of the equity instrument is recognized over the term of the consulting agreement. Stock-based compensation related to non-employees is accounted for based on the fair value of the related stock or options or the fair value of the services, whichever is more readily determinable in accordance with FASB ASC 718.

Income Tax/Deferred Tax: FASB ASC 740 (Prior Authoritative Literature: SFAS No. 109, "Accounting for Income Taxes"), requires recognition of deferred tax assets and liabilities for the expected future tax consequences of events that have been included in the financial statements or tax returns. Under this method, deferred tax assets and liabilities are determined based on differing treatment of items for financial reporting and income tax reporting purposes. The deferred tax balances are adjusted to reflect tax rates by tax jurisdiction, based on currently enacted tax laws, which will be in effect in the years in which the temporary differences are expected to reverse. Optex Systems Holdings has recognized deferred income tax benefits on net operating loss carry-forwards to the extent Optex Systems Holdings believes it will be able to utilize them in future tax filings.

Earnings per Share: Basic earnings per share is computed by dividing income available to common shareholders (the numerator) by the weighted-average number of common shares outstanding (the denominator) for the period. Diluted earnings per common share gives effect to the assumed exercise of stock options when dilutive. Diluted earnings per share is computed by assuming that any dilutive convertible securities outstanding were converted, with related preferred stock dividend requirements and outstanding common shares adjusted accordingly. It is also assumes that outstanding common shares were increased by shares issuable upon exercise of those stock options for which market price exceeds the exercise price, less shares which could have been purchased by us with the related proceeds. In period of losses, diluted loss per share is computed on the same basis as basic loss per share as the inclusion of any other potential shares outstanding would be anti-dilutive.

If Optex Systems Holdings had recorded income applicable to common shareholders for the period September 27, 2009 through October 3, 2010, weighted average number of common shares outstanding would have increased by 43,288,096 and for the period October 15, 2008 through September 27, 2009, weighted average number of common shares outstanding would have increased by 42,570,745, reflecting the addition of dilutive securities in the calculation of diluted earnings per share.

Note 3 - Recent Accounting Pronouncements

In June 2008, FASB issued FASB ASC 260-10-55 (Prior authoritative literature: FASB Staff Position EITF 03-6-1, "Determining Whether Instruments Granted in Share-Based Payment Transactions are Participating Securities"). FASB ASC 260-10-55 clarifies that share-based payment awards that entitle their holders to receive nonforfeitable dividends or dividend equivalents before vesting should be considered participating securities. As participating securities, we will be required to include these instruments in the calculation of our basic earnings per share, and we will need to calculate basic earnings per share using the "two-class method." Restricted stock is currently included in our dilutive earnings per share calculation using the treasury stock method. The two-class method of computing earnings per share is an earnings allocation formula that determines earnings per share for each class of common stock and participating security according to dividends declared (or accumulated) and participation rights in undistributed earnings. FASB ASC 260-10-55 is effective for financial statements issued for fiscal years beginning after December 15, 2008, and all

interim periods within those fiscal years. As such, Optex Systems Holdings adopted these provisions at the beginning of the fiscal year ending October 3, 2010. Adoption of FASB ASC 260-10-55 did not have a material effect on Optex Systems Holdings' financial statements.

In May 2009, FASB issued FASB ASC 855-10 (Prior authoritative literature: SFAS No. 165, "Subsequent Events"). FASB ASC 855-10 establishes principles and requirements for the reporting of events or transactions that occur after the balance sheet date, but before financial statements are issued or are available to be issued. FASB ASC 855-10 is effective for financial statements issued for fiscal years and interim periods ending after June 15, 2009. As such, Optex Systems Holdings adopted these provisions at the beginning of the interim period ended June 28, 2009. Adoption of FASB ASC 855-10 did not have a material effect on Optex Systems Holdings' financial statements.

In February 2010, FASB issued ASU 2010-09 "Subsequent Event (Topic 855) Amendments to Certain Recognition and Disclosure Requirements". ASU 2010-09 removes the requirement for an SEC filer to disclose a date in both issued and revised financial statements. Revised financial statements include financial statements revised as a result of either correction of an error or retrospective application of GAAP. All of the amendments in ASU 2010-09 are effective upon issuance of the final ASU, except for the use of the issued date for conduit debt obligors, which is effective for interim or annual periods ending after June 15, 2010. The Company adopted ASU 2010-09 in February 2010 and therefore omitted the disclosure previously required as referenced above.

In June 2009, FASB issued ASC 105-10 (Prior authoritative literature: SFAS No. 168, "The FASB Accounting Standards Codification TM and the Hierarchy of Generally Accepted Accounting Principles - a replacement of FASB Statement No. 162"). FASB ASC 105-10 establishes the FASB Accounting Standards Codification TM (Codification) as the source of authoritative accounting principles recognized by the FASB to be applied by nongovernmental entities in the preparation of financial statements in conformity with GAAP. FASB ASC 105-10 is effective for financial statements issued for fiscal years and interim periods ending after September 15, 2009. As such, Optex Systems Holdings adopted these provisions at the beginning of the interim period ending October 3, 2010. Adoption of FASB ASC 105-10 did not have a material effect on Optex Systems Holding's financial statements.

In September 2006, the FASB issued FASB ASC 820-10 (Prior authoritative literature: FASB Statement 157, "Fair Value Measurements"). FASB ASC 820-10 defines fair value, establishes a framework for measuring fair value under GAAP and expands disclosures about fair value measurements. FASB ASC 820-10 applies under other accounting pronouncements that require or permit fair value measurements. Accordingly, FASB ASC 820-10 does not require any new fair value measurements. However, for some entities, the application of FASB ASC 820-10 will change current practice. The changes to current practice resulting from the application of FASB ASC 820-10 relate to the definition of fair value, the methods used to measure fair value and the expanded disclosures about fair value measurements. The provisions of FASB ASC 820-10 are effective as of January 1, 2008, with the cumulative effect of the change in accounting principle recorded as an adjustment to opening retained earnings. However, delayed application of this statement is permitted for nonfinancial assets and nonfinancial liabilities, except for items that are recognized or disclosed at fair value in the financial statements on a recurring basis (at least annually), until fiscal years beginning after November 15, 2008, and interim periods within those fiscal years. The adoption of FASB ASC 820-10 did not have a material impact on Optex Systems Holdings' financial position, results of operations, or cash flows.

In December 2007, FASB issued FASB ASC 805 (Prior authoritative literature: SFAS No. 141(R), "Business Combinations") and FASB ASC 810-10-65 (Prior authoritative literature: SFAS No. 160, "Accounting and Reporting of Noncontrolling Interest in Consolidated Financial Statements, an amendment of ARB No. 51"). These new standards will significantly change the accounting for and reporting of business combinations and non-controlling (minority) interests in consolidated financial statements. FASB ASC 805 and FASB ASC 810-10-65 are required to be adopted simultaneously and are effective for the first annual reporting period beginning on or after December 15, 2008. Earlier adoption is prohibited. As such, Optex Systems Holdings adopted these provisions at the beginning of the annual reporting period beginning September 28, 2009. Adoption of FASB ASC 805 and FASB ASC 810-10-65 did not have a material effect on Optex Systems Holding's financial statements.

In March 2008, FASB issued FASB ASC 815-10 (Prior authoritative literature: SFAS No. 161, "Disclosures about Derivative Instruments and Hedging Activities—an amendment of FASB Statement No. 133"). FASB ASC 815-10 requires enhanced disclosures about an entity's derivative and hedging activities. FASB ASC 815-10 is effective for financial statements issued for fiscal years and interim periods beginning after November 15, 2008 with early application encouraged. As such, Optex Systems Holdings adopted these provisions at the beginning of the fiscal year ended October 3, 2010. The adoption of FASB ASC 815-10 did not have a material impact Optex Systems Holdings' financial position, results of operations, or cash flows.

In May 2008, FASB issued FASB ASC 944 (Prior authoritative literature: SFAS No. 163, "Accounting for Financial Guarantee Insurance Contracts—an interpretation of FASB Statement No. 60"). FASB ASC 944 interprets Statement 60 and amends existing accounting pronouncements to clarify their application to the financial guarantee insurance contracts included within the scope of that Statement. FASB ASC 944 is effective for financial statements issued for fiscal years beginning after December 15, 2008, and all interim periods within those fiscal years. As such, Optex Systems Holdings adopted these provisions at the beginning of the fiscal year ended October 3, 2010. The adoption of FASB ASC 944 did not have a material impact Optex Systems Holdings' financial position, results of operations, or cash flows.

Note 4 — Acquisition of Substantially All of the Assets of Optex Systems, Inc. (Texas)

Acquisition of Assets of Optex Systems, Inc. (Texas) by Optex Systems, Inc. (Delaware) on October 14, 2008

On October 14, 2008, in a purchase transaction that was consummated via public auction, Optex Systems, Inc. (Delaware) (Successor) purchased all of the assets of Optex Systems, Inc. (Texas) (Predecessor) in exchange for \$15 million of Irvine Sensors Corporation debt owned by it and the assumption of approximately \$3.8 million of certain Optex Systems, Inc. (Texas) liabilities. The \$15 million of Irvine Sensors Corporation debt was contributed by Longview and Alpha to Optex Systems, Inc. (Delaware), in exchange for a \$6 million note payable from Optex Systems, Inc. (Delaware) and a \$9 million equity interest in Optex Systems, Inc. (Delaware) (which consisted of the issuance by Optex Systems, Inc. (Delaware) of 45,081,350 and 4,918,650 shares of its common stock to each of Longview Fund and Alpha, respectively). On October 30, 2008, Alpha sold its Optex Systems, Inc. (Delaware) common stock to Arland Holdings, Ltd. There was no contingent consideration associated with the purchase. Longview and Arland Holdings, Ltd. owned Optex Systems, Inc. (Delaware) together until February 20, 2009, when Longview sold 100% of its equity interests in Optex Systems, Inc. (Delaware) to Sileas, as discussed below.

Optex Systems, Inc. (Delaware) purchased all of the assets of Optex Systems, Inc. (Texas), including: intellectual property, production processes and know-how, and outstanding contracts and customer relationships. Optex Systems, Inc. (Delaware) also assumed certain liabilities of Optex Systems, Inc. (Texas) consisting of accounts payable and accrued liabilities. Optex Systems Holdings' management intends to improve the business's ability to serve its existing customers and to attract new customers by providing quality products and superior service which will be achieved by improving Optex Systems Holdings' working capital availability as opposed to the limited working capital that was available during the time period in which the assets were owned by Irvine Sensors Corporation.

Pro forma revenue and earnings per share information is presented cumulatively in Note 5.

Secured Promissory Note Issued in Connection with Purchase by Optex Systems, Inc. (Delaware) (Successor)

In connection with the public sale of the Optex Systems, Inc. (Texas) (Predecessor) assets to Optex Systems, Inc. (Delaware) (Successor), Optex Systems, Inc. (Delaware) delivered to Longview and Alpha Secured Promissory Notes, due September 19, 2011, in the principal amounts of \$5,409,762 and \$540,976, respectively. On February 20, 2009, Longview sold its Optex Systems, Inc. (Delaware) promissory note to Sileas, as described below. On March 27, 2009, Sileas and Alpha exchanged their Notes plus accrued and unpaid interest of \$159,780 for 1,027 shares of Optex Systems, Inc. (Delaware) Series A preferred stock.

Acquisition by Sileas on February 20, 2009

On February 20, 2009, Sileas purchased 100% of the equity and debt interest held by Longview, representing 90% of Optex Systems, Inc. (Delaware). Currently, Sileas is the majority owner of Optex Systems Holdings.

Secured Promissory Note Due February 20, 2012/Longview Fund, LP

As a result of the transaction described above between Sileas and Longview Fund, LP on February 20, 2009, Sileas, currently majority owner of Optex Systems Holdings executed and delivered to Longview, a Secured Promissory Note due February 20, 2012 in the principal amount of \$13,524,405. The Note bears simple interest at the rate of 4% per annum, and the interest rate upon an event of default increases to 10% per annum. In the event Optex Systems Holdings sells or conveys all or substantially all its assets to a third party entity for more than nominal consideration, other than a reorganization into Sileas or reincorporation in another jurisdiction, then this Note shall be immediately due and owing without demand. In the event that such a major transaction occurs prior to the maturity date resulting in

the Sileas receiving net consideration with a fair market value in excess of the principal and interest due under the terms of the secured note (the "Optex Consideration"), then in addition to paying the principal and interest due, Sileas shall also pay an amount equal to 90% of the Optex Consideration. The obligations of Sileas under the note are secured by a security interest in Optex Systems Holdings' common and preferred stock owned by Sileas that was granted to Longview pursuant to a Stock Pledge Agreement delivered by Sileas to Longview and also by a lien on all of the assets of Sileas.

Optex Systems Holdings has not guaranteed the note and Longview is not entitled to pursue Optex Systems Holdings in the event of a default by Sileas. Therefore, there are no actual or potential cash flow commitments from Optex Systems Holdings. In the event of default by Sileas on its obligations under the note, Longview would only be entitled to receive the Optex Systems Holdings common and preferred stock held by Sileas.

Note 5 – Reorganization Plan and Private Placement

Reorganization/Share Exchange

On March 30, 2009, a reorganization occurred whereby the then existing shareholders of Optex Systems, Inc. (Delaware) exchanged their shares of common stock with the shares of common stock of Optex Systems Holdings as follows: (i) the outstanding 85,000,000 shares of Optex Systems, Inc. (Delaware) common stock were exchanged by Optex Systems Holdings for 113,333,282 shares of Optex Systems Holdings common stock, (ii) the outstanding 1,027 shares of Optex Systems, Inc. (Delaware) Series A preferred stock were exchanged by Optex Systems Holdings for 1,027 shares of Optex Systems Holdings Series A preferred stock and (iii) the 8,131,667 shares of Optex Systems, Inc. (Delaware) common stock purchased in the private placement were exchanged by Optex Systems Holdings for 8,131,667 shares of Optex Systems Holdings common stock. Following the reorganization, Optex Systems, Inc. (Delaware) remained a wholly-owned subsidiary of Optex Systems Holdings.

Shares outstanding of Optex Systems Holdings just prior to the closing of the reorganization consisted of 17,449,991 shares which included 1,250,000 shares issued on March 27, 2009 as payment for Investor Relations Services. On June 29, 2009, 700,000 of the issued investor relations shares were surrendered to Optex Systems Holdings and cancelled upon termination of one of the Investor Relations contracts.

Private Placement

Prior to the closing of the reorganization agreement, as of March 30, 2009, Optex Systems, Inc. (Delaware) accepted subscriptions from accredited investors for a total of 27.1 units, for \$45,000 per unit, with each unit consisting of 300,000 shares of common stock, of Optex Systems, Inc. (Delaware) and warrants to purchase 300,000 shares of common stock for \$0.45 per share for a period of five years from the initial closing, which were issued by Optex Systems, Inc. (Delaware) after the closing referenced above. Gross proceeds to Optex Systems, Inc. (Delaware) were \$1,219,750, and after deducting (i) a cash finder's fee of \$139,555, (ii) non-cash consideration of indebtedness owed to an investor of \$146,250, and (iii) stock issuance costs of \$59,416, net proceeds were \$874,529. The finder also received five year warrants to purchase 2.39 units, at an exercise price of \$49,500 per unit.

The following table represents the reorganization and private placement transactions which occurred on March 30, 2009 reflected in March 29, 2009 statements due to the election to report as of the accounting acquirers' period end:

Optex Systems Holdings, Inc.
Balance Sheet Adjusted for Reorganization and Private Placement

	Jnaudited Quarter ed March 29, 2009	•	ganization ustments (1)	Pl	Private acement justments	udited Quarter ed March 29, 2009
Assets						
Current Assets	\$ 8,880,436	\$	187,500	\$	929,738	\$ 9,997,674
Non current Assets	10,422,425		-		-	10,422,425

Total Assets	\$ 19,302,861	\$ 187,500	\$ 929,738	\$ 20,420,099
Liabilities				
Loans Payable	146,709		(146,250)	459
Other Current Liabilities	4,416,403	-	55,209	4,471,612
Total Liabilities	\$ 4,563,112	\$ -	\$ (91,041)	\$ 4,472,071
Equity				
Optex Systems Holdings, Inc. – (par \$0.001per share, 200,000,000 shares authorized, 138,914,940 shares issued and outstanding as				
of March 29, 2009)	113,333	17,450	8,132	138,915
Optex Systems Holdings, Inc. preferred stock (par value \$0.001per share, 5,000 shares authorized, 1027 shares of Series A Preferred				
issued and outstanding)	1			1
Additional Paid in Capital	15,046,446	170,050	1,012,647	16,229,143
Retained Earnings	(420,031)			(420,031)
Total Stockholders Equity	\$ 14,739,749	\$ 187,500	\$ 1,020,779	\$ 15,948,028
Total Liabilities and Stockholders Equity	\$ 19,302,861	\$ 187,500	\$ 929,738	\$ 20,420,099
53				

(1) Sustut Exploration, Inc. Balance Sheet as of the March 30, 2009 reorganization. Other assets include \$187,500 in prepaid expenses for investor relation services to be realized over the next 12 months. The services were prepaid by the issuance of 1,250,000 Sustut shares by Sustut prior to March 30, 2009. The original prepaid expense covered April 2009 through April 2010. On June 29, 2009 700,000 of these shares were returned to Optex Systems Holdings due to the cancellation of one of the investor relations agreements. The amortized expense related to the remaining 550,000 shares has been reflected on the Consolidated Statement of Operations for Optex Systems Holdings as expensed.

The expenses reflected by Optex Systems Holdings on its Statement of Operations were increased by \$63,750 for fiscal year 2009 and \$18,750 for 2010 (as a non-cash expense) as a result of the issuance of the 1,250,000 shares for Investor Relations Services by Sustut and subsequent return of 700,000 shares to Optex Systems Holdings and are carried on the Optex Systems Holdings' Balance Sheet as a prepaid expense. The same Investor Relations agreements also called for an aggregate cash payment \$36,000 for 2009. Therefore, the total pre-tax impact of the agreements for Investor Relations Services was \$99,750 for fiscal 2009 including both the cash expense and the amortization of the prepaid expense which is carried on the Condensed Consolidated Balance Sheet of Optex Systems Holdings.

The accompanying unaudited pro forma financial information for the consolidated successor year ended September 27, 2009 present the historical financial information of the accounting acquirer. The pro forma financial information is presented for information purposes only. Such information is based upon the standalone historical results of each company and does not reflect the actual results that would have been reported had the acquisition been completed when assumed, nor is it indicative of the future results of operations for the combined enterprise.

The following represents condensed pro forma revenue and earnings information for the fiscal year ended September 27, 2009 as if the acquisition of Optex Systems, Inc. (Texas) and the reorganization had occurred on the first day of the fiscal year.

	Unaudited, Pro forma			
	`	Year Ended		
	Se	September 27,		
	2009			
Revenues	\$	27,580,737		
Net Income (Loss) applicable to common shareholders	\$	(362,149)		
Diluted earnings per share	\$	(0.00)		
Weighted Average Shares Outstanding		139,045,625		

The unaudited, pro forma information depicted above reflect the impacts of reduced interest expense, increased intangible amortization expenses, the elimination of corporate allocation costs from Irvine Sensors Corporation and the elimination of employee stock bonus compensation previously allocated from Irvine Sensors Corporation to reflect the costs of the ongoing entity.

Note 6 - Property and Equipment

A summary of property and equipment at October 3, 2010 and September 27, 2009 is as follows:

	Estimated Useful Life	Year Ended October 3, 2010	Se	Year Ended ptember 27, 2009
Property and Equipment		,		,
Furniture and Equipment	3-5yrs	\$ 175,859	\$	159,724
Machinery and Equipment	5 yrs	1,063,199		1,034,440
Leasehold Improvements	7 yrs	217,916		147,107
Less: Accumulated Depreciation		(1,160,677)		(1,094,526)
Net Property & Equipment		\$ 296,297	\$	246,745
Depreciation Expense		\$ 66,151	\$	99,984

Depreciation expense included in cost of goods sold and general and administrative expense for fiscal 2010 is \$42,615 and \$23,536, respectively. Depreciation expense included in cost of goods sold and general and administrative expense for fiscal 2009 is \$61,628 and \$38,356 respectively, inclusive of \$9,691 of depreciation for September 28 through October 14, 2008 Optex Systems, Inc. (Texas) predecessor.

Note 7 – Accrued Liabilities

The components of accrued liabilities for years ended October 3, 2010 and September 27, 2009 are summarized below:

Year Ended	Year Ended
October 3, 201\square	eptember 27, 2009

Customer Advance		
Payments	\$ - \$	80,753
Deferred Rent Expense	115,914	27,860
Accrued Vacation	178,324	153,291
Property Taxes	18,057	17,532
Accrued Interest	207	-
Franchise Taxes	1,986	5,100
Operating Expenses	119,021	244,884
Payroll & Payroll		
Related	140,421	141,625
Total Accrued Expenses	\$ 573,930 \$	671,045

Note 8 - Commitments and Contingencies

Leases

Pursuant to a lease amendment effective January 4, 2010, Optex Systems Holdings leases its office and manufacturing facilities under a non-cancellable operating lease expiring July 31, 2015 several non-cancellable operating leases for office and manufacturing equipment. Total expenses under facility lease agreements for the fiscal year ended October 3, 2010 was \$256,755. Total expenses for manufacturing and office equipment year ended 2010 was \$30,946. Total expenses under facility lease agreements for the fiscal year ended September 27, 2009 was \$309,693

and total expenses for manufacturing and office equipment was \$2,726.

At October 3, 2010, the remaining minimum lease payments under the non-cancelable operating leases for equipment, office and facility space were as follows:

	Operating
	Leases
Fiscal Year	
2011	\$ 251,152
2012	236,112
2013	231,574
2014	241,748
2015	201,457

Total minimum lease payments \$ 1,162,043

Pursuant to the terms of the amendment to the facilities lease, there was no base rent payment due from January 1, 2010 through July 31, 2010, and the total value of this rent abatement is \$133,898. The value of the deferred rent expense will be amortized monthly at a rate of \$1,998 per month over the life of the lease. The total unamortized deferred rent as of October 3, 2010 was \$115,914. Commencing on August 1, 2010, the base rent payment is \$19,128 per month.

Note 9 - Transactions with a Related Party

There are were no transactions with Related Parties during fiscal years 2010 or 2009 except as described below in Note 10 Debt Financing.

Note 10 - Debt Financing

Related Parties

Short Term Note Payable/Longview Fund - On September 23, 2008, Optex Systems, Inc. (Delaware) borrowed \$146,709 from Longview and issued a promissory note dated September 23, 2008, to Longview in connection therewith. Pursuant to an to the promissory note, dated January 20, 2009, the maturity date was extended until March 31, 2009. On March 30, 2009 in conjunction with the reorganization and private placement, Longview Fund purchased 3.25 units of the private placement using \$146,250 of the outstanding note payable as consideration for the purchase. (See Note 5). In the year ended 2009, Optex Systems paid \$459 against the principal balance recorded interest expenses and paid \$7,557 as a result of the interest accrued on the note prior to its conversion to common stock.

Short term note payable (Qioptic) - On November 20, 2008, Optex Systems, Inc. (Delaware) issued a promissory note to Qioptiq Limited in the amount of \$117,780. The note originated as a trade payable as of September 28, 2008 in the amount of \$227,265, and was paid in full, including accrued interest expense of \$2,733, as of March 29, 2009.

Short Term Note Payable/Longview Fund - On October 27, 2009, Optex Systems Holdings borrowed \$250,000 from the Longview Fund, a related party, pursuant to a promissory note, with an original maturity date of December 1, 2009, which was extended to July 15, 2010 pursuant to an allonge dated January 5, 2010. The note carried an interest rate of 10% per annum, and all accrued and unpaid interest thereon was due upon maturity. The note required Optex Systems Holdings to make a prepayment equal to 50% of the then outstanding principal amount plus accrued and unpaid interest thereon upon the closing of a credit facility or other equity or debt financing from which the net proceeds to Optex Systems Holdings were at least \$900,000, with any remaining unpaid balance due on July 15,

2010. In exchange for the allonge, Optex Systems Holdings granted Longview a warrant to purchase 100,000 shares of its restricted common stock with an exercise price of \$0.15 per share and with a term of three years. In conjunction with the Peninsula Bank financing (below) on March 22, 2010, Optex Systems Holdings paid to Longview a principal prepayment of \$125,000 and \$10,000 in accrued interest. The remaining principal amount of the note of \$125,000 plus all accrued and unpaid interest thereon was paid in full on June 4, 2010.

Credit Facility - Peninsula Bank Business Funding

Effective March 4, 2010, Optex Systems, Inc. (Delaware) entered into a Loan and Security Agreement ("Agreement") with Peninsula Bank Business Funding, a division of the Private Bank of the Peninsula ("Lender").

The Agreement provides for a revolving line of credit of up to \$2,000,000, based upon advances to be made against percentages of eligible receivables as set forth in the Agreement. The material terms of the Agreement are as follows:

- The interest rate for all advances shall be the greater of 8.5% and the then in effect prime rate plus 3.5% and subject to a minimum quarterly interest payment of \$16,000.
 - Interest shall be paid monthly in arrears.
- The expiration date of the Agreement is March 4, 2011, at which time any outstanding advances, and accrued and unpaid interest thereon, will be due and payable.
- In connection with the entry into the Agreement by the Lender, Optex Systems, Inc.(Delaware) paid the Lender a facility fee of \$20,000 and issued a warrant to Lender to purchase 1,000,000 shares of its common stock. The warrant bears an exercise price of \$0.10 per share and expires on March 3, 2016.
- The obligations of Optex Systems, Inc. (Delaware) to the Lender are secured by a first lien on all of its assets (including intellectual property assets should it have any in the future) in favor of the Lender.
- The Agreement contains affirmative and negative covenants that require Optex Systems, Inc. (Delaware) to maintain certain minimum cash and EBITDA levels on a quarterly basis and contains other customary covenants. The Agreement also contains customary events of default. Upon the occurrence of an event of default that remains uncured after any applicable cure period, the Lender's commitment to make further advances may terminate, and the Lender would also be entitled to pursue other remedies against Optex Systems, Inc. (Delaware) and the pledged collateral.
- Pursuant to a guaranty executed by Optex Systems Holdings in favor of Lender, Optex Systems Holdings has guaranteed all obligations of Optex Systems, Inc. (Delaware) to Lender.

During the three months ending June 27, 2010, Optex Systems Holdings realized negative EBITDA of (\$78,986) as compared to a loan covenant requirement of \$350,000 and as such did not meet the EBITDA covenant of the Loan Security Agreement for the third fiscal quarter of 2010. On August 3, 2010, Peninsula Bank Business Funding waived the Company's requirement to meet the EBITDA requirement set forth in Section 6.8 of its agreement with the Company for the quarter ended June 27, 2010. In addition, Peninsula Bank Business Funding agreed to amend Sections 6.8(c) and (d) of the aforesaid agreement to adjust the minimum EBITDA covenant for the fiscal quarter ending October 3, 2010 to \$20,000, and for the fiscal quarter ending January 2, 2011 to \$200,000.

During the three months ending October 3, 2010, Optex Systems Holdings realized negative EBITDA of (\$1,223,139) as compared to a loan covenant requirement of \$20,000 and as such did not meet the EBITDA covenant of the Loan Security Agreement for the fourth fiscal quarter of 2010. On November 23, 2010, Peninsula Bank Business Funding waived the Company's requirement to meet the EBITDA requirement set forth in Section 6.8 (c) of the August 3, 2010 amended Agreement for the fourth quarter ended October 3, 2010. In addition, on November 29, 2010 Peninsula Bank Business Funding agreed to a second amendment for Sections 6.8 (d) of the Agreement to adjust the minimum EBITDA covenant for the fiscal quarter ending January 2, 2011 to \$95,000.

As of October 3, 2010, the outstanding balance on the line of credit is \$1,106,852. For the period ended October 3, 2010, the total interest expense against the outstanding line of credit balance was \$37,148.

Note 11 – Intangible Assets and Goodwill

On October 14, 2008, in a purchase transaction that was consummated via public auction, Optex Systems, Inc. (Delaware) (Successor) purchased all of the assets of Optex Systems, Inc. (Texas) (Predecessor) in exchange for \$15 million of Irvine Sensors Corporation debt owned by it and the assumption of approximately \$3.8 million of certain Optex Systems, Inc. (Texas) liabilities (see Note 4). Optex Systems, Inc. (Delaware) has allocated the consideration for its acquisition of the Purchased Assets among tangible and intangible assets acquired and liabilities assumed based upon their fair values. Assets that met the criteria for recognition as intangible assets apart from goodwill were also valued at their fair values.

The purchase price was assigned to the acquired interest in the assets and liabilities of Optex Systems Holdings as of October 14, 2008 as follows:

Assets:	
Current assets, consisting primarily of inventory of \$5,383,929 and accounts receivable of	
\$1,404,434	\$ 7,330,910
Identifiable intangible assets	4,036,789
Purchased Goodwill	7,110,416
Other non-current assets, principally property and equipment	343,898
Total assets	\$ 18,822,013
Liabilities:	
Current liabilities, consisting of accounts payable of \$1,953,833 and accrued liabilities of	
\$1,868,180	3,822,013
Acquired net assets	\$ 15,000,000

The goodwill of Optex Systems Holdings, Inc. was reviewed as of October 3, 2010 and in light of a reduction of new and expected orders culminating in a lower backlog and reduced revenue forecasts. The review indicated that goodwill was impaired, as determined based on a projected cash flow analysis of Optex Systems Holdings future operations. The impairment loss for goodwill was \$7,110,415. The goodwill was written off as a component of general and administrative operating expenses during fiscal year 2010.

The following table summarizes the estimate of the fair values of the intangible assets as of the asset transfer date:

	Total
Contracted Backlog - Existing Orders	\$ 2,763,567
Program Backlog - Forecasted Indefinite Delivery/Indefinite Quantity awards	1,273,222
Total Intangible Asset to be amortized	\$ 4,036,789

The amortization of identifiable intangible assets associated with the Optex Systems Inc. (Texas) acquisition on October 14, 2008 expensed for fiscal years 2010 and 2009 was \$1,037,581 and \$2,071,193, respectively. The expenses split between manufacturing cost of sales and general and administrative cost were \$718,290 and \$319,291, respectively, for 2010. The expenses split between manufacturing cost of sales and general and administrative cost were \$1,666,558 and \$404,635, respectively, for 2009. The identifiable intangible assets and recorded goodwill are amortized over five years for book purposes and is deductible over 15 years for income tax purposes.

As of the year ended September 27, 2009, the total unamortized balance of intangible assets was \$1,965,596. The amortizable intangible assets were tested for impairment as of September 27, 2009 based on discounted cash flows and no impairment was required. As of October 3, 2010 the intangible assets were reviewed in light of a reduction of expected delivery orders against contracted orders and higher than expected costs on those orders. The review indicated that intangible assets were impaired, as determined based on a projected cash flow analysis of Optex System Holdings future operations. The impairment loss recorded in 2010 for intangible assets was \$928,016 and was split between cost of goods sold and general and administrative costs in the amount of \$150,534, and \$777,482 respectively.

As of the year ended October 3, 2010, after impairment, the total unamortized balance of intangible assets was zero.

Note 12-Stock Based Compensation

On March 26, 2009, the Board of Directors adopted the 2009 Stock Option Plan providing for the issuance of up to 6,000,000 shares to Optex Systems Holdings officers, directors, employees and to independent contractors who provide services to Optex Systems Holdings.

Options granted under the 2009 Stock Option Plan vest as determined by the Board of Directors of Optex Systems Holdings or a committee set up to act as a compensation committee of the Board of Directors and terminate after the earliest of the following events: (i) expiration of the option as provided in the option agreement, (ii) 90 days following the date of termination of the employee, or (iii) ten years from the date of grant (five years from the date of grant for incentive options granted to an employee who owns more than 10% of the total combined voting power of all classes of Optex Systems Holdings stock at the date of grant). In some instances, granted stock options are immediately exercisable into restricted shares of common stock, which vest in accordance with the original terms of the related options. Optex Systems Holdings recognizes compensation expense ratably over the requisite service period.

The option price of each share of common stock is determined by the Board of Directors or a committee set up to act as a compensation committee, provided that with respect to incentive stock options, the option price per share will in all cases be equal to or greater than 100% of the fair value of a share of common stock on the date of the grant, except an incentive option granted under the 2009 Stock Option Plan to a shareholder that owns more than 10% of the total combined voting power of all classes of Optex Systems Holdings' stock, will have an exercise price of not less than 110% of the fair value of a share of common stock on the date of grant. No participant may be granted incentive stock options, which would result in shares with an aggregate fair value of more than \$100,000 first becoming exercisable in one calendar year.

On March 30, 2009, 1,414,649 stock options with an exercise price of \$0.15 were granted to an officer of Optex Systems Holdings which vest as follows: 34% after the first year, and 33% each after the second and third years. These options carry a grant expiration date of seven years after issuance. On May 14, 2009, 1,267,000 stock options were issued to other Optex Systems Holdings employees, including 250,000 shares to one officer. These stock options vest 25% per year after each year of employment and carry a grant expiration date of seven years after issuance. For shares granted as of May 14, 2009, Optex Systems Holdings anticipates an annualized employee turnover rate of 3% per year, and as such anticipates that only 1,174,786 of the 1,267,000 shares will vest as of the end of the contract term.

As of October 3, 2010, 776,981 of the awarded stock options had vested and 83,000 shares had been forfeited due to employee turnover. As of September 27, 2009, none of the awarded stock options had vested and 14,000 shares had been forfeited due to employee turnover.

Optex Systems Holdings recorded compensation costs for options and shares granted under the plan amounting to \$97,311 and \$39,528 for the fiscal years ended October 3, 2010 and September 27, 2009, respectively The impact of this expense was immaterial to the basic and diluted net loss per share for the fiscal years ended October 3, 2010 and September 27, 2009. A deduction is not allowed for income tax purposes until nonqualified options are exercised. The amount of this deduction will be the difference between the fair value of Optex Systems Holdings' common stock and the exercise price at the date of exercise. For the year ended October 3, 2010, estimated deferred tax assets related to option compensation costs were \$33,085 and have been recorded for the tax effect of the financial statement expense. For the year ended September 27, 2009 the estimated deferred tax assets related to option compensation costs were \$13,440 and have been recorded for the tax effect of the financial statement expense. No tax deduction is allowed for incentive stock options. Accordingly no deferred tax asset is recorded for GAAP expense related to these options.

Management has valued the options at their date of grant utilizing the Black-Scholes-Merton option pricing model. The fair value of the underlying shares was determined based on the opening price of Optex Systems Holdings' publicly-traded shares as of September 28, 2009. Further, the expected volatility was calculated using the historical volatility of a diversified index of companies in the defense, homeland security, and space industry in accordance with FASB ASC 718-10-S99-1 (Prior authoritative literature: Ouestion 6 of SAB Topic 14.D.1). In making this determination and trying to find another comparable company, Optex Systems Holdings considered the industry, stage of life cycle, size and financial leverage of such other entities. Based on the development stage of Optex Systems Holdings, similar companies with sufficient historical data were not available. Optex Systems Holdings utilized the three year volatility of the SPADE Defense Index, which is a diversified index of 58 companies in the same industry as Optex Systems Holdings. The risk-free interest rate is based on the implied yield available on U.S. Treasury issues with an equivalent term approximating the expected life of the options depending on the date of the grant and expected life of the options. The expected life of options used was based on the contractual life of the option grant. Optex Systems Holdings determined the expected dividend rate based on the assumption and expectation that earnings generated from operations are not expected to be adequate to allow for the payment of dividends in the near future and the assumption that Optex Systems Holdings does not presently have any intention of paying cash dividends on its common stock.

Optex Systems Holdings has granted stock options to officers and employees as follows:

	Date of Grant	Shares Granted	Exercise Price	Shares Outstanding As of 10/03/10	Expiration Date	Vesting Date
03/30/09		480,981	\$ 0.15	480,981	03/29/2016	03/30/2010
03/30/09		466,834	\$ 0.15	466,834	03/29/2016	03/30/2011
03/30/09		466,834	\$ 0.15	466,834	03/29/2016	03/30/2012
05/14/09		316,750	\$ 0.15	296,000	05/13/2016	05/14/2010
05/14/09		316,750	\$ 0.15	296,000	05/13/2016	05/14/2011
05/14/09		316,750	\$ 0.15	296,000	05/13/2016	05/14/2012
05/14/09		316,750	\$ 0.15	296,000	05/13/2016	05/14/2013
Total		2,681,649		2,598,649		

The following table summarizes the status of Optex Systems Holdings' aggregate stock options granted under the incentive stock option plan:

Subject to Exercise	Number of Shares Remaining Options		Weighted Average Intrinsic Price	Weighted Average Life (Years)	Aggregate Value
Outstanding as of September 27, 2009	2,667,649	\$	0.21	5.14	560,206
		ф		5.14	300,200
Granted – 2010	_	_\$	_		
Forfeited – 2010	(69,000)	\$	_		
Exercised – 2010	_	_\$			_
Outstanding as of October 3, 2010	2,598,649	\$	_	4.13	_
-					
Exercisable as of October 3, 2010	776,981	\$	_		_

There were no new options granted or exercised during the year ended October 3, 2010. The total intrinsic value of options forfeited during the year ended October 3, 2010 was \$0.

The following table summarizes the status of Optex Systems Holdings' aggregate non-vested shares granted under the 2009 Stock Option Plan (See Note 9):

	Number of		
	Non-	Weigh	ted-
	vested Averag		ige
	Shares Gran		ıt-
	Subject to	Dat	e
	Options	Fair V	alue
Non-vested as of September 27, 2009	2,667,649	\$	0.14
Non-vested granted — year ended October 3, 2010		-\$	0.00
Vested — year ended October 3, 2010	(776,981)	\$	0.12
Forfeited — year ended October 3, 2010	(69,000)	\$	0.15
Non-vested as of September 29, 2009	1,821,668	\$	0.15

The total share-based compensation expense of Optex Systems Holdings, Inc during fiscal years 2010 and 2009 attributable to the stock option grants was \$97,311 and \$39,528, respectively, and was charged as general and administrative compensation expense for each of the respective fiscal years. As of October 3, 2010, the unrecognized compensation cost related to non-vested share based compensation arrangements granted under the plan was approximately \$226,979. These costs are expected to be recognized on a straight line basis from March 30, 2009 through May 13, 2013. The total fair value of options and shares vested during the year ended October 3, 2010 was \$93,742.

For the fiscal year ended September, 27, 2009, Optex Systems Holdings issued 480,000 shares of common stock at a market value of \$0.30 per share for a total \$144,000 and paid \$150,000 cash to a vendor in support of an investor relations agreement executed on June 29, 2009. Pursuant to the agreement, the shares are earned over the life of the contract at the rate of 40,000 shares per month through June 2010. Optex Systems Holdings expensed \$ 108,000 and \$36,000 for shares earned during fiscal years 2010 and 2009, respectively and the unamortized balance of shares issued against the contract is zero as of October 3, 2010.

There were no stock options issued to Optex Systems Holdings employees or equity instruments issued to consultants and vendors in fiscal 2010.

Warrant Agreements:Optex Systems Holdings calculates the fair value of warrants issued with debt or preferred stock using the Black-Scholes-Merton valuation method. The total proceeds received in the sale of debt or preferred stock and related warrants are allocated among these financial instruments based on their relative fair values. The discount arising from assigning a portion of the total proceeds to the warrants issued is recognized as interest expense for debt from the date of issuance to the earlier of the maturity date of the debt or the conversion dates using the effective yield method.

As of October 3, 2010, Optex Systems Holdings had the following warrants outstanding:

		Warrants	Exercise Outstanding as of		Expiration	
	Grant Date	Granted	Price	10/03/10	Date	Term
Private Placement Stock						
Holders	3/30/2009	8,131,667	\$ 0.450	8,131,667	3/29/2014	5 years
Finder Fee on Private						
Placement	3/30/2009	717,000	\$ 0.165	717,000	3/29/2014	5 years
Longview Fund Allonge						
Agreement	1/5/2010	100,000	\$ 0.150	100,000	1/4/2013	3 years
Peninsula Bank Business						
Funding - Line of Credit	3/4/2010	1,000,000	\$ 0.100	1,000,000	3/3/2016	6 years
Total Warrants		9,948,667		9,948,667		

During the period ended October 3, 2010, Optex Systems Holdings recorded a total of \$19,500 in interest expense related to the outstanding warrants and has an unamortized interest balance of \$12,500. These warrants are not included in the computation of weighted average of shares as it would be anti-dilutive.

Note 13 – Stockholders Equity

Common stock:

Optex Systems, Inc. (Texas) was authorized to issue 100,000 shares of no par common stock. At September 28, 2008 there were 18,870 shares issued and 10,000 shares outstanding.

The common stock, treasury stock and additional paid in capital accounts have been presented to reflect the ownership structure of Optex Systems, Inc. (Texas) as it existed prior to the acquisition by Irvine Sensors Corporation, since Optex Systems, Inc. (Texas) is presenting its financial statements as a separate, stand-alone entity.

On October 14, 2008, in a purchase transaction that was consummated via public auction, Optex Systems, Inc. (Delaware) (Successor) purchased all of the assets of Optex Systems, Inc. (Texas) (Predecessor) in exchange for \$15 million of Irvine Sensors Corporation debt owned by it and the assumption of approximately \$3.8 million of certain Optex Systems, Inc. (Texas) liabilities. The \$15 million of Irvine Sensors Corporation debt was contributed by Longview and Alpha to Optex Systems, Inc. (Delaware), in exchange for a \$6 million note payable from Optex Systems, Inc. (Delaware) and a \$9 million equity interest in Optex Systems, Inc. (Delaware) (which consisted of the issuance by Optex Systems, Inc. (Delaware) of 45,081,350 and 4,918,650 shares of its common stock to each of Longview Fund and Alpha, respectively). On October 30, 2008, Alpha sold its Optex Systems, Inc. (Delaware) common stock to Arland Holdings, Ltd. There was no contingent consideration associated with the purchase. Longview and Arland Holdings, Ltd. both owned Optex Systems, Inc. (Delaware) until February 20, 2009, when Longview sold 100% of its equity interests in Optex Systems, Inc. (Delaware) to Sileas Corp., as discussed below.

On February 20, 2009, Sileas purchased 100% of the equity and debt interest held by Longview, representing 90% of Optex Systems, Inc. (Delaware). As of the date of this transaction, Sileas is the majority owner of Optex Systems Holdings.

Stock Split

On March 26, 2009, Optex Systems, Inc. (Delaware)'s Board of Directors reconfirmed a 1.7:1 forward split of its common stock to holders of record as of February 23, 2009. Accordingly, as a result of the forward split, the 45,081,350 shares of common stock held by Sileas were split into 76,638,295 shares, and the 4,918,650 shares of common stock held by Arland Holdings, Ltd. were split into 8,361,705 shares.

As of March 30, 2009, Optex Systems, Inc. (Delaware) was authorized to issue 200,000,000 shares of \$0.001 par value common stock, of which 85,000,000 shares were issued and outstanding as follows:

Sileas Corporation	76,638,295
Arland Holdings, Ltd.	8,361,705
Total Outstanding	85,000,000

Reorganization & Private Placement:

On March 29, 2009, as a result of the reorganization agreement and private placement, the 85,000,000 outstanding shares of Optex Systems, Inc. (Delaware) as of March 30, 2009 were exchanged for 113,333,282 shares of Optex Systems Holdings (formerly Sustut Exploration, Inc.). An additional 8,131,667 shares were issued in connection with the private placement closed prior to the reorganization.

On June 29, 2009, 750,000 common shares were sold to in a private transaction for gross proceeds of \$150,000.

Each share of stock entitles the holder to one vote on matters brought to a vote of the shareholders.

Optex Systems Holdings granted an officer at the consummation of the reorganization, options to purchase 1,414,649 shares with an exercise price of \$0.15 per share. The options vest 34% one year following the date of grant, and 33% on each of the second and third anniversaries following the date of grant. See Note 12 - Stock Based Compensation.

Series A preferred stock

On March 24, 2009, Optex Systems Holdings filed a Certificate of Designation with the Secretary of State of the State of Delaware authorizing a series of preferred stock, under its articles of incorporation, known as "Series A preferred

stock". This Certificate of Designation was approved by Optex Systems Holdings' Board of Directors and Shareholders at a Board Meeting and Shareholders Meeting held on February 25, 2009. The Certificate of Designation sets forth the following terms for the Series A preferred stock: (i) number of authorized shares: 1,027; (ii) per share stated value: \$6,000; (iii) liquidation preference per share: stated value; (iv) conversion price: \$0.15 per share as adjusted from time to time; and (v) voting rights: votes along with the common stock on an as converted basis with one vote per share.

The Series A preferred stock entitles the holders to receive cumulative dividends at the rate of 6% per annum, payable in cash at the discretion of Board of Directors. Each share of preferred stock is immediately convertible into common shares at the option of the holder which entitles the holder to receive the equivalent number of common shares equal to the stated value of the preferred shares divided by the conversion price, which was initially set at \$0.15 per share.

Holders of preferred shares receive preferential rights in the event of liquidation. Additionally the preferred stock shareholders are entitled to vote together with the common stock on an "as-converted" basis.

On March 27, 2009, Sileas and Alpha exchanged their promissory notes in the total amount of \$6,000,000 plus accrued and unpaid interest thereon into 1,027 shares of Series A preferred stock. On March 30, 2009, shares of Optex Systems, Inc. Series A preferred stock was exchanged on a 1:1 basis for Series A preferred stock of Optex Systems Holdings. As of the years ended October 3, 2010 and September 27, 2009, Optex Systems has recorded \$389,551 and \$186,246 of dividends payable on Series A preferred shares, respectively.

Cancellation of Common Stock

On June 29, 2009 Optex cancelled an investor relations agreement resulting in the return of 700,000 shares of common stock previously issued by Sustut prior to the reverse Merger on March 30, 2009. The shares were valued at \$105,000, returned to Optex System Holdings, Inc., and then cancelled. (see also Note 12 on new investor relations shares issued).

During the year ended October 3, 2010 there were no new issues of common or preferred stock.

Note 14 - Income Taxes

The income tax provisions as of October 3, 2010 and September 27, 2009 include the following:

	2010	2009
Current income tax expense:		
Federal	\$ (32,389) \$	426,514
State	-	-
	\$ (32,389) \$	426,514
Deferred income tax provision (benefit):		
Federal	(3,372,724)	(711,177)
State	-	-
Change in valuation allowance	3,090,405	_
	\$ (282,316) \$	(711,177)
Provision for (Benefit from) income taxes, net	\$ (314,705) \$	(284,663)

The current income tax expense for period ending October 3, 2010 relates to changes in the actual income tax return filed in June 2010, for tax year ending September 27, 2009 as compared to the estimated taxes as of September 27, 2009.

The income tax provision for Optex Systems as of October 3, 2010 differs from those computed using the statutory federal tax rate of 34%, due to the following permanent differences:

	2010	%	2009	%
Tax benefit at statutory federal				
rate	\$ (3,375,071)	34%	\$ (127,211)	34%
Change in valuation and other	3,060,366	(30.7)%	(157,452)	42%
	\$ (314,705)	3.3%	\$ (284,663)	76%

Deferred income taxes recorded in the balance sheets results from differences between financial statement and tax reporting of income and deductions. A summary of the composition of the deferred income tax assets (liabilities) follows:

	As	of October 3, 2010	As of September 27, 2009
		2010	2009
Stock Options	\$	46,525	\$ 13,440
Inventory Reserve		(84,339)	(40,427)
Unicap		40,051	54,494
Contract Loss Reserve		181,962	178,900
Fixed assets		(37,141)	(58,476)
Goodwill Amortization		2,256,372	-
Intangible Asset Amortization		1,189,509	612,707
Net Operating Losses		553,012	
Other		(62,050)	(49,461)
Subtotal	\$	4,083,901	\$ 711,177
Valuation allowance		(3,090,405)	-
Net deferred asset (liability)	\$	993,496	\$ 711,177

Since Optex Systems, Inc. (Texas) was acquired in a transaction effected as an asset purchase, Optex Systems, Inc. (Delaware) would only be entitled to tax deductions generated after the date of the acquisition on October 14, 2008. Accordingly, no deferred tax assets have been recorded in the accompanying financial statements for net operating losses generated by Optex Systems, Inc. (Texas) prior to that date.

As the result of the assessment of the FASB ASC 740-10 (Prior Authoritative Literature: FASB Interpretation No. 48 ("FIN 48"), "Accounting for Uncertainty in Income Taxes — An Interpretation of FASB Statement No. 109"), Optex Systems Holdings has no unrecognized tax benefits. By statute, the tax year ending in October 3, 2010 is open to examination by the major taxing jurisdictions to which the Optex Systems Holdings is subject.

Cash paid for income taxes for the fiscal years ended October 3, 2010 and September 27, 2009 were \$119,847 and \$488,799, respectively. As of October 3, 2010 Optex Systems Holdings other assets includes \$214,521 of overpaid income taxes from estimated quarterly deposits. We expect recovery of the overpaid tax amount in the next 12 months.

Note 15 — Defined Contribution Plan

The Company sponsors a defined contribution pension plan under Section 401(k) of the Internal Revenue Code for all employees. Company contributions are voluntary and at the discretion of the Board of Directors. The Company's contribution expense for the both the fiscal years ended October 3, 2010, and September 27, 2009 was zero.

Note 16 — Subsequent Events

On November 23, 2010, Peninsula Bank Business Funding waived the Company's requirement to meet the EBITDA requirement set forth in Section 6.8 (c) of the August 3, 2010 amended Agreement for the fourth quarter ended October 3, 2010. In addition, on November 29, 2010 Peninsula Bank Business Funding agreed to a second amendment for Section 6.8 (d) of the Agreement to adjust the minimum EBITDA covenant for the fiscal quarter ending January 2, 2011 to \$95,000.

Item 9. Changes in and Disagreements With Accountants on Accounting and Financial Disclosure.

None.

Item 9A. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

As of October 3, 2010, management performed, with the participation of our Principal Executive Officer and Principal Financial Officer, an evaluation of the effectiveness of our disclosure controls and procedures as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act. Our disclosure controls and procedures are designed to ensure that information required to be disclosed in the report we file or submit under the Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the SEC's forms, and that such information is accumulated and communicated to our management including our Principal Executive Officer and our Principal Financial Officer, to allow timely decisions regarding required disclosures. Based on the evaluation, our Principal Executive Officer and our Principal Financial Officer concluded that, as of October 3, 2010, our disclosure controls and procedures were effective.

Management's Report on Internal Control Over Financial Reporting

Management is responsible for establishing and maintaining adequate internal control over financial reporting, as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act. Internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements in accordance with GAAP. Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projection of any evaluation of effectiveness to future periods is subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

Management has conducted, with the participation of our Principal Executive Officer and our Principal Financial Officer, an assessment, including testing of the effectiveness, of our internal control over financial reporting as of October 3, 2010. Management's assessment of internal control over financial reporting was conducted using the criteria in Internal Control over Financial Reporting - Guidance for Smaller Public Companies issued by the Committee of Sponsoring Organizations of the Treadway Commission.

A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis. In connection with our management's assessment of our internal control over financial reporting as required under Section 404 of the Sarbanes-Oxley Act of 2002, we have not identified any material weaknesses in our internal control over financial reporting as of October 3, 2010. We have thus concluded that our internal control over financial reporting was effective as of October 3, 2010.

PART III

Item 10 Directors, Executive Officers and Corporate Governance

MANAGEMENT

Our board of directors directs the management of the business and affairs of our company as provided in our certificate of incorporation, our by-laws and the General Corporation Law of Delaware. Members of our board of directors keep informed about our business through discussions with senior management, by reviewing analyses and reports sent to them, and by participating in board and committee meetings.

Directors and Executive Officers

The following table sets forth information regarding the members of our board of directors and our executive officers and other significant employees. All of our current officers and directors were appointed on March 30, 2009, the closing date of the reorganization.

The following table sets forth certain information with respect to the directors and executive officers of Optex Systems Holdings:

Name	Age	Position
Stanley A. Hirschman	64	President, Secretary, Treasurer & Director
Merrick D. Okamoto	50	Director

Ronald F. Richards	44	Chairman of the Board
Danny Schoening	46	Chief Operating Officer
Karen L. Hawkins	45	Vice President of Finance and Controller
65		

Stanley A. Hirschman. Mr. Hirschman served as a Director and President of Optex Systems, Inc. (Delaware) since September 28, 2008 and assumed the same roles on behalf of Optex Systems Holdings on March 30, 2009, in which roles he is committed to providing Optex his management experience and provides direction and oversight of other executive officers and management. From 1997 to 2009, he was president of CPointe Associates, Inc., a Plano, Texas consulting group, and provided consulting services to small and medium sized companies. As of October 2009, in order to meet his responsibilities at Optex, he concluded his active role at CPointe. Additionally, since February 2009 he has been the majority beneficial owner of Sileas Corp (which has no active business), the majority shareholder of Optex Systems Holdings, During the past five years, Mr. Hirschman has also sat on the following Boards: Goldspring, Inc., Bravo Brands, 5G Wireless Communications, Axion Power International, Inc., SVI Media, Inc., Bronco Energy, Energy + Engine Technology, Dalrada Financial, Datascension, iWorld Projects & Systems, Inc. and South Texas Oil. Prior to establishing CPointe Associates, he was Vice President Operations, Software Etc., Inc., a 396 retail store software chain, from 1989 until 1996. He has also held executive positions with T.J. Maxx, Gap Stores and Banana Republic. Mr. Hirschman is a member of the National Association of Corporate Directors, regularly participates in the KMPG Audit Committee Institute and is a graduate of the Harvard Business School Audit Committees in the New Era of Governance symposium. He is active in community affairs and serves on the Advisory Board of the Salvation Army Adult Rehabilitation Centers.

Merrick D. Okamoto. Mr. Okamoto served as a Director of Optex Systems, Inc. (Delaware) since October 2008 and has served as a Director of Optex Systems Holdings since March 30, 2009. In 2001, Mr. Okamoto co-founded Viking Asset Management, LLC and is the President and a Managing Member. Viking Asset Management is the investment advisor to Longview Fund, LP and Longview Fund International, Ltd. Limited partners in Viking's family of funds are comprised of institutions, private banks, family offices and high net worth individuals from around the world. Mr. Okamoto has completed financings for hundreds of public and private companies across a broad array of industries and sectors. In 1998, Mr. Okamoto co-founded and was the President of TradePortal.com, Inc. TradePortal.com, Inc. is a software development company and its wholly owned subsidiary, TradePortal Securities, Inc., a direct access execution brokerage firm. Mr. Okamoto was instrumental in developing the proprietary Trade MatrixTM software platform. In 2000, TradePortal.com, Inc. sold a minority stake to Thomson Reuters (TRI:NYSE), a US \$12 billion revenue company. In 1995, he founded First Stage Capital, Inc. which specializes in investment banking and consulting to public and private companies. From 1983 to 1994, he was employed in the securities industry with Shearson Lehman Brothers, Prudential Securities and Paine Webber. Mr. Okamoto is widely recognized as an advanced trader specializing in short-term trading and has more than 25 years of extensive experience in technical market analysis techniques and has been a frequent speaker at national trading venues. From 1987 to 1990, he created and hosted the television program, The Income Report in Los Angeles. He has also appeared on CNN and The MacNeil-Lehrer Report.

Ronald F. Richards. Mr. Richards has served as a Director of Optex Systems, Inc. (Delaware) since October 2008 and has served as a Director of Optex Systems Holdings since March 30, 2009, as well as the Chairman of the Board of Optex Systems Holdings. Mr. Richards is the founder and Managing Director of Gray Wolf Partners, LLC, a strategic and financial advisory firm. From February 2007 to October 2008, he served as a Managing Director of Viking Asset Management, LLC where his responsibilities included: (i) sourcing, conducting due diligence, and structuring potential investment opportunities and (ii) working with portfolio companies to enhance shareholder value. He previously served as Chief Financial Officer and Senior Vice President, Business Development of Biopure Corporation, a publicly traded biotechnology company developing oxygen therapeutics and as a Managing Director, Corporate Finance of Wells Fargo Van Kasper. Mr. Richards has over 22 years of experience working with public and private companies in the areas of investment banking, corporate finance, law and accounting. He has structured and executed numerous public offerings and private placements raising a total of more than \$660 million. He also co-authored PIPES: A CEO's Guide to Successful Private Placements in Public Equities. Mr. Richards holds JD, MBA and BA degrees from UCLA. He is a member of the State Bar of California and a retired Certified Public Accountant.

Danny Schoening, Mr. Schoening joined Optex Systems, Inc. (Texas) in January 2008. Upon the acquisition of the assets of Optex Systems, Inc. (Texas) by Optex Systems, Inc. (Delaware), Mr. Schoening became the COO of Optex Systems, Inc. (Delaware) (as of September 28, 2008) and he commenced service with Optex Systems Holdings as its Chief Operating Officer as of the date of the reorganization, March 30, 2009. He has been instrumental in establishing the systems and infrastructure required to continue Optex System's rapid growth. This activity was rewarded with Optex System's recent ISO9001:2000 Certification. From February 2004 to January 2008, Danny was the Vice President of Operations for The Finisar Corporation AOC Division for 4 years where he led a team of up to 200 employees to produce vertical cavity lasers for the data communications industry at production rates of hundreds of thousands of units per week. Prior to Finisar, Danny was the Director of Operations for multiple divisions of Honeywell International. Serving the Automotive, Medical, Aerospace, and Consumer Commercial Markets, During this 17 year period, Danny was recognized with Honeywell's Lund Award, their highest award for developing employee resources. Danny has a broad experience level in the following technologies: Mechanical Assembly Processes, Micro-Electronic Assembly Processes, Laser Manufacturing, Plastic Molding, Metal Machining, Plating, Thick Film Printing, Surface Mount Technology, Hall Effect Technology and MEMS based Pressure Devices. Danny received a Bachelors of Science in Manufacturing Engineering Technology from the University of Nebraska, an MBA from Southern Methodist University, and holds three U.S. patents.

Karen L. Hawkins. Ms. Hawkins has served Optex Systems Holdings as its Vice President, Finance and Controller, since the date of the reorganization, March 30, 2009 and was the controller of Optex Systems, Inc. (Delaware). effective September 28, 2009. She began her employment with Optex Systems, Inc. (Texas) in April 2007. Ms. Hawkins is a Certified Public Accountant since 1992 with over 22 years experience in Financial Accounting and Management, primarily focused in the Defense and Transportation Industries. She has a strong background in both Financial & Cost Accounting, with extensive Government Pricing, Financial Analysis, and Internal Auditing experience. Her past history also includes Program Management, Materials Management and Business Development. She brings over 14 years direct experience in Government Contracting with a strong knowledge of Cost Accounting Standards Board and Federal Acquisition Regulation. Her previous employment includes General Dynamics – Ordinance and Tactical Division, Garland (formerly known as Intercontinental Manufacturing) for over 13 years from November, 1994 through March, 2007. During her tenure there she served in the roles of Controller (Accounting & IT), Program Manager over a \$250M 3 year Army Indefinite Delivery/Indefinite Quantity (Indefinite Delivery/Indefinite Quantity) type contract, as well as Materials Manager with oversight of Purchasing, Production Control & Warehousing functions. Prior to her employment at General Dynamics, Ms. Hawkins served in various finance and accounting positions at Luminator, a Mark IV Industries Co, and Johnson Controls, Battery Division -Garland. Karen received her Bachelor's Degree in Business Administration in Accounting from Stephen F. Austin State University in Texas in 1986.

Family Relationships

There are no family relationships among the officers and directors.

Presiding Director

Our Chairman, Ronald F. Richards, acts as the presiding director at meetings of our board of directors. In the event that Mr. Richards is unavailable to serve at a particular meeting, responsibility for the presiding director function will rotate among the chairmen of each of the committees of our board of directors.

Corporate Governance

Our board of directors believes that sound governance practices and policies provide an important framework to assist them in fulfilling their duty to stockholders. Our board of directors is working to adopt and implement many "best practices" in the area of corporate governance, including annual review of constitution of separate committees for the areas of audit and compensation, maintenance of a majority of independent directors, and written expectations of management and directors, among other things. In 2010, all directors attended 75% of our meetings of the board of directors.

Code of Ethics

Our board of directors has adopted a Code of Ethics which has been distributed to all directors, and executive officers, and will be distributed to employees and will be given to new employees at the time of hire. The Financial Code of Ethics contains a number of provisions that apply principally to our Principal Executive Officer, Principal Financial Officer and other key accounting and financial personnel. A copy of our Code of Business Conduct and Ethics can be found under the "Investor Relations" section of our website (www.optexsys.com) under the section for corporate governance. We also intend to disclose any amendments or waivers of our Code on our website.

Board and Committee Meetings

We are incorporated under the laws of the State of Delaware. The interests of our stockholders are represented by the board of directors, which oversees our business and management.

The board of directors meets regularly during the year and holds special meetings and acts by unanimous written consent whenever circumstances require. The board held 4 meetings (including special meetings) and took action by unanimous written consent 1 time during our fiscal year ended October 3, 2010.

If the board of directors convenes a special meeting, the non-management directors meet in executive session if circumstances warrant.

Board Committees

At this time, the board of directors currently has an inactive Audit Committee which did not meet in fiscal year 2010, of which Ronald F. Richards is the sole member, and we do not have a formal charter at this time due to the size of the Committee but intend to adopt one at a future date.

Board nominations

Stockholders wishing to bring a nomination for a director candidate before a stockholders meeting must give written notice to our Corporate Secretary, either by personal delivery or by United States mail, postage prepaid. The stockholder's notice must be received by the Corporate Secretary not later than (a) with respect to an Annual Meeting of Stockholders, 90 days prior to the anniversary date of the immediately preceding annual meeting, and (b) with respect to a special meeting of stockholders for the election of directors, the close of business on the tenth day following the date on which notice of the meeting is first given to stockholders. The stockholder's notice must set forth all information relating to each person whom the stockholder proposes to nominate that is required to be disclosed under applicable rules and regulations of the SEC, including the written consent of the person proposed to be nominated to being named in the proxy statement as a nominee and to serving as a director if elected. The stockholder's notice must also set forth as to the stockholder making the nomination (i) the name and address of the stockholder, (ii) the number of shares held by the stockholder, (iii) a representation that the stockholder is a holder of record of stock of the Optex Systems Holdings, entitled to vote at the meeting and intends to appear in person or by proxy at the meeting to nominate the person named in the notice, and (iv) a description of all arrangements or understandings between the stockholder and each nominee.

Stockholder Communications with the Board of Directors

Stockholders may communicate directly with the board of directors or any board member by writing to them at Optex Systems Holdings, Inc., 1420 Presidential Drive, Richardson, TX 75081. The outside of the envelope should prominently indicate that the correspondence is intended for the board of directors or for a specific director. The secretary will forward all such written communications to the director to whom it is addressed or, if no director is specified, to the entire board of directors.

Director Attendance at Annual Meetings of Stockholders

Directors are encouraged to attend annual meetings, although such attendance is not required.

Board Independence

Our board of directors has determined that one of our directors would meet the independence requirements of the American Stock Exchange, if such standards applied to the Company. In the judgment of the board of directors, Mr. Hirschman and Mr. Okamoto do not meet such independence standards. In reaching its conclusions, the board of directors considered all relevant facts and circumstances with respect to any direct or indirect relationships between the Company and each of the directors, including those discussed under the caption "Certain Relationships and Related Transactions" below. Our board of directors determined that any relationships that exist or existed in the past between the Company and each of the independent directors were immaterial on the basis of the information set forth in the above-referenced sections.

Director Compensation

See table below under "Executive Compensation – Director Compensation."

Item 11 Executive Compensation

Executive Compensation

The board of directors administers our option compensation plan. Our Principal Executive Officer and other members of management regularly discuss our compensation issues with the Board of Directors. Subject to Board review, modification and approval, Mr. Hirschman typically makes recommendations respecting bonuses and equity incentive awards for the other members of the executive management team. The Board establishes all bonus and equity incentive awards for Mr. Hirschman in consultation with other members of the management team.

Summary Compensation Table

The following table sets forth, for the years indicated, all compensation paid, distributed or accrued for services, including salary and bonus amounts, rendered in all capacities by Optex Systems Holdings' principal executive officer, principal financial officer and all other executive officers who received or are entitled to receive remuneration in excess of \$100,000 during the stated periods. These officers are referred to herein as the "named executive officers." Except as provided below, none of our executive officers received annual compensation in excess of \$100,000 during the last two fiscal years.

						(Option	A	ll Other		
		Salary	Bonus		Stock	Αv	wards (\$)C	Con	npensation	1	Total
Name and Principal Position	Year	(\$)	(\$)	Av	vards (\$)		(5)	((\$) (8)		(\$)
Stanley A. Hirschman,	2010(7)	\$ 51,000	\$	\$	-	\$	-	\$	16,650	\$	67,650
President (6)	2009(4)	-	-		-		-		25,000		25,000
Danny Schoening,	2010	\$ 196,574	\$ 57,300	\$	-	\$	21,584	\$	15,524	\$	290,982
COO (6)	2009	182,932	11,000		-		10,588		-		204,520
	2008(1,2)	122,646	10,300		7,500		-		-		140,446
Karen Hawkins, VP											
Finance / Controller (6)	2010	\$ 146,575	\$ 7,450	\$	-	\$	14,854	\$	6,041	\$	174,920
	2009	133,647	7,271		-		5,516		-		146,434
	2008	132,473	300		-		-		-		132,773
Andrey Oks, CEO, CFO, Secretary,											
Treasurer and Director	2008(3)	-	-		10,000		-		-		10,000

- 1 The compensation depicted is not reflective of a full year's compensation as Danny Schoening did not begin employment until the second quarter of fiscal year 2008. For Mr. Schoening, information is for service as an officer of Optex Texas and Optex Delaware. Given the fact that there has not been a change in fiscal year but rather adoption of the fiscal year of the accounting acquirer, there has been no adjustment made to treat the period since the change in fiscal year as a stub period, and all numbers presented are for complete fiscal years.
- 2 Stock awards include issues of 10,000 common shares of Irvine Sensors Common Stock on January 16, 2008 at the then current market share price of \$0.75 per share.
- 3Mr. Oks was appointed as an officer of Sustut as of September 15, 2008 and resigned as of March 29, 2009. Mr. Oks was given 10,000,000 shares of restricted stock as compensation for services which was forfeited to Sustut on the date of his resignation.
- 4 Mr. Hirschman's compensation in 2009 consisted solely of Director's Fees. He received no other compensation.
- 5The amounts in the "Option awards" column reflect the dollar amounts recognized as the executive portion of compensation expense for financial statement reporting purposes for each named executive officer during fiscal 2009 and fiscal 2010, as required by FASB ASC 718 (prior authoritative literature SFAS 123(R), disregarding any estimates for forfeitures relating to service-based vesting conditions. For the assumptions relating to these valuations, see note 12 to our fiscal 2009 audited financial statements. Andrey Oks was an executive of Sustut Exploration, Inc. during 2008, prior to the reverse merger on March 30, 2009. Concurrent with the reverser merger and name change to Optex Systems Holdings, Inc on March 30, 2009 Optex Systems Holdings adopted the fiscal year end of the accounting acquirer and changed the period end from December 31 to a fiscal year end of September. There were no earnings of either of these individuals subsequent to the reverse merger and adoption of the accounting acquirer's fiscal period. All compensation expense shown for these individuals prior to the March 30, 2009 reorganization are depicted in the calendar year ending December 31, 2008.
- 6Danny Schoening, Karen Hawkins and Stanley A. Hirschman were all executives of Optex Systems Holdings subsequent to the March 30, reorganization. Prior to the reorganization Danny Schoening and Karen Hawkins were

executives of Optex Systems, Inc (Texas) and Optex Systems, Inc (Delaware) and Stanley Hirschman became an executive of Optex Systems, Inc (Delaware) in September 2008. Both Optex Systems, Inc. (Texas) and Optex Systems, Inc (Delaware) had previously been operating under an October through September fiscal year end and as such, compensation for these individuals is depicted in fiscal years beginning in October and ending in September for each of the years 2008 and 2009.

7This includes director fees paid through January 2010 of \$10,000, listed as "Other". Commencing February 1, 2010 Stanley Hirschman was paid a salary for which he received \$51,000 in 2010, which is listed as "Salary".

8 Other compensation includes employee and dependant medical insurance benefits offered as part of executive compensation.

Option Grants in Last Fiscal Year

There were no plan based awards made to our named executive officers during the fiscal year ended October 3, 2010.

Employment Agreement

Optex Systems Holdings entered into an employment agreement with Danny Schoening dated December 1, 2008. The term of the agreement commenced as of December 1, 2008 and the initial term would have expired on June 1, 2010, but has automatically renewed through December 1, 2011. Thereafter, the term of the agreement shall be automatically extended for successive 18 month periods, unless Optex Systems Holdings shall provide a written notice of termination at least ninety (90) days, or the Mr. Schoening shall provide a written notice of termination at least 90 days, prior to the end of the initial term or any extended term, as applicable. During the first eighteen months of the term of the agreement, Optex Systems Holdings paid to Schoening a base salary at the annual rate of \$190,000, and his base salary for the first renewal term has continued at the same rate. Schoening was paid a one-time bonus of \$10,000 at the commencement of the employment agreement in December 2008 and was granted 1,414,649 options to purchase common stock of Optex Systems Holdings at an exercise price of \$0.15 per share at the time of the closing of the reorganization.

On each subsequent renewal date of the commencement of employment, Schoening's base salary shall be reviewed by the Board and may be increased to such rate as the Board, in its sole discretion, may hereafter from time to time determine. During the term of the agreement, Schoening shall be entitled to receive bonuses of up to 30% of his base salary per year at the discretion of Optex Systems Holdings' Board of Directors pursuant to performance objectives to be determined by the Board of Directors. Any bonuses shall be payable in cash and shall be paid within ninety (90) days of any year anniversary of the date of the agreement. Upon closing of the reorganization, Optex Systems Holdings granted Schoening stock options equal to 1% of the issued and outstanding shares of Optex Systems Holdings immediately after giving effect to the reorganization, with 34% of the options having vested on March 30, 2010, and 33% of the options vesting on each of March 31, 2011 and March 31, 2012.

The employment agreement events of termination thereof: (i) death of Mr. Schoening; (ii) termination by Optex Systems Holdings for cause (including conviction of a felony, commission of fraudulent acts, willful misconduct by Mr. Schoening, continued failure to perform duties after written notice, violation of securities laws and breach of the employment agreement), (iii) termination without cause by Optex Systems Holdings and (iv) termination by Mr. Schoening for good reason (including breach by Optex Systems Holdings of its obligations under the agreement, the requirement for Mr. Schoening to move more than 100 miles away for his employment without consent, and merger or consolidation that results in more than 66% of the combined voting power of the then outstanding securities of Optex Systems Holdings or its successor changing ownership or a sale of all or substantially all of Optex Systems Holdings' assets, without the surviving entity assuming the obligations under the agreement). For a termination by Optex Systems Holdings for cause or upon death of Mr. Schoening, Mr. Schoening shall be paid salary and bonus earned through the date of termination. For a termination by Optex Systems Holdings without cause or by Mr. Schoening with good reason, Mr. Schoening shall also be paid six months base salary in effect and all granted stock options shall remain exercisable for a period of two years after such termination, with all unvested stock options immediately vesting. The agreement contains a standard non-solicitation and non-compete agreement that extends for one year subsequent to termination thereof.

Optex Systems Holdings does not have any other employment agreements with its executive officers and directors.

Equity Compensation Plan Information

Optex Systems Holdings currently has an option compensation plan covering the issuance of options for the purchase of up to 6,000,000 shares. The purpose of the Plan is to assist Optex Systems Holdings in attracting and retaining highly competent employees and to act as an incentive in motivating selected officers and other employees of Optex Systems Holdings and its subsidiaries, and directors and consultants of Optex Systems Holdings and its subsidiaries, to achieve long-term corporate objectives. There are 6,000,000 shares of common stock reserved for issuance under this Plan. As of October 3, 2010, Optex Systems Holdings had issued 2,681,649 share options under this Plan of which 776,981 shares had vested as of October 3, 2010.

Outstanding Equity Awards as of October 3, 2010

Option Awards Number of shares underlying unexercised options Non-Plan Equity Incentive Plan Awards # Exercise **Expiration** Exercisable Unexercisable Total Granted Price Name Date **Footnotes Danny Schoening** 480,981 933,668 1,414,649 0.15 3/29/2016 (1) Karen Hawkins 62,500 187,500 250,000 0.15 5/13/2016 (2)

- Options granted on March 30, 2009 pursuant to employment agreement and reverse Merger. Shares vest over 3 years at a rate of 34%, 33% and 33% for each respective anniversary date subsequent to 2009 and expire after seven years. As of October 3, 2010 480,981 of the options had vested.
- (2) Options granted on May 14, 2009 pursuant to employee stock option compensation plan. Shares vest over 4 years at a rate of 25% per year each respective anniversary date subsequent to 2009 and expire after seven years. As of October 3, 2010 62,500 of the options had vested.

Nonqualified deferred compensation

We had no non-qualified deferred compensation plans during year ended October 3, 2010.

Director Compensation

The following table provides information regarding compensation paid to directors for services rendered during the year ended October 3, 2010.

		Fees									
	F	Earned or			N	on-Equit	y Nonqual	ified			
		Paid in	Stock	Op	tion Inc	entive Pl	an Deferr	ed	All Othe	er	
		Cash	Awards	Aw	ards Co	mpensati	o@compens	atio60	ompensa	tion	
Name		(\$)	(\$)	(\$)	(\$)	Earnings	s (\$)	(\$)	-	Γotal (\$)
Ronald F. Richards											
(1)	\$	120,000		_	_		_	_		\$	120,000
Stanley A. Hirschman											
(2)		10,000		_	_						10,000
Merrick Okamoto (3)		_	-	_	_		_				_

- (1) Director Fees paid monthly from October 2009 through September 2010. Mr. Richards is paid \$2,500 monthly as an Independent Director, \$2,500 monthly for serving as Chairman of the Audit Committee, and \$5,000 monthly for serving as Chairman of the Board of Directors.
- (2) Director Fees paid monthly from October 2009 through January 2010. Mr. Hirschman was paid \$2,500 monthly as a Director. Effective as of February 1, 2010, director fees to Mr. Hirschman were discontinued, and he was paid a direct salary from Optex Systems Holdings.
 - (3) Mr. Okamoto serves as a non-independent director and does not earn directors fees.

The members of our board of directors are actively involved in various aspects of our business ranging from relatively narrow board oversight functions to providing hands-on guidance to our executives and scientific staff with respect to

matters within their personal experience and expertise. We believe that the active involvement of all directors in our principal business and policy decisions increases our board of directors' understanding of our needs and improves the overall quality of our management decisions.

With the exception of Mr. Hirschman, our directors are compensated separately for service as members of our board of directors. As of February 1, 2010, Mr. Hirschman was paid a salary from Optex Systems Holdings as disclosed in the executive compensation table above.

Nonqualified deferred compensation

We had no non-qualified deferred compensation plans during year ended October 3, 2010.

Post-Termination Compensation

We have not entered into change in control agreements with any of our named executive officers or other members of the executive management team other than the provision with respect to Mr. Schoening described above. No awards of equity incentives under our 2009 Stock Option Plan provide for immediate vesting upon a change in control. However, our Board of Directors has the full and exclusive power to interpret the plans, including the power to accelerate the vesting of outstanding, unvested awards. A "change in control" is generally defined as (1) the acquisition by any person of 66% or more of the combined voting power of our outstanding securities or (2) the occurrence of a transaction requiring stockholder approval and involving the sale of all or substantially all of our assets or the merger of us with or into another corporation.

Item 12 Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters

On December 13, 2010, we had 139,444,940 shares of common stock, and 1,027 shares of Series A preferred stock issued and outstanding. The following table sets forth certain information with respect to the beneficial ownership of our securities as of September 20, 2010, for (i) each of our directors and executive officers; (ii) all of our directors and executive officers as a group; and (iii) each person who we know beneficially owns more than 5% of our common stock.

Beneficial ownership data in the table has been calculated based on Commission rules that require us to identify all securities that are exercisable for or convertible into shares of our common stock within 60 days of September 20, 2010 and treat the underlying stock as outstanding for the purpose of computing the percentage of ownership of the holder.

Except as indicated by the footnotes following the table, and subject to applicable community property laws, each person identified in the table possesses sole voting and investment power with respect to all capital stock held by that person. The address of each named executive officer and director, unless indicated otherwise by footnote, is c/o Optex Systems Holdings' corporate headquarters.

Except as otherwise set forth below, the address of each of the persons listed below is Optex Systems Holdings' address.

Title of Class Common Stock	Name of Beneficial Owner	Number of Shares	Preferred Conversion (4)	Combined Ownership	Percentage of Outstanding Shares
5% Holders	Arland Holdings, Ltd. (1) Sileas Corporation (2,3)	11,148,935 102,184,347	37,040,000	11,148,935 139,224,347	5.89% 73.52%
Directors and Officers:	Stanley A. Hirschman (2) Danny Schoening (5)(8) Karen Hawkins (9)	102,184,347 102,665,328 62,500	37,040,000 37,040,000	139,224,347 139,698,460	73.52% 73.62%

Edgar Filing: General Moly, Inc - Form PRE 14A

	Ronald F. Richards	-	-	-	-
	Merrick Okamoto(7)	1,950,000	-	1,950,000	1.40%
Directors and officers as a group (5 Individuals)		104,134,347	37,040,000	141,710,960	75.02%
72					

			Percentage
			of
	Name of Beneficial	Number of	Outstanding
Title of Class	Owner	Shares	Shares
Preferred Stock			
5% Holders	Sileas Corporation (2,3)	926	90.0%
	Alpha Capital Anstalt (7)	101	10.0%

- 1 Represents shares held by Arland Holdings, Ltd., which is located at 551 5th Avenue, Suite 1601, New York, NY 10176. Arie Rabinowitz has voting control over the shares held by Arland Holdings, Ltd.
- 2 Represents shares held by Sileas of which Stanley Hirschman, a Director/Officer Optex Systems Holdings, has a controlling interest (80%); therefore, under Rule 13d-3 of the Exchange Act, Mr. Hirschman is deemed to be the beneficial owner, along with Mr. Schoening.
- 3 Sileas' ownership interest in Optex Systems Holdings has been pledged to Longview as security for a loan in connection with the acquisition of Longview's interests in Optex Delaware by Sileas. Investment decisions for Longview are made by its investment advisor, Viking Asset Management, LLC. Mr. Peter Benz is the Chairman, Chief Executive Officer and a Managing Member of Viking Asset Management and may be deemed to control its business activities, including the investment activities of Longview. Mr. Merrick Okamoto who is a director of Optex Systems Holdings is the President and a Managing Member of Viking Asset Management and may be deemed to control its business activities, including the investment activities of Longview. In the event of a default by Sileas on its debt obligation to Longview, the shares held by Sileas may be returned to Longview. Viking and Longview each may be deemed to have shared voting and dispositive authority over the shares of Optex Systems Holdings' common stock if they are returned to Longview. In such an event, Mr. Benz and Mr. Okamoto, as control persons of Viking and/or Longview, may be deemed to beneficially own all such shares; however, they have stated that they would disclaim such beneficial ownership were this to occur.
- 4Represents shares of common stock issuable upon conversion of preferred stock held by the stockholder. Sileas Corporation holds 90% or 926 of the preferred shares which are convertible into 37,040,000 common shares. Alpha Capital owns the remaining 10% or 101 preferred shares convertible into 4,040,000 common shares, representing less than 2.13% total beneficially ownership.
- 5Represents 102,184,347 shares held by Sileas of which Mr. Schoening, an Officer of Optex Systems Holdings, has a controlling interest (15%); therefore, under Rule 13d-3 of the Exchange Act, Mr. Schoening is deemed to be the beneficial owner, along with Mr. Hirschman, of those shares.
- 7 Represents shares held by Alpha Capital Anstalt, which is located at Pradfant 7, 9490 Furstentums, Vaduz, Lichtenstein. Konrad Ackerman has voting control and investment power over the shares held by Alpha Capital Anstalt
- 8 Represents 975,000 shares of Common Stock and 975,000 warrants held by Longview Fund, LP. Investment decisions for Longview are made by its investment advisor, Viking Asset Management, LLC. Mr. Merrick Okamoto who is a director of Optex Systems Holdings is the President and a Managing Member of Viking Asset Management and may be deemed to control its business activities, including the investment activities of Longview. Mr. Okamoto, as a control person of Viking and/or Longview, may be deemed to beneficially own all such shares; however, he disclaims such beneficial ownership.
- 8 Includes options to purchase 480,981 shares of our common stock which have vested and are currently exercisable. 9 Represents options to purchase 62,500 shares of our common stock which have vested and are currently exercisable.

Item 13 Certain Relationships and Related Transactions, and Director Independence

Relationship between Optex Systems, Inc. (Texas), Irvine Sensors Corporation and Longview and Alpha

Longview and Alpha were owed certain debt by Irvine Sensors Corporation including debt evidenced by (i) a December 29, 2006 Term Loan and Security Agreement executed by Irvine Sensors Corporation and Longview and Alpha, and (ii) a series of secured promissory notes purchased by them and issued to them on December 29, 2006, July 19, 2007 and November 28, 2007. As of August 24, 2008, the total amount due under all of the described notes was approximately \$18.4 million. Optex Systems, Inc. (Texas), which was and is a wholly owned subsidiary of Irvine Sensors Corporation, was a guarantor of all of those notes, and pursuant to related security agreements Longview and Alpha had a validly perfected, fully enforceable security interest in all personal property of Optex Systems, Inc. (Texas). On September 19, 2008, pursuant to an Assignment and Stock/Note Issuance Agreement, Alpha and Longview transferred and assigned to Optex Systems, Inc. (Delaware) which assumed, \$15 million of their respective interests and rights in the aforesaid notes and obligations to Optex Systems, Inc. (Delaware) in exchange for \$9 million of equity and \$6 million of debt.

Acquisition of Assets of Optex Systems, Inc. (Texas) by Optex Systems, Inc. (Delaware) on October 14, 2008

On October 14, 2008, in a purchase transaction that was consummated via public auction, Optex Systems, Inc. (Delaware) purchased all of the assets of Optex Systems, Inc. (Texas) in exchange for \$15 million of Irvine Sensors Corporation debt owned by it and the assumption of approximately \$3.8 million of certain Optex Systems, Inc. (Texas) liabilities. The \$15 million of Irvine Sensors Corporation debt was contributed by Longview and Alpha to Optex Systems, Inc. (Delaware) in exchange for a \$6 million note payable from Optex Systems, Inc. (Delaware) and a \$9 million equity interest in Optex Systems, Inc. (Delaware). Longview and Alpha owned Optex Systems, Inc. (Delaware) until February 20, 2009, when Longview sold 100% of its interests in Optex Systems, Inc. (Delaware) to Sileas, as discussed below. In referring to these transactions, Optex Systems, Inc. (Delaware) is considered to be the successor entity to Optex Systems, Inc. (Texas), the predecessor entity.

Secured Promissory Notes and Common Shares Issued in connection with Purchase by Optex Systems, Inc. (Delaware)

In connection with the public sale of the Optex Systems, Inc. (Texas) assets to Optex Systems, Inc. (Delaware), Optex Systems, Inc. (Delaware) delivered to each of Longview and Alpha a Secured Promissory Note due September 19, 2011 in the principal amounts of \$5,409,762 and \$540,976, respectively. Each Note bears simple interest at the rate of 6% per annum, and the interest rate upon an event of default increases to 8% per annum. After 180 days from the issue date, the principal amount of the Notes and accrued and unpaid interest thereon may be converted into Optex Systems, Inc. (Delaware) common stock at a conversion price of \$1.80 per share (pre-split and pre-reorganization price). The Notes may be redeemed prior to maturity at a price of 120% of the then outstanding principal amount plus all accrued and unpaid interest thereon. The obligations of Optex Systems, Inc. (Delaware) under the Notes are secured by a lien against all of the assets of Optex Systems, Inc. (Delaware) in favor of Longview and Alpha. In addition, Optex Systems, Inc. (Delaware) issued common stock to each of Longview and Alpha in the quantities of 45,081,350 and 4,918,650, respectively. On October 30, 2008, Alpha sold its Optex Systems, Inc. (Delaware) common stock to Arland Holding, Ltd. On February 20, 2009, Longview sold its Note to Sileas (see below).

Acquisition by Sileas of Longview's Interests in Optex Systems, Inc. (Delaware) on February 20, 2009

On February 20, 2009, Sileas purchased 100% of the equity and debt interest held by Longview, representing 90% of Optex Systems, Inc. (Delaware), in a private transaction. The primary reason for the acquisition was to eliminate shareholder control of Optex Systems Holdings by Longview and to limit any perception of control over the day-to-day operations of Optex Systems Holdings, whether or not such control actually existed. While Longview makes investments in a variety of companies, it strives to invest passively and leave the day-to-day operations of the companies in its investment portfolio to the management teams of those companies. In addition, the acquisition allowed Optex Systems Holdings to avoid potential conflicts of interest or other related business issues that might have adversely affected Optex Systems Holdings' operations as a result of Longview's investments in other companies.

The purchase price for the acquisition was \$13,524,405. Sileas issued a purchase money note to Longview for the full amount of the purchase price in exchange for 45,081,350 shares of common stock of Optex Systems Holdings (representing 90% of the outstanding shares) and transfer to Sileas of a note dated December 2, 2008, issued by Optex Systems Holdings to Longview in the principal amount of \$5,409,762. No contingent consideration is due the seller in the transaction. The obligations of Sileas under the Note are secured by a security interest in Optex Systems Holdings' common and preferred stock owned by Sileas that was granted to Longview pursuant to a Stock Pledge Agreement delivered by Sileas to Longview and also by a lien on all of the assets of Sileas. On March 27, 2009, Sileas and Alpha (which owned the balance of the \$6,000,000 of the notes) exchanged the \$6,000,000 aggregate principal amount of notes, plus accrued and unpaid interest thereon, for 1,027 shares of Optex Systems, Inc. (Delaware) Series A preferred stock.

Sileas has no operations or business activities other than holding the stock and notes described above and has no revenues, and it holds no assets other than the stock and notes described above. The management of Sileas believes that the value of its common stock and preferred stock holdings in Optex Systems Holdings will increase over time. Sileas plans to repay Longview, no later than the maturity date, through some combination of a recapitalization of Sileas equity and debt and partial or full liquidation of its interests in Optex Systems Holdings. Sileas will be limited by the extent of the stock price of Optex Systems Holdings and limitations on ability to resell the stock it owns in Optex Systems Holdings.

Secured Promissory Note Due February 20, 2012/Longview Fund, LP

As a result of the transaction described above between Sileas and Longview on February 20, 2009, Sileas, the new majority owner of Optex Systems, Inc. (Delaware), executed and delivered to Longview, a Secured Promissory Note due February 20, 2012 in the principal amount of \$13,524,405. The Note bears simple interest at the rate of 4% per annum, and the interest rate upon an event of default increases to 10% per annum. In the event that a Major Transaction occurs prior to the maturity date resulting in the Borrower receiving Net Consideration with a fair market value in excess of the principal and interest due under the terms of this Secured Note, then in addition to paying the principal and interest due, Sileas shall also pay an amount equal to 90% of the consideration. "Major Transaction" refers to a transaction whereby Optex Systems, Inc. (Delaware) would consolidate or merge into or sell or convey all or substantially all of its assets to a third party entity for more than nominal consideration, and "Net Consideration" refers to the fair market value of the consideration received in connection with a Major Transaction less all outstanding liabilities of Optex Systems, Inc. (Delaware).

Reorganization/Share Exchange

On March 30, 2009, a reorganization occurred whereby the then existing shareholders of Optex Systems, Inc. (Delaware) exchanged their shares of common stock with the shares of common stock of Optex Systems Holdings as follows: (i) the outstanding 85,000,000 shares of Optex Systems, Inc. (Delaware) common stock were exchanged by Optex Systems Holdings for 113,333,282 shares of Optex Systems Holdings common stock, (ii) the outstanding 1,027 shares of Optex Systems, Inc. (Delaware) Series A preferred stock were exchanged by Optex Systems Holdings for 1,027 shares of Optex Systems Holdings Series A preferred stock and (iii) the 8,131,667 shares of Optex Systems, Inc. (Delaware) common stock purchased in the private placement, which also occurred on March 30, 2009, were exchanged by Optex Systems Holdings for 8,131,667 shares of Optex Systems Holdings common stock. The per share price in the private placement was \$0.15 per share of common stock, and the closing date was March 30, 2009. Optex Systems, Inc. (Delaware) remains a wholly-owned subsidiary of Optex Systems Holdings.

At the time of the reorganization, 25,000,000 post split shares owned by Andrey Oks, the former CEO of Optex Systems Holdings, were cancelled. Immediately prior to the closing, 17,449,991 shares of Optex Systems Holdings common stock were outstanding. The 17,449,991 shares from the 17,999,995 shares outstanding as of December 31, 2008 plus the 26,999,996 shares issued in conjunction with the 2.5:1 forward stock split authorized by the Sustut Board and shareholders and effected on February 27, 2009 less retirement of Andrey Oks' 25,000,000 shares and cancellation of 3,800,000 shares previously issued to Newbridge Securities Corporation, shares plus issuance of 1,250,000 shares in payment for two investor relations agreements. The total outstanding common shares of Optex Systems Holdings subsequent to the closing of the reorganization is as follows:

Existing Sustut Shareholders	17,449,991
Optex Systems, Inc. (Delaware) shares exchanged	113,333,282
Optex Systems, Inc. (Delaware) Private Placement shares exchanged	8,131,667
Total Shares after reorganization	138,914,940
Cancellation of shares - American Capital Ventures	(700,000)
Private placement - June 29, 2009	750,000
Issuance of shares as consideration - ZA Consulting	480,000
Shares Outstanding on October 3, 2010	139,444,940

Short Term Note Payable/Longview Fund - On September 23, 2008 Optex Systems, Inc. (Texas) borrowed \$146,709 from Longview and issued a promissory note dated September 23, 2008, to Longview in connection therewith. The September 23, 2008 Note bore interest at the rate of 10% per annum with interest accruing until the maturity date of the September 23, 2008 Note, which was originally set as November 7, 2008. On March 30, 2009 in conjunction with the reorganization and Private Placement, Longview purchased 3.25 units of the Private Placement using \$146,250 of the amount due under the Note as consideration for the purchase. The outstanding balance related to the original note issue of \$459 plus \$11,101 of accrued interest was paid in September 2009.

On October 27, 2009, Optex Systems Holdings borrowed \$250,000 from Longview on an unsecured basis pursuant to a promissory note, which originally expired on December 1, 2009, but was extended until July 15, 2010. The note bore interest at the rate of 10% per annum, and all accrued and unpaid interest was due upon maturity.

In exchange for the extension, Optex Systems Holdings granted Longview a warrant to purchase 100,000 shares of restricted common stock with an exercise price of \$0.15 per share and a term of three years. On March 22, 2010, Optex Systems Holdings repaid \$125,000 in principal plus \$10,000 in accrued interest on the outstanding Longview note. On June 4, 2010, Optex Systems Holdings paid off the remaining principal balance and all accrued and unpaid interest thereon.

We made requests of counsel representing Sustut's directors and officers to obtain additional information into the principles behind their determination that the securities of the registrant issued in the March 30, 2009 share exchange represented "fair market value" to acquire the business operations of Optex Systems, Inc. (Delaware), and they were not able to provide any information. We confirm that we have no affiliation with Sustut's former counsel, Anslow & Jacklin, who was our only source of information regarding the prior history of Sustut and that the result of our request was that they stated they had no information and were not able to obtain further information on this issue.

We have not been able to provide further background as to how the merger consideration was determined beyond the fact that it was determined by negotiation between Sustut and Optex Systems, Inc. (Delaware). Thus, we have invoked Rule 409(b) which states: "(b) The registrant shall include a statement either showing that unreasonable effort or expense would be involved or indicating the absence of any affiliation with the person within whose knowledge the information rests and stating the result of a request made to such person for the information."

Transactions with Executive Management

See the "Executive Compensation" section for a discussion of the material elements of compensation awarded to, earned by or paid to our named executive officers. Other than as stated in the "Executive Compensation" section, we have not entered into any transactions with executive management.

Item 14 Principal Accounting Fees and Services

The following table sets forth the fees paid to date for services rendered by EFP Rotenberg as successor by merger, effective October 1, 2009, of the registered public accounting firm Rotenberg & Co., LLP during fiscal years ended October 3, 2010 and September 27, 2009, respectively.

Fee Category	2010	2009
Audit Fees (1)	\$ 113,281	\$ 189,000
Audit-Related		
Fees-registration		
statement consents (2)	\$ 28,988	\$ 31,260
Tax Fees	\$ 8,000	-
All Other Fees (3)	\$ 5,063	-

- (1) Audit Fees are fees for professional services performed by EFP Rotenberg LLP for the audit of our annual consolidated financial statements and review of consolidated financial statements included in our 10-Q filings for the fiscal years ended October 3, 2010 and September 27, 2009, respectively.
- (2) Fees paid in related to consent for S-1 registration statement and procedures associated with SEC comment letter for S-1 registration statement.
- (3) Fees paid in related to Sarbanes-Oxley training and compliance requirements.

CHANGES IN REGISTRANT'S CERTIFYING ACCOUNTANT

On October 8, 2009, Optex Systems Holdings, Inc. ("the Company") received notice that its current auditors, Rotenberg and Co., LLP, had resigned in connection with their merger with EFP Group, which was effective as of October 1, 2009. The Company has engaged the new firm resulting from the merger, EFP Rotenberg, LLP, to continue as the Company's independent registered public accounting firm. All of the partners and employees of Rotenberg and Co., LLP and EFP Group have joined the new firm, EFP Rotenberg, LLP. EFP Rotenberg, LLP is currently registered with the PCAOB.

Rotenberg and Co. LLP was engaged by the Company on March 30, 2009 and has performed reviews for the quarters ended March 29, 2009 and June 28, 2009. Rotenberg and Co. LLP has not performed any audit services or rendered any audit report from the time of its engagement through the date of cessation of the client-auditor relationship on October 1, 2009. There have been no disagreements with Rotenberg and Co. LLP or reportable events since the date of their engagement on March 30, 2009 through the cessation of the client-auditor relationship on October 1, 2009.

On October 17, 2009, with the approval of the Company's Board of Directors, EFP Rotenberg, LLP was engaged as the Company's independent registered public accountant effective concurrent with the merger. Prior to such engagement, during the two most recent fiscal years, the Company has not consulted with EFP Rotenberg, LLP on any matter.

The Company provided Rotenberg and Co., LLP with a copy of its Current Report on Form 8-K/A prior to its filing with the Securities and Exchange Commission ("SEC") on October 19, 2009 and requested that Rotenberg and Co., LLP furnish the Company with a letter addressed to the SEC stating whether it agrees with the above statements and, if it does not agree, the respects in which it does not agree, a copy of which is filed as Exhibit 16.1 therewith.

PART IV

Item 15 Exhibits

Exhibits

Exhibit No. Description 2.1 Agreement and Plan of Reorganization, dated as of the March 30, 2009, by and between registrant, a Delaware corporation and Optex Systems, Inc., a Delaware corporation (1). 3.1 Certificate of Incorporation, as amended, of Optex Systems Holdings, Inc (2). 3.2 Bylaws of Optex Systems Holdings (1). 10.1 2009 Stock Option Plan (1). 10.2 Employment Agreement with Danny Schoening (1). 10.3 Lease for 1420 Presidential Blvd., Richardson, TX (1). 10.4 Form of Warrant (3) 10.5 Specimen Stock Certificate (3)

10.6	Contract W52H0905D0248 with Tank-automotive and Armaments Command, dated July 27, 2005 (5) (6)
10.7	Contract W52H0909D0128 with Tank-automotive and Armaments Command, dated March 24, 2009 (5)
10.8	Contract W52H0905D0260 with Tank-automotive and Armaments Command, dated August 3, 2005 (5) (6)
10.9	PO# 40050551 with General Dynamics, dated June 8, 2009 (5) (6)
10.10	Contract 9726800650 with General Dynamics, dated April 9, 2007 (5) (6)
10.11	Form of Subscription Agreement (4)
10.12	Single Source Supplier Purchase Orders with TSP Inc. (5)
10.13	Single Source Supplier Purchase Orders with SWS Trimac (5)
77	

10.14	Since Source Supplier Purchase Orders with Danaher Controls (5)
10.15	Single Source Supplier Purchase Orders with Spartech Polycast (5)
10.16	Third Amendment to Lease, between Aquiport DFWIP and Optex Systems, Inc., dated January 7, 2010 (5)
10.17	\$250,000 principal amount Note in favor of the Longview Fund, L.P., dated October 27, 2009 (9)
10.18	Investor Relations Agreement, dated April 1, 2009 between Optex Systems and American Capital Ventures, Inc. (9)
10.19	Form of Loan and Security Agreement between Optex Systems, Inc. and Peninsula Bank Business Funding, dated March 4, 2010 (5)
10.20	Form of Unconditional Guaranty executed by Optex Systems Holdings, Inc. in favor of Peninsula Bank Business Funding, dated March 4, 2010 (5)
10.21	Form of Warrant issued by Optex Systems Holdings, Inc. to Peninsula Bank Business Funding, dated March 4, 2010 (5)
10.22	Allonge to Promissory Note, dated January 5, 2010 (9)
10.23	Showcase Agreement between Optex Systems, Inc. and ECON Corporate Services, Inc., dated April 1, 2009 (9)
10.24	Consulting Agreement dated June 29, 2009, between ZA Consulting, Inc. and Optex Systems, Inc. (9)
10.25	Purchase Order dated June 28, 2010 with TACOM-Warren (7)
10.26	First Amendment to Loan and Security Agreement, dated August 3, 2010, by and between Peninsula Bank Business Funding and Optex Systems, Inc. (8)
10.27	Waiver by Peninsula Bank Business Funding to Optex Systems, Inc., dated November 24, 2010 (10)
10.28	Second Amendment to Loan and Security Agreement, dated November 29, 2010, by and between Peninsula Bank Business Funding and Optex Systems, Inc. (10)
21.1	List of Subsidiaries – Optex Systems, Inc. (1)
31.1	Certification of the Principal Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of the Principal Accounting Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification of the Principal Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

Certification of the Principal Accounting Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

- (1) Incorporated by reference from our Current Report on Form 8-K dated April 3, 2009.
- (2) Incorporated by reference from our Amendment No. 1 to Registration Statement on Form S-1 filed on September 28, 2009
- (3) Incorporated by reference from our Registration Statement on Form S-1 filed on May 19, 2009
- (4) Incorporated by reference from our Form 10-K for the fiscal year ended October 3, 2010, filed on January 11, 2010
- (5)Incorporated by reference from our Amendment No. 4 to Registration Statement on Form S-1 filed on June 14, 2010
- (6) This exhibit is missing part of the original bid/solicitation package as such information can only be obtained from third parties with which the registrant has no affiliation, and registrant has made requests from such third parties for such information, and such parties have not been able to provide such information.
- (7) Incorporated by reference from our Current Report on Form 8-K dated July 2, 2010
- (8) Incorporated by reference from our Form 10-Q for the quarter ended on June 27, 2010, filed on August 11, 2010
- (9) Incorporated by reference from our Amendment No. 5 to Registration Statement on Form S-1 filed on July 23, 2010
- (10)Incorporated by reference from our Amendment No.11 to Registration Statement on Form S-1 filed on the date of filing of this Amendment No. 1 to Form 10-K and filed immediately preceding the filing of this Amendment No. 1 on Form 10-K.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

OPTEX SYSTEMS HOLDINGS, INC.

By: /s/ Stanley A. Hirschman

Stanley A. Hirschman, Principal Executive

Officer and Director

Date: February 2, 2011

By: /s/ Karen Hawkins

Karen Hawkins, Principal Financial Officer

and Principal
Accounting Officer

Date: February 2, 2011

Pursuant to the requirements of the Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Merrick Okamoto Merrick Okamoto	Director	February 2, 2011
/s/ Ronald F. Richards Ronald F. Richards	Director	February 2, 2011
/s/ Stanley A. Hirschman Stanley A. Hirschman	Principal Executive Officer and Director	February 2, 2011
/s/ Karen Hawkins Karen Hawkins	Principal Financial Officer and Principal Accounting Officer	February 2, 2011
79		